SECURITIES AND EXCHANGE COMMISSION

FORM 8-K

Current report filing

Filing Date: 2021-08-02 | Period of Report: 2021-07-30 SEC Accession No. 0001493152-21-018256

(HTML Version on secdatabase.com)

FILER

urban-gro, Inc.

CIK:1706524| IRS No.: 465158469 | State of Incorp.:DE | Fiscal Year End: 1231 Type: 8-K | Act: 34 | File No.: 001-39933 | Film No.: 211134919 SIC: 5080 Machinery, equipment & supplies Mailing Address 1751 PANORAMA PT UNIT G LAFAYETTE CO 80026 Business Address 1751 PANORAMA PT UNIT G LAFAYETTE CO 80026 720-390-3880

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): July 30, 2021

urban-gro, Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation)

001-39933

(Commission File Number) (IRS Employer

46-5158469

Identification No.)

<u>1751 Panorama Point, Unit G</u> <u>Lafayette, Colorado 80026</u> (Address of principal executive offices) (Zip Code)

Registrant's telephone number, including area code (720) 390-3880

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

[] Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

[] Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

[] Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

[] Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of Each Class	Trading Symbol(s)	Name of Each Exchange on Which Registered
Common stock, par value \$0.001 per share	UGRO	Nasdaq

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company [X]

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. []

Item 1.01. Entry Into a Material Definitive Agreement.

The information set forth in Item 2.01 of this Current Report on Form 8-K regarding the Purchase Agreement (as such term is defined in Item 2.01 below) is incorporated herein by reference.

Item 2.01. Completion of Acquisition or Disposition of Assets.

As previously disclosed in the Current Report on Form 8-K filed by urban-gro, Inc., (the "Company") on June 29, 2021 the Company, urban-gro Architect Holdings, LLC, (the "Buyer"), the 2WRGA Shareholders, the 2WRCO Shareholders, the MJ12 Shareholders, the 2WRMS Shareholders (collectively, the "Sellers" and each a "Seller"), and Sam Andras, an individual (the "Seller Representative") entered into a Stock Purchase Agreement (the "Purchase Agreement"), pursuant to which the Buyer was to purchase all of the issued and outstanding capital stock of 2WR of Colorado, Inc., a Colorado corporation ("2WRCO"), 2WR of Georgia, Inc., a Georgia corporation ("2WRGA"), and MJ12 Design Studio, Inc., a Colorado corporation ("MJ12").

On July 30, 2021 the Company closed the Purchase Agreement (the "Closing") and entered into an Amendment No. 1 to the Purchase Agreement to implement various amendments to the Purchase Agreement (the "Amendment").

For more information on the Purchase Agreement, see the June 29, 2021 Form 8-K, which is incorporated into this Item 2.01 by reference. The Purchase Agreement and Amendment are filed as Exhibit 2.1 and 2.2, respectively, to this Current Report on Form 8-K and are incorporated herein by reference.

Item 3.02. Unregistered Sales of Equity Securities.

On July 30, 2021, in connection with the Closing, the Company issued 202,066 unregistered shares of Company common stock to the Sellers, of which 75,775 shares were escrowed for a period of up to twelve months to satisfy indemnification obligations in the Purchase Agreement, if necessary.

The issuances of the Company's common stock in connection with the foregoing were not registered under the Securities Act of 1933, as amended (the "Securities Act"), in reliance of the exemption from registration provided by Section 4(a)(2) of the Securities Act because such issuances did not involve a public offering.

Item 7.01. Regulation FD Disclosure.

On August 2, 2021 the Company issued a press release announcing the Closing. A copy of this press release is filed as Exhibit 99.1 hereto and is incorporated herein by reference.

The information in this Current Report on Form 8-K under Item 7.01 is being "furnished" and not "filed" with the SEC for purposes of Section 18 of the Exchange Act, or otherwise subject to the liabilities under such section. Furthermore, such information shall not be deemed incorporated by reference in any filing under the Securities Act, or the Exchange Act, unless specifically identified as being incorporated therein by reference.

Item 9.01. Financial Statements and Exhibits.

(a) Financial Statements of Businesses Acquired.

(i) The audited consolidated financial statements of 2WR of Colorado, Inc., 2WR of Georgia, Inc., and MJ12 Design Studio, Inc. as of December 31, 2020 and 2019 and the years ended December 31, 2020 and 2019, including the notes related thereto, are filed as Exhibit 99.2 and incorporated herein by reference.

(ii) The Company will file the unaudited consolidated financial statements of 2WR of Colorado, Inc., 2WR of Georgia, Inc., and MJ12 Design Studio, Inc. as of March 31, 2021 and for the three months ended March 31, 2021, including the notes related thereto as part of an amendment to this Current Report on Form 8-K no later than 71 calendar days after the required filing date for this Current Report on Form 8-K.

(b) Pro Forma Financial Information.

The Company will file the unaudited pro forma condensed combined financial information of urban-gro, Inc. and 2WR of Colorado, Inc., 2WR of Georgia, Inc., and MJ12 Design Studio, Inc. as of March 31, 2021 and December 31, 2020 and for the three months ended March 31, 2021and the year ended December 31, 2020, including the notes related thereto as part of an amendment to this Current Report on Form 8-K no later than 71 calendar days after the required filing date for this Current Report on Form 8-K.

(d) Exhibits.

Number	Exhibit
2.1	Stock Purchase Agreement
2.2	Amendment No. 1 to Stock Purchase Agreement
99.1	Press Release dated August 2, 2021
99.2	Audited consolidated financial statements of 2WR of Colorado, Inc., 2WR of Georgia, Inc., and MJ12 Design Studio, Inc. as of December 31, 2020 and 2019 and the years ended December 31, 2020 and 2019

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the Company has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

URBAN-GRO, INC.

Date: August 2, 2021

By: /s/ Bradley Nattrass

Bradley Nattrass Chief Executive Officer

Exhibit 2.1

EXECUTION VERSION

STOCK PURCHASE AGREEMENT

dated as of

June 28, 2021

by and among

URBAN-GRO ARCHITECT HOLDINGS, LLC,

AS THE BUYER

and

URBAN-GRO, INC,

AS THE PARENT

and

THE 2WRCO SHAREHOLDERS,

THE 2WRGA SHAREHOLDERS,

THE MJ12 SHAREHOLDERS, AND

THE 2WRMS SHAREHOLDERS

AS THE SELLERS

and

SAM ANDRAS

AS THE SELLER REPRESENTATIVE

STOCK PURCHASE AGREEMENT

THIS STOCK PURCHASE AGREEMENT (this "*Agreement*") is entered into this 28th day of June, 2021 by and among urbangro Architect Holdings, LLC, a Delaware limited liability company (the "*Buyer*"), urban-gro, Inc., a Delaware corporation ("*Parent*"), the 2WRGA Shareholders, the 2WRCO Shareholders, the MJ12 Shareholders, the 2WRMS Shareholders (collectively, the "*Sellers*" and each a "*Seller*"), and Sam Andras, an individual (the "*Seller Representative*", sometimes referred to herein collectively with the Sellers and the Buyer as the "*Parties*," and each, a "*Party*").

Recitals:

A. 2WR of Colorado, Inc., a Colorado corporation ("2WRCO"), 2WR of Georgia, Inc., a Georgia corporation ("2WRGA"), MJ12 Design Studio, Inc., a Colorado corporation ("MJ12") and 2WR of Mississippi, P.C., a Mississippi professional corporation

("2WRMS" and together with 2WRGA, 2WRCO and MJ12, the "Companies" and each individually, a "Company") are currently engaged in the Business;

B. Sam Andras, Scott Allen, Michael Starr, Sandra Schaaf, Emily Flournoy and Sara Flora (collectively, the "2WRGA Shareholders") are the holders of 100% of the capital stock of 2WRGA (collectively, the "2WRGA Shares"), which are the only issued and outstanding capital stock of 2WRGA;

C. Sam Andras, Scott Allen, Michael Starr, and Geoff McKinney (collectively, the "2WRCO Shareholders") are the holders of 100% of the capital stock of 2WRCO (collectively, the "2WRCO Shares"), which are the only issued and outstanding capital stock of 2WRCO;

D. Sam Andras, Scott Allen, Michael Starr and Geoff McKinney (the "*MJ12 Shareholders*") are the holders of 100% of the capital stock of MJ12 (collectively, the "*MJ12 Shares*" and together with the 2WRGA Shares and the 2WRCO Shares, the "*Purchased Shares*"), which are the only issued and outstanding capital stock of MJ12;

E. Sam Andras, Scott Allen, Michael Starr and Geoff McKinney (the "*2WRMS Shareholders*") are the holders of 100% of the capital stock of 2WRMS (collectively, the "*2WRMS Shares*"), which are the only issued and outstanding capital stock of 2WRMS;

F. The Sellers wish to sell the Purchased Shares to the Buyer, and the Buyer wishes to purchase the Purchased Shares from the Sellers, upon the terms and subject to the conditions set forth in this Agreement;

G. In connection with the acquisition of the Purchased Shares, the Buyer desires to affiliate with 2WRMS, and the Sellers desire to effect such affiliation in accordance with the terms set forth herein; and

H. The Parties desire to make certain representations, warranties, covenants and agreements in connection with the transactions contemplated by this Agreement.

Agreement:

NOW, THEREFORE, in consideration of the mutual agreements and covenants herein contained and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the Parties hereby agree as follows:

ARTICLE I: DEFINITIONS

1.1 <u>Definitions</u>. Except as otherwise provided herein, all capitalized terms used herein shall have the meanings assigned to them in <u>Exhibit A</u> to this Agreement.

ARTICLE II: PURCHASE AND SALE

2.1 <u>The Stock Purchase</u>. Upon the terms and subject to the conditions set forth in this Agreement, at the Closing, the Buyer hereby agrees to purchase from the Sellers all of the Purchased Shares and the Sellers hereby agree to sell to the Buyer all of the Purchased Shares, free and clear of any and all Liens, for the Purchase Price. At the Closing, an amount equal to \$7,100,000, as adjusted pursuant to Section 2.2(a) hereof, (the "*Initial Purchase Price*"), will be paid as follows:

(a) The Buyer shall pay, to the accounts designated in writing by those Persons entitled to repayment of the Funded Indebtedness (other than the Assumed Indebtedness) as set forth on a memorandum of funds flow detailing the payments to be made to the Parties, or to third parties on behalf of the Parties in connection herewith (the "*Flow of Funds Memorandum*"), by wire transfer of immediately available funds, an amount equal to the Companies' Funded Indebtedness owing to such Persons, which payments, in the aggregate, shall be sufficient to satisfy any and all obligations of the Companies to such Persons.

(b) The Buyer shall pay, to the accounts designated in writing by those Persons entitled to payment of the Transaction Fees as set forth on the Flow of Funds Memorandum, by wire transfer of immediately available funds, an amount equal to the Transaction Fees owing to such Persons, which payments shall be sufficient to satisfy any and all obligations of the Companies to such Persons. (c) Parent shall pay \$2,000,000 of the Initial Purchase Price on behalf of the Buyer through the issuance of such number of unregistered shares (the "*Closing Payment Shares*") of its common stock, par value \$0.001 ("*Parent Common Stock*") determined by dividing \$2,000,000 by the daily volume weighted average closing price (the "*Transaction Share Price*"), based on a trading day from 9:30 a.m. to 4:00 p.m., Eastern time and as reported by Bloomberg Financial LP, of the Parent Common Stock on the Exchange for a period consisting of ten (10) consecutive trading days ending on the Business Day prior to the Closing Date. The number of Closing Payment Shares shall be rounded to the nearest whole number. The Closing Payment Shares shall be issued in accordance with the rules of the Exchange and as follows:

(i) Closing Payment Shares (the "*Escrowed Shares*") having a value equal to \$750,000, as determined using the Transaction Share Price determined for the Closing, shall be delivered to the Escrow Agent pursuant to the terms and conditions of <u>Section 2.6</u> to secure the Sellers' indemnification obligations. The number of Escrowed Shares shall be rounded to the nearest whole number.

(ii) The remainder of the Closing Payment Shares shall be delivered to the Sellers on the Closing Date based upon the allocation set forth on <u>Schedule 2.1(c)(ii)</u> (the "*Sellers Allocation Schedule*").

(d) For each outstanding Company PPP Loan, the Buyer shall deposit the principal amount outstanding plus accrued interest by wire transfer of immediately available funds, into an escrow account established with the lender (each, a "*PPP Lender*"), pursuant to an escrow agreement with the applicable PPP Lender in form and substance reasonably acceptable to the Buyer and the Seller Representative (a "*PPP Escrow Agreement*").

(e) The Buyer shall pay to each Seller, by wire transfer of immediately available funds to those accounts designated by each Seller prior to Closing, such Seller's portion of the balance of the Initial Purchase Price (in such amounts as provided in the Flow of Funds Memorandum) after making the payments, issuances and deposits set forth in accordance with Sections 2.1(a)-(d).

2.2 Purchase Price Adjustment.

(a) Interim Working Capital. The Parties acknowledge and agree that the total aggregate Purchase Price to be paid by the Buyer hereunder is based on the assumptions that (x) the Companies' Working Capital as of the Closing shall be equal to \$650,000 (the "Working Capital Target") and (y) the Companies' Cash as of the Closing shall be equal to at least \$300,000 (the "Minimum Closing Cash"). Three (3) Business Days prior to the Closing Date, the Sellers shall provide a statement (the "Estimated Closing Statement") setting forth a good faith estimate of (i) the Companies' Working Capital as of the Closing Date (the "Estimated Vorking Capital"), (ii) the Companies' Cash as of the Closing Date (the "Estimated Assumed Indebtedness"). To the extent the Estimated Working Capital exceeds the Working Capital Target, the Initial Purchase Price payable by the Buyer at Closing shall be increased by the amount of such excess. To the extent the Minimum Closing Cash exceeds the Estimated Closing Cash, the Initial Purchase Price payable by the Buyer at Closing shall be decreased by the amount of such excess. To the extent the Minimum Closing Cash exceeds the Estimated Closing Capital Purchase Price payable by the Buyer at Closing shall be decreased by the amount of such excess. To the extent the Minimum Closing Cash exceeds the Estimated Closing Capital shall be decreased by the amount of such excess. To the extent the Minimum Closing Cash exceeds the Estimated Closing Capital shall be decreased by the Buyer at Closing shall be decreased by the amount of such excess. The Initial Purchase Price payable by the Buyer at Closing shall be decreased by the amount of such excess. The Initial Purchase Price payable by the Buyer at Closing shall be decreased by the amount of such excess. The Initial Purchase Price payable by the Buyer at Closing shall be decreased by the amount of such excess. The Initial Purchase Price payable by the Buyer at Closing shall be decreased by the amount of such excess. The Initial Purchase Price payable by

3

(b) <u>Baseline Closing Statement</u>. Within sixty (60) calendar days after the Closing Date, the Buyer shall deliver to the Seller Representative a statement (the "*Baseline Closing Statement*") setting forth the Companies' (i) consolidated balance sheet as of the Closing, (ii) Working Capital as of the Closing, (iii) Cash as of the Closing, (iv) Assumed Indebtedness as of the Closing and (v) the resulting calculation of the final Purchase Price. Without limiting the foregoing, the Baseline Closing Statement shall be prepared using the same accounting methods, policies, practices and procedures, with consistent classifications and estimation methodologies, as were used in the preparation of the Sample Working Capital Calculation.

(c) Review and Dispute of Baseline Closing Statement. The Seller Representative shall have thirty (30) days after the Buyer's delivery to the Seller Representative of the Baseline Closing Statement to deliver written notice to the Buyer (the "Dispute Notice") setting forth in reasonable detail any and all items of disagreement related to the Baseline Closing Statement. Unless the Seller Representative shall have delivered a Dispute Notice during the required thirty (30) day period, the Baseline Closing Statement delivered by the Buyer shall be final and binding upon the Parties. If the Seller Representative timely delivers a Dispute Notice to the Buyer, the Parties shall negotiate in good faith and use their respective commercially reasonable efforts to resolve the differences concerning any items of disagreement and any items agreed upon shall be final and binding upon the Parties. If any such items remain unresolved for a period of fifteen (15) days after the Buyer's receipt of the Dispute Notice (unless such time frame is mutually extended by the Parties), the Buyer and the Seller Representative shall submit the disputed items for final and binding resolution by an independent certified public accountant (the "Accounting Firm") selected by the mutual agreement of the Buyer and the Seller Representative. If the Buyer and the Seller Representative cannot agree on the Accounting Firm after ten (10) days of discussion, the Buyer and the Seller Representative shall each select a certified public accountant (each an "Accountant Representative") and the Accountant Representatives shall select the Accounting Firm by mutual agreement. The Buyer and the Seller Representative shall each be solely responsible for all costs and fees relating to their Accountant Representative. The Accounting Firm will only consider those items and amounts set forth in the Baseline Closing Statement as to which the Buyer and the Seller Representative have disagreed on the terms specified above and not resolved during the fifteen (15) day period (or longer, if mutually extended) described above and must resolve the disputes using the same accounting methods, policies, practices and procedures, with consistent classifications and estimation methodologies, as were used in the preparation of the Sample Working Capital Calculation. The Buyer and the Seller Representative shall request that the Accounting Firm render a determination as to any disputed items within twenty (20) days after its retention, and the Parties shall cooperate fully with the Accounting Firm so as to enable it to make such determination as quickly and as accurately as practicable. The Accounting Firm shall make its determination based exclusively on presentations and supporting material provided by the Parties and not pursuant to any independent review and such determination as to any disputed items shall be made by the Accounting Firm within the dollar ranges set forth in the version of the Baseline Closing Statement delivered by the Buyer and the Dispute Notice delivered by the Seller Representative. The Accounting Firm shall deliver a version of the Baseline Closing Statement incorporating its determinations to the Parties, which shall be final and binding upon the Parties. The fees, costs and expenses of the Accounting Firm shall be allocated to and paid by the Buyer and the Sellers based upon the percentage which the portion of the contested amounts not awarded to the Sellers bears to the amounts actually contested by the Seller Representative, as determined by the Accounting Firm. Other than such fees, costs and expenses of the Accounting Firm, the Parties shall each be responsible for their own costs and expenses incurred in connection with any actions taken pursuant to this Section 2.2(c). Any final closing statement determined in accordance with this Section 2.2(c) is referred to herein as the "Definitive Closing Statement."

4

(d) <u>Purchase Price Adjustment</u>. Following the final determination of the Definitive Closing Statement in accordance with <u>Section 2.2(c)</u>: (i) if the final Purchase Price as set forth on the Definitive Closing Statement (the "*Final Purchase Price*") exceeds the Initial Purchase Price, the Buyer shall pay to the Sellers the amount by which the Final Purchase Price exceeds the Initial Purchase Price; (ii) if the Initial Purchase Price exceeds the Final Purchase Price (the "*Excess Payment*") the Sellers shall pay to the Buyer an amount equal to the Excess Payment the ("*Excess Amount*") pursuant to the terms of <u>Section 2.2(e)</u>. Any payments required to be paid by the Buyer by this <u>Section 2.2(d)</u> shall be paid within five (5) Business Days of the relevant determination. For the avoidance of doubt, the Parties agree and acknowledge that any Purchase Price adjustment will not preclude any Buyer Indemnified Party from recovering any indemnifiable Losses in accordance with Article VIII.

(e) <u>Set-off Against Earnout</u>. In the event of an Excess Payment, an amount equal to 12.5% of the Excess Amount shall be withheld and set off by the Buyer against each of the Earnout Payments payable to the Sellers until the Excess Amount has been paid in full.

2.3 The Closing.

(a) Subject to the terms and conditions of this Agreement, the closing of the purchase and sale of the Purchased Shares contemplated by this Agreement (the "*Closing*") shall take place via the electronic exchange of documents, and shall be held no later than three (3) Business Days after the last of the conditions to Closing set forth in <u>Article X</u> have been satisfied or waived (other than conditions which, by their nature, are to be satisfied on the Closing Date), or at such other time or on such other date or at such other place as Seller Representative and the Buyer may mutually agree upon in writing (the day on which the Closing takes place being "*Closing Date*"). The Closing shall be deemed to be effective as of 12:01 AM Eastern Time on the Closing Date.

(b) At the Closing, simultaneously with the actions set forth in <u>Sections 2.1</u> and <u>2.4</u>, the Sellers will assign and transfer to the Buyer all of the Sellers' right, title and interest in and to the Purchased Shares.

2.4 Closing Deliveries.

(a) At the Closing, the Sellers shall deliver or cause to be delivered to the Buyer:

(i) stock certificates representing the Purchased Shares, accompanied by stock powers for the Purchased Shares, duly executed by the applicable Seller;

(ii) payoff letters, in form and substance reasonably satisfactory to the Buyer, as to the full payment of all Funded Indebtedness (other than the Assumed Indebtedness), and the release of the Companies from further Liability related thereto and the release of any Liens related thereto, each effective upon payment pursuant to <u>Section 2.1(b)</u> of the applicable Funded Indebtedness specified on the Flow of Funds Memorandum;

5

(iii) an appropriate receipt and release in a form and substance acceptable to the Buyer from each Person entitled to Transaction Fees, acknowledging that such Person has received all amounts due such Person from the Companies, and releasing the Companies from any further Liability with respect thereto, each effective upon payment pursuant to <u>Section</u> 2.1(c) of the applicable Transaction Fees specified on the Flow of Funds Memorandum;

(iv) an employment agreement with each of the Key Employees, duly executed by the applicable Key Employee, in form and substance reasonably satisfactory to the Buyer (the "*Employment Agreements*");

(v) a management and administrative services agreement by and among the Buyer, 2WRMS and the 2WRMS Shareholders, duly executed by 2WRMS and the 2WRMS Shareholders in form and substance reasonably satisfactory to the Buyer (the "*Management Services Agreement*");

(vi) a share transfer agreement in form and substance reasonably satisfactory to the Buyer, duly executed by each of the 2WRMS Shareholders (the "*Share Transfer Agreement*");

(vii) an asset transfer agreement by and among the Buyer and the Companies, duly executed by each of the Companies in form and substance reasonably satisfactory to the Buyer (the "Asset Transfer Agreement") to transfer certain assets and liabilities between the Buyer and the Companies as set forth therein;

(viii) an escrow agreement in form and substance reasonably satisfactory to the Buyer, duly executed by the Shareholder Representative (the "*Escrow Agreement*");

(ix) written resignations of all members of each Company's board of directors and officers;

(x) a good standing certificate for each Company from the state of its organization and the Secretary of State in each other jurisdiction in which such Company is required to qualify to do business as a foreign corporation, in each case dated as of a date as near as reasonably practicable to the Closing Date;

(xi) the written consents, approvals, waivers, notices or similar authorizations required to be obtained or given by any Person in order to consummate the transactions contemplated by the Transaction Documents, including those consents, approvals, waivers, notices or similar authorizations set forth on <u>Schedule 2.4(a)(x)</u>, in form and substance reasonably acceptable to the Buyer;

(xii) evidence satisfactory to the Buyer that the Companies 401(k) Plan and the other Employee Benefits Plan have been terminated;

6

(xiii) evidence satisfactory to the Buyer of payment in full of all Transaction Fees effective as of the Closing;

(xiv) an investment questionnaire from each Seller in form and substance satisfactory to the Buyer;

(xv) an agreement, in form and substance reasonably satisfactory to the Buyer, terminating the Wenzel Phantom Share Agreement, duly executed by 2WRCO and Darcey Faye Wenzel (the "Wenzel Phantom Share Termination Agreement");

(xvi) for each outstanding Company PPP Loan at signing, either (i) a PPP Escrow Agreement, duly executed by the Seller Representative, the applicable PPP Lender and the applicable Company, along with evidence reasonably satisfactory to the Buyer that a loan forgiveness application has been submitted to the PPP Lender for such Company PPP Loan or (ii) evidence reasonably satisfactory to the Buyer that each of the loan forgiveness applications for Company PPP Loans outstanding at signing has been forgiven by the U.S. Small Business Administration;

(xvii) no less than three (3) Business Days prior to the Closing Date, <u>Schedule A</u> setting forth the Companies' Funded Indebtedness included in Assumed Indebtedness as of the Closing Date;

(xviii) no less than three (3) Business Days prior to the Closing Date, the Estimated Closing Statement, including (A) the Estimated Working Capital, (B) the Estimated Closing Cash and (C) the Estimated Assumed Indebtedness; and

(xix) such other documents or instruments in form and substance reasonably acceptable to the Buyer as the Buyer may deem reasonably necessary or as may be required to consummate the transactions contemplated hereby.

(b) At the Closing, the Buyer shall deliver or cause to be delivered to the Sellers:

(i) the cash balance of the Initial Purchase Price;

(ii) the Closing Payment Shares (less the Escrowed Shares);

(iii) evidence reasonably satisfactory to the Sellers that the Buyer has made the payments required by <u>Section</u>

<u>2.1;</u>

(iv) each of the Employment Agreements, duly executed by an authorized officer of the Buyer;

(v) the Management Services Agreement, duly executed by the Buyer;

(vi) the Share Transfer Agreement, duly executed by the Buyer;

(vii) the Asset Transfer Agreement, duly executed by the Buyer; and

(viii) the Escrow Agreement, duly executed by the Buyer.

7

(c) At the Closing, the Buyer shall also deliver, or cause to be delivered, the Escrowed Shares to the Escrow Agent.

2.5 Earnout.

(a) Definitions.

(i) As used herein, the term "Calculation Period" means each quarter of the Earnout Period.

(ii) As used, the term "*Change of Control*" means any of the following events: (A) substantially all of the assets of the Buyer or the Transferred Entities are sold to one (1) or more Persons who were not affiliated with the Buyer or Parent immediately before such sale; (B) fifty percent (50%) or more of the capital stock having voting rights in the Buyer are sold to or exchanged with one or more Persons who were not shareholders or Affiliates of the Buyer, respectively, immediately before such sale, in a transaction or series of related transactions that results in the fifty percent (50%) or more threshold being crossed; (C) fifty percent (50%) or more of the capital stock having voting rights in each of the Transferred Entities are, in a transaction or series of related transactions

that results in the fifty percent (50%) or more threshold being crossed, sold to or exchanged with one or more Persons who were not shareholders or Affiliates of the Buyer, respectively, immediately before such sale, such that the majority of the voting rights of each of the Transferred Entities are held by one or more Persons who were not shareholders or Affiliates in the Buyer immediately before such sales, and (D) the Buyer is merged with or converted into another Person and 50% or more of the voting or management rights in the surviving or converted Person are held by one or more Persons who were not shareholders or Affiliates in the Buyer immediately before such surviving or converted Person are held by one or more Persons who were not shareholders or Affiliates in the Buyer immediately before such surviving or converted Person are held by one or more Persons who were not shareholders or Affiliates in the Buyer immediately before such surviving or converted Person are held by one or more Persons who were not shareholders or Affiliates in the Buyer immediately before such surviving or converted Person are held by one or more Persons who were not shareholders or Affiliates in the Buyer immediately before such merger.

(iii) As used herein, the term "*Earnout Period*" means the period consisting of the first eight (8) full quarters of the Companies following the Closing Date.

(iv) As used herein, the term "*Quarterly Gross Profit*" means an amount equal to (A) the revenue of the Companies on a consolidated basis *minus* (B) the direct costs of such revenue including, but not limited to, the appropriate allocation of the cost of operational staff compensation as reasonably determined by the Buyer, consistent with the Companies' historical practices and determined in accordance with GAAP.

(b) <u>Earnout Payment</u>. As additional consideration for the Purchased Shares, the Buyer shall pay to the Sellers in accordance with respect to each Calculation Period within the Earnout Period an amount, if any, (each, an "*Earnout Payment*") equal to twenty percent (20%) of the Companies' Quarterly Gross Profit for such Calculation Period after the end of each such Calculation Period; <u>provided</u>, <u>however</u>, that in no event shall the Buyer be obligated to pay the Sellers more than \$2,000,000 in the aggregate for all Calculation Periods during the Earnout Period.

8

(c) Procedures Applicable to Determination of the Earnout Payments.

(i) On or before the date which is forty-five (45) days after the last day of each Calculation Period, the Buyer shall prepare and deliver to the Seller Representative a written statement (in each case, an "*Earnout Calculation Statement*") setting forth in reasonable detail its determination of Quarterly Gross Profit for the applicable Calculation Period and its calculation of the resulting Earnout Payment (in each case, an "*Earnout Calculation*").

(ii) Seller Representative shall have thirty (30) days after receipt of the Earnout Calculation Statement for each Calculation Period (in each case, the "Review Period") to review the Earnout Calculation Statement and the Earnout Calculation set forth therein. During the Review Period, Seller Representative and its accountants shall have the right to inspect the Companies' books and records during normal business hours at the Companies' offices, upon reasonable prior notice and solely for purposes reasonably related to the determinations of the Companies' Quarterly Gross Profit and the resulting Earnout Payment and, upon reasonable request, Buyer shall provide access to a data room hosting electronic copies of such relevant records to the Seller Representative and its accountants request. The Companies shall promptly furnish to the Seller Representative and its accountant any information reasonably requested by them regarding the Companies' revenues, costs and expenses. Prior to the expiration of the Review Period, which Review Period shall be automatically extended for an additional ten (10) days upon the delivery of written notice from Seller Representative to Buyer prior to the expiration of the Review Period, Seller Representative may object to the Earnout Calculation set forth in the Earnout Calculation Statement for the applicable Calculation Period by delivering a written notice of objection (an "Earnout Calculation Objection Notice") to the Buyer. Any Earnout Calculation Objection Notice shall specify the items in the applicable Earnout Calculation disputed by Seller Representative and shall describe in reasonable detail the basis for such objection, as well as the amount in dispute. If Seller Representative fails to deliver an Earnout Calculation Objection Notice to the Buyer prior to the expiration of the Review Period, as may be extended as set forth herein, then the Earnout Calculation set forth in the Earnout Calculation Statement shall be final and binding on the Parties hereto.

(iii) If Seller Representative timely delivers an Earnout Calculation Objection Notice, the Buyer and Seller Representative shall negotiate in good faith to resolve the disputed items and agree upon the resulting amount of the Companies' Quarterly Gross Profit and the Earnout Payment for the applicable Calculation Period. If the Buyer and Seller Representative are unable to reach agreement within thirty (30) days after such an Earnout Calculation Objection Notice has been given, any unresolved disputes shall be promptly submitted for final and binding resolution by the Accounting Firm. The Buyer and the Seller Representative shall request that the Accounting Firm render a determination as to any disputed described in the Earnout Calculation Objection Notice that remain within twenty (20) days after its retention, and the Parties shall cooperate fully with the Accounting Firm so as to enable it to make such determination as quickly and as accurately as practicable. The Accounting Firm shall make its determination based exclusively on presentations and supporting material provided by the Parties and not pursuant to any independent review and such determination as to any disputed items shall be made by the Accounting Firm within the dollar ranges set forth by the parties. The Accounting Firm shall

deliver a version of the Earnout Calculation Statement incorporating its determinations to the Parties, which shall be final and binding upon the Parties. The fees and expenses of the Accounting Firm in resolving all such disputed items shall be borne fifty percent (50%) by the Sellers and fifty percent (50%) by the Buyer.

(d) Independence of Earnout Payments. The Buyer's obligation to pay each of the Earnout Payments to the Sellers in accordance with Section 2.5(a) is an independent obligation of the Buyer and is not otherwise conditioned or contingent upon the satisfaction of any conditions precedent to any preceding or subsequent Earnout Payment and the obligation to pay an Earnout Payment to the Sellers shall not obligate the Buyer to pay any preceding or subsequent Earnout Payment. For the avoidance of doubt and by way of example, if the conditions precedent to the payment of the Earnout Payment for the first Calculation Period are not satisfied, but the conditions precedent to the payment for the second Calculation Period are satisfied, then the Buyer would be obligated to pay such Earnout Payment for the second Calculation Period for which the corresponding conditions precedent have been satisfied, and not the Earnout Payment for the first Calculation Period.

(e) <u>Timing of Payment of Earnout Payments</u>. Subject to <u>Section 2.5(g)</u>, any Earnout Payment that the Buyer is required to pay pursuant to <u>Section 2.5(a)</u> hereof shall be paid in full no later than five (5) Business Days following the date upon which the determination of the Companies' Quarterly Gross Profit for the applicable Calculation Period becomes final and binding upon the parties as provided in <u>Section 2.5(b)</u> (including any final resolution of any dispute raised by Seller Representative in an Earnout Calculation Objection Notice). Each Earnout Payment shall be delivered to the Sellers based upon the allocation set forth on the Sellers Allocation Schedule. The Buyer shall have the option, in its sole discretion, to pay each of the Earnout Payments in cash or through the issuance of shares of Parent Common Stock (such shares, the "*Earnout Shares*" and together with the Closing Payment Shares, the "*Parent Shares*"). The number of Earnout Shares to be issued for any Calculation Period shall be determined by dividing the amount of the Earnout Payment for such Calculation Period to be paid in Earnout Shares by the Transaction Share Price determined for the period consisting of the ten (10) consecutive trading days ending on the Business Day prior to the date of payment. Any cash payments shall be made by wire of immediately available funds to the accounts set forth on the Flow of Funds Memorandum or to such accounts of the Sellers as are designated in writing by the applicable Seller pursuant to <u>Section 12.1</u>. Any portion of the Earnout Payment paid to the Sellers pursuant to this <u>Section 2.5</u> shall be treated as additional Purchase Price.

(f) <u>Right of Set Off.</u> The Buyer shall have the right to withhold and set off against any amount otherwise due to be paid pursuant to this <u>Section 2.5</u> against any amount owed to the Buyer by the Sellers or the Companies pursuant to the terms hereof, provided that the amount of such set off has been determined pursuant to a final judgment or mutual agreement of the Parties.

(g) <u>No Security</u>. The Parties hereto understand and agree that (i) the contingent rights to receive any Earnout Payment shall not be represented by any form of certificate or other instrument, are not transferable except by operation of Laws relating to descent and distribution, divorce and community property, and do not constitute an equity or ownership interest in Buyer or any of the Companies, (ii) no Seller shall have any rights as a securityholder of the Buyer or any of the Companies as a result of such Seller's contingent right to receive any Earnout Payment hereunder, and (iii) no interest is payable with respect to any Earnout Payment.

10

(h) Operation of the Companies. The Sellers agree and acknowledge that, except as expressly provided to the contrary in this Section 2.5, (i) during the term of the Earnout Period, the Buyer has the right to operate the Companies (and all other components of the Buyer's business) in any way that the Buyer deems appropriate in the Buyer's sole discretion, including making changes based upon market conditions, the requirements of its lenders, or other circumstances, even if such operations or changes may have an adverse effect on the Quarterly Gross Profit of the Companies; (ii) the Buyer has no obligation to operate the Companies (or any other component of the Buyer's business) in a manner designed to achieve any specific level of Quarterly Gross Profit; (iii) the Buyer is not obligated to operate the Companies in a manner consistent with the manner the Sellers operated the Companies prior to the Closing Date; (iv) neither the Buyer nor any of its Affiliates makes any representations, warranties or covenants to the Sellers as to the amount or type of funding which may be provided to the Companies; (v) the potential Earnout Payment is speculative in nature and subject to numerous factors outside the Buyer's control; (vi) the Buyer owes no fiduciary duty to the Sellers; and (vii) the Parties solely intend the express provisions of this Agreement to govern their contractual relationship; provided that, during the Earnout Period, the Buyer shall (except with the prior written consent of the Seller Representative which may only be withheld to protect the legitimate interests of the Sellers):

(i) Act in good faith, and shall not knowingly take, and shall cause the Companies not to knowingly take, any action with respect to the business, operations or affairs of the Companies with the sole intent or purpose of reducing the Quarterly Gross Profit of the Companies;

(ii) Use commercially reasonable efforts to devote sufficient resources to the Companies to allow them to operate in a manner materially consistent with Sellers' Ordinary Course of Business prior to the Closing;

(iii) Calculate, and cause the Companies to calculate, the Quarterly Gross Profit in accordance with GAAP; and

(iv) Use commercially reasonable efforts to provide the Companies with adequate working capital to fund the operations of the Business reasonably consistent with past practices and commercially reasonable business practices.

(i) Earnout Payment Upon a Change of Control.

(i) Notwithstanding the foregoing and anything in this Agreement to the contrary, if a Change of Control occurs during the Earnout Period, no later than ninety (90) days after the date of such Change of Control, the Buyer shall pay to the Sellers, in full satisfaction of the Buyer and Parent's obligations pursuant to this <u>Section 2.5</u>, an amount equal to Two Million Dollars (\$2,000,000) *less* the value of any Earnout Payments actually paid to the Sellers prior to the date of such payment. Such payment may be made in cash or through the issuance of Earnout Shares pursuant to the terms of this <u>Section 2.5</u>. Parent shall be jointly and severally liable with the Buyer for any such payment owed pursuant to this <u>Section 2.5(i)(i)</u>.

11

(ii) Nothing herein shall limit or otherwise affect the Buyer, Parent or any of its Affiliate's ability to effect a Change of Control or to sell or transfer all or any portion of the assets or business of the Companies or all or any portion of the assets or capital stock of the Buyer, Parent or any of its Affiliates (whether by sale of stock, merger, consolidation, reorganization or otherwise).

2.6 Escrowed Shares.

(a) At Closing, the Buyer shall issue and deliver to the Escrow Agent the Escrowed Shares. For a period of twelve (12) months after the Closing Date, the Escrow Agent shall hold the Escrowed Shares in escrow pursuant to the terms of the Escrow Agreement. On the first (1^{st}) anniversary of the Closing Date, the Escrow Agent shall release the Escrowed Shares in the manner described in the Escrow Agreement.

(b) For so long as any Escrowed Shares remain subject to escrow pursuant to the Escrow Agreement, each Seller hereby grants to and appoints the Chief Financial Officer of the Buyer, solely in his capacity as an officer of the Buyer, and any individual who shall hereafter succeed to such office of the Buyer, and any other designee of the Buyer in its sole discretion (in each case, the "Attorneyin-Fact"), each of them individually, as such Seller's proxy and Attorney-in-Fact (with full power of substitution) to vote (in person or by attorney or proxy) or act by written consent with respect to the Escrowed Shares on every matter for which a vote of the Buyer's shareholders is taken, whether by meeting, written consent or otherwise. Each Seller will deliver or cause to be delivered to the Attorneyin-Fact, at the request of the Buyer, duly executed by such Seller, all proxies from time to time necessary or appropriate to so enable the Attorney-in-Fact to vote such Escrowed Shares. Each Seller shall promptly deliver or cause to be delivered to the Attorney-in-Fact, at the request of the Buyer, such further proxies, powers of attorney and other instruments and documents as may be necessary in the opinion of the Buyer to give effect to the rights granted hereunder. This proxy is coupled with an interest, shall be irrevocable in respect of a Seller until such time as all of such Seller's Escrowed Shares have been released from escrow pursuant to Section 2.6(b), shall, to the fullest extent permitted by applicable Law, continue despite the mental incompetence or other legal incapacity or termination of the existence of the Seller, shall survive the death or disability of the Seller, and shall bind the respective heirs, executors, administrators, other legal representatives and permitted assigns of the Seller. Each Seller hereby revokes any proxy previously granted by such Seller with respect to the Escrowed Shares. If a Seller fails to so execute and deliver any such forms of proxy and/or other documents, then the Attorney-in-Fact be and is hereby irrevocably nominated, constituted and appointed the attorney of such Seller to execute and deliver on account of the Seller such forms of proxy and/or other documents that may be required, from time to time, to enable the Attorney-in-Fact to vote the Seller's Escrowed Shares.

(c) The Attorney-in-Fact shall act only in accordance with the instructions of the board of directors of the Buyer, in the board of directors' sole and absolute discretion without reference to any other parties (and based on their sole determination of what would be in the best interests of the Buyer), and if no such instructions are received by the Attorney-in-Fact with respect to any particular matter in respect of which the vote or consent of the shareholders of the Buyer is requested, the Escrowed Shares that are, at such time,

subject to this Agreement shall not be voted nor shall any written consent or resolution be signed by the Attorney-in-Fact in respect thereof.

(d) The Attorney-in-Fact shall exercise its rights and duties hereunder in good faith, but: (i) shall incur no Liability except for its fraud or willful misconduct; (ii) may rely on all documents and notices delivered to it in connection with its appointment as Attorney-in-Fact; and (iii) may act on the advice of any qualified expert (including the Buyer's legal counsel and auditor) and shall be fully justified in acting (or declining to act) in accordance with such advice, and the Attorney-in-Fact shall not be responsible, save as aforesaid, for any loss, destruction, depreciation, damage or other consequence occasioned by its acting (or declining to act) in accordance with any such advice. The Attorney-in-Fact shall not be personally liable or responsible for the payment of the fees of any such expert or for other costs or expenses in case of any litigation arising under or in connection with its appointment as Attorney-in-Fact or the exercise of its powers in such role. Each Seller hereby releases the Attorney-in-Fact and the Buyer from any and all Losses, whether arising, directly or indirectly, by way of statute, contract, tort or otherwise in connection with its appointment as Attorney-in-Fact for anything whatsoever, other than as a result of the fraud or willful misconduct of the Attorney-in-Fact.

ARTICLE III: REPRESENTATIONS AND WARRANTIES CONCERNING THE COMPANIES

The Sellers hereby jointly and severally represent and warrant to the Buyer as of the Closing Date, as follows:

3.1 Existence: Good Standing. Each of the Companies is a corporation duly formed, validly existing and in good standing under the laws of the state of its organization and has all requisite power and authority to own and operate its properties and to conduct its Business, as now conducted. Schedule 3.1 sets forth each Company's jurisdiction of organization, the other jurisdictions in which it is qualified to do business, and its directors and officers. Each Company is duly licensed or qualified to do business as a foreign corporation and is in good standing under the laws of each other jurisdiction under which such licensing or qualification is necessary. The Sellers have delivered to the Buyer correct and complete copies of each Company's Organizational Documents. Except as set forth on Schedule 3.1, no Company has, during the five (5) years preceding the date of this Agreement, been known by or used any other corporate, trade or fictitious name or acquired all or substantially all of the assets, capital stock or operating units of any Person. No Company has, during the five (5) years preceding the date business location at any address other than the addresses set forth on Schedule 3.1.

3.2 Capitalization; Subsidiaries.

(a) The records of each Company reflect that the applicable Sellers are the record owners of one hundred percent (100%) of the capital stock of such Company. <u>Schedule 3.2</u> sets forth the capitalization of each Company. All of the issued capital stock of each Company is duly authorized, validly issued, fully paid and nonassessable, and none of it was issued in violation of any pre-emptive rights, rights of first offer or first refusal or similar rights, or in violation of the Securities Act or any other applicable securities law. Other than the Wenzel Phantom Share Agreement, there are no authorized or outstanding (i) options, warrants, calls, rights of first refusal or other rights of any character to acquire equity or debt interests from any Company or any phantom stock, stock appreciation rights or other contract rights intended to provide an economic return based on changes in the value of any debt or equity securities of any Company, (ii) rights or Contracts to which any Company is a party requiring, or convertible securities convertible into equity interests of such Company, or (iii) contract rights or options pursuant to which any Company is required to or has the right to redeem, purchase or otherwise reacquire any equity securities, or other instrument convertible or exercisable into equity securities, of such Company. There are no Contracts to which any Company is a party or to which any other Person is a party with respect to: (x) the voting of any capital stock of such Company (including any proxy or director or manager nomination rights); (y) the transfer of or transfer restrictions on any equity interests of such Company; or (z) the granting of any other rights in respect of any equity interests of such Company (*e.g.*, stock appreciation rights).

(b) No Company owns or controls any equity security or other interest of any other corporation, partnership, limited liability company or other business entity. No Company is a participant in any joint venture, partnership, limited liability company or similar arrangement. No Company has consolidated or merged with, acquired all or substantially all of the assets of, or acquired the stock of or any interest in any corporation, partnership, limited liability company or other business entity since its inception.

3.3 <u>No Conflict</u>. The consummation of the transactions contemplated by this Agreement and the other Transaction Documents, or compliance by any Company, the Seller Representative or the Sellers with any of the provisions thereof will not conflict with, or result in any violation or breach of, or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation or to loss of a material benefit under, or give rise to any obligation of any Seller or any Company to make any payment under, or to the increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or result in the creation of any Company; (b) any Material Contract or Permit; (c) any Order; or (d) any applicable Law. Except as set forth on <u>Schedule 3.3</u>, no consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Entity is required on the part of any Seller or any Company in connection with (i) the execution and delivery of this Agreement and the other Transaction Documents, the compliance by the Sellers, the Seller Representative and each Company with any of the provisions hereof and thereof, or the consummation of the transactions contemplated hereby and thereby, or (ii) the continuing validity and effectiveness immediately following the Closing of any Company Permit or novation agreement under the Federal Acquisition Regulation.

3.4 Financial Statements; Absence of Undisclosed Liabilities.

(a) Attached as <u>Schedule 3.4(a)</u> are correct and complete copies of: (i) the unaudited balance sheets of the Companies on a consolidated basis for each of the years 2020 and 2019, and statements of income, shareholders' equity and cash flow for each of the fiscal years then ended (the "*Financial Statements*"). Except as set forth on <u>Schedule 3.4(a)</u>, each of the Financial Statements has been prepared in accordance with GAAP, consistently applied, by the Companies without modification of the accounting principles used in the preparation thereof throughout the periods presented and presents fairly, in all material respects, the consolidated financial position, results of operations and cash flows of the Companies as at the dates and for the periods indicated therein.

14

(b) As of the date of this Agreement, no Company has any Liabilities other than: (i) Liabilities disclosed on the Financial Statements; (ii) accounts payable and accrued expenses incurred after December 31, 2020 in the Ordinary Course of Business (none of which results from, arises out of, relates to, is in the nature of, or was caused by any breach of Contract, breach of warranty, tort, infringement or violation of Law).

3.5 <u>Absence of Certain Changes</u>. Since December 31, 2020, there has not been any event or change in the Business which has had or would reasonably be expected to have a Material Adverse Effect. Except as set forth in <u>Schedule 3.5</u>, since December 31, 2020, the Companies have conducted the Business and its affairs only in the Ordinary Course of Business, and no Company has:

(a) entered into, amended or terminated any Material Contract, outside of the Ordinary Course of Business;

(b) made any capital expenditure(s) or entered into any commitment to make any capital expenditure, in each case that involves more than \$25,000 individually or \$50,000 in the aggregate;

(c) changed its authorized or issued capital stock or other equity interests; declared or paid any dividend or made any distribution or payment in respect of its capital stock or other equity interests; split, combined or reclassified its capital stock or other equity interests;

(d) issued, sold or otherwise disposed of any of its securities, or granted any options, warrants or other rights to acquire (including upon conversion, exchange or exercise) any of its securities or declared, set aside, made or paid any dividend or distribution with respect to its capital stock (whether in cash or in kind) or redeemed, purchased or otherwise acquired any of its securities or amended any of its Organizational Documents;

(e) paid any bonuses, or increased any salaries or other compensation, to any of its directors, officers, employees or independent contractors;

(f) entered into or amended any employment, severance or similar Contract with any director, officer or other employee of such Company or granted any increase in the base compensation of any director, officer or employee or entered into any collective bargaining agreement;

(g) made any loan or advance to any Person, other than the extension of trade credit in the Ordinary Course of Business and advances to officers and other employees in excess of \$1,000;

(h) entered into any transaction that would be required to be disclosed under <u>Section 3.17</u>;

(i) adopted, increased, accelerated or modified the schedule of payments or benefits under any Employee Benefit Plan for or with any of the directors, officers, employees or independent contractors of such Company, except in the Ordinary Course of Business;

15

(j) sustained any material damage to or destruction or loss of any material property owned or used by such Company, whether or not covered by insurance, or waived or released any right of material value;

(k) incurred any material Indebtedness or redeemed, retired or prepaid any Indebtedness for borrowed money, other than trade payables incurred in the Ordinary Course of Business and Funded Indebtedness being paid off at Closing, or guarantees of the obligations of another Person;

(l) sold, leased, encumbered or otherwise transferred any tangible or intangible assets other than in the Ordinary Course of Business and for fair consideration;

(m) acquired or agreed to acquire by merging or consolidating with, or by way of any other business combination, or by purchasing or exchanging any material portion of the capital stock, other equity interests or assets of, or by any other manner, any other Person;

(n) canceled, compromised, waived or released any right or claim (or series of related rights or claims) or any Funded Indebtedness (or series of related Funded Indebtedness) owed to it, in any case involving more than \$5,000;

(o) changed its accounting or Tax reporting principles, methods or policies or made or rescinded any election relating to Taxes or settled or compromised any claim relating to Taxes or, except as set forth on <u>Schedule 3.4(a)</u>, departed from GAAP;

(p) lost any distribution channel, sales location, other customer or other supplier, or source of supply of inventory, utilities or contract services or received any oral or written notice that such a loss may be pending or that there may be a material change in the relationship between such Company and any of the foregoing;

(q) made any material payments to, discounted in favor of or extended any other consideration to any supplier or customer related to the Business, other than in the Ordinary Course of Business;

(r) commenced any Proceeding against any other Person or received written notice of the commencement of any Proceeding against any Company, in each case involving more than \$5,000 or settled or compromised any such Proceeding; or

(s) claimed any Tax credit under the CARES Act or the FFCRA, funded or paid any Qualified Leave Wages, Qualified Health Plan Expenses, or any Medicare Tax on Qualified Leave Wages, from amounts allocated to or reserved for the payment of employment Taxes (including amounts already withheld) or that are set aside for deposit with the IRS, requested an "advance payment of employer credits" on IRS Form 7200, or otherwise, applied for any "Paycheck Protection Program" loan, Economic Injury Disaster Loan or any similar programs in any state, local or non-U.S. jurisdiction, or deferred any payment of Taxes otherwise due (including through any automatic extension or other grant of relief provided by an applicable Governmental Entity as a result of Section 2302 of the CARES Act, IRS Notice 2020-18, IRS Notice 2020-20 or IRS Notice 2020-23 or any similar or additional federal, state, local or foreign extension or deferral granted by an applicable Governmental Entity related to the COVID-19 pandemic);

(t) taken any other action in respect of any officer, employee or independent contractor's service, compensation or benefits, in each case, in response to the COVID-19 pandemic; or

(u) agreed to do any of the foregoing.

3.6 <u>Contracts.</u> <u>Schedule 3.6</u> sets forth a list of all of the following Contracts to which any Company is a party (collectively, the *"Material Contracts"*):

(a) each Contract in excess of \$50,000.00 between a Company and the Companies' ten (10) largest customers for the twelve (12) month period ended March 31, 2021;

(b) each Contract in excess of \$50,000.00 between a Company and the Companies' ten (10) largest suppliers for the twelve (12) month period ended March 31, 2021;

(c) each Contract under which any Company: (i) sold or purchased products or services pursuant to which the aggregate of payments due to or from any Company in the one (1) year period ending on the date of this Agreement, was equal to or exceeded \$25,000; or (ii) anticipates selling or purchasing products or services during the one year period after the date of this Agreement in which the aggregate payments due to or from any Company are expected to equal or exceed \$25,000;

(d) each Contract pursuant to which any Company is currently providing architectural services in a state with Laws requiring that licensed architects own equity in any entity that provides architectural services;

(e) each Contract for the employment of any executive or employee of any Company;

(f) each Contract under which any Company has agreed to indemnify any third Person with respect to, or to share, the Tax Liability of any third Person, other than Contracts with suppliers or customers in the Ordinary Course of Business in which no payments on account of Tax Liabilities have been made or incurred or are reasonably expected to be made or incurred;

(g) each Contract involving a commitment of any Company to make a capital expenditure or to purchase a capital asset involving more than \$25,000 individually or \$50,000 in the aggregate;

(h) each Contract that contains a covenant not to compete, most favored nation clause, or any other agreement or obligation that materially limits or will materially limit any Company from engaging in any business;

(i) each lease or similar agreement pursuant to which: (i) any Company is the lessee of, or holds or uses, any machinery, equipment, vehicle or other tangible personal property owned by any third Person for an annual rent in excess of \$5,000; (ii) any Company is the lessor of, or makes available for use by any third Person, any tangible personal property owned by it for an annual rent in excess of \$5,000; or (iii) any Company is the lessee of, or holds or uses, any real property owned by any third Person for an annual rent in excess of \$5,000; or (iii) any Company is the lessee of, or holds or uses, any real property owned by any third Person for an annual rent in excess of \$5,000; or (iii) any Company is the lessee of, or holds or uses, any real property owned by any third Person for an annual rent in excess of \$5,000;

17

(j) each licensing agreement or other Contract with respect to Intellectual Property, including any agreement with any current or former employee, consultant, or contractor regarding the appropriation or the non-disclosure of any Intellectual Property;

(k) each Contract establishing a partnership, joint venture or similar arrangement;

(l) each asset purchase agreement, stock purchase agreement, and other acquisition or divestiture agreements and similar Contracts, including any Contracts relating to the sale, lease or disposal of any material properties or assets of any Company, for consideration in excess of \$5,000;

(m) each Contract relating to Indebtedness in excess of \$5,000;

(n) each Government Contract (as defined in Section 3.23);

(o) each Contract in excess of \$1,000.00 that was not entered into in the Ordinary Course of Business;

(p) each collective bargaining agreement and other Contract to or with any labor union or other employee representative of a group of employees;

(q) each Contract providing for payments to or by any Person based on sales, purchases or profits, other than direct payments for goods;

(r) each Contract entered into other than in the Ordinary Course of Business that contains or provides for an express undertaking by any Company to be responsible for consequential, incidental or punitive damages;

(s) each written warranty, guaranty or other similar undertaking with respect to contractual performance other than in the Ordinary Course of Business;

(t) each Contract, agreement, arrangement or understanding providing for rights of indemnification or exculpation by any Company in favor of any officer, director, employee or agent or financial advisor of such Company or any other Person;

(u) each settlement, conciliation, or similar Contract entered into at any time during the five-year period immediately preceding the date of this Agreement;

(v) each Contract for which the consequences of a default or termination would have a Material Adverse Effect on the Companies, the Companies' assets or the Business (financial or otherwise);

(w) each Contract related to a transaction that would be required to be disclosed pursuant to Section 3.17; and

18

(x) all commitments and agreements to enter into any of the foregoing.

Except as set forth in <u>Schedule 3.6</u>, all Material Contracts are the legally valid and binding obligations of the applicable Companies and, to the Knowledge of the Companies, are legally valid and binding obligations of the other respective parties thereto, subject to the Equitable Exceptions, and are in full force and effect. Except as set forth in <u>Schedule 3.6</u>: (i) no Company is in material breach of, or default under, any Material Contract; (ii) to the Companies' Knowledge, no other party to any Material Contract is in breach thereof or default thereunder; and (iii) no Company has received any notice of termination with respect to any Material Contract. The Sellers have made available to the Buyer true and complete copies of the Material Contracts (including all amendments thereto). No party to any Contract shall be entitled to payments from any Company post-Closing for products or services provided to any Company pre-Closing. No party to any Material Contract has threatened to or informed the Sellers that they have repudiated any provision of any Material Contract.

3.7 <u>Litigation</u>. Except as set forth on <u>Schedule 3.7</u>, since January 1, 2016 there has not been, and currently there is no, Proceeding pending or, to the Knowledge of the Companies, threatened against any Company or any of the Companies' properties, assets or officers or directors (in their capacities as such) and, to the Knowledge of the Companies, there are no existing facts or circumstances that would reasonably be expected to result in such a Proceeding in either case. Except as set forth on <u>Schedule 3.7</u>, no Company is subject to any outstanding Order. No Company holds or possesses any commercial tort claims as of the date hereof.

3.8 <u>Taxes</u>.

(a) But for those Tax Returns, if any, currently on extension for the Tax year ending December 31, 2020, the Companies have timely filed, or has caused to be timely filed on its behalf, all Tax Returns required to be filed by any Company in accordance with all applicable Laws. To the Knowledge of the Companies, all such Tax Returns were true, correct and complete in all material respects. To the Knowledge of the Companies, all Taxes owed or required to be remitted by any Company (whether or not shown on any Tax Return) have been timely paid or remitted in full. No written claim has ever been made by any Governmental Entity in a jurisdiction where a Company does not file Tax Returns that such Company is or may be subject to taxation or required to file Tax Returns in that jurisdiction.

(b) Each Company has deducted, withheld and timely paid to the appropriate Governmental Entity all Taxes required to be deducted, withheld or paid in connection with amounts paid or owing to any employee, independent contractor, creditor, member or other third party, and each Company has complied with all applicable reporting and recordkeeping requirements, and all IRS Forms W-2 and 1099 required with respect thereto have been accurately completed and timely filed.

(c) No dispute, audit, investigation, proceeding, claim, or other action concerning any Tax Liability or Tax Return of the Companies is pending or being conducted or has been asserted in writing by a Governmental Entity. All deficiencies asserted, or assessments made, against any Company as a result of any examinations by any Governmental Entity have been fully paid. The Sellers have provided or made available to the Buyer true, correct and complete copies of all Tax Returns, examination reports, and statements of deficiencies filed, assessed against, or agreed to by any Company since December 31, 2014. No Company has waived any statute of limitations in respect of Taxes. No Company has agreed to, or is a beneficiary of, any extension of time with respect to the filing of any Tax Return.

(d) No Company is a party to or bound by any Tax sharing, Tax indemnity or Tax allocation agreement. No Company has been a member of an affiliated, combined, consolidated or unitary Tax group for Tax purposes. No Company has any Liability for the Taxes of any Person under Treasury Regulations Section 1.1502-6 (or any corresponding provision of state, local or foreign Law), as a transferee or successor, by Contract or otherwise.

(e) There are no Liens for Taxes upon any of the assets of any Company.

(f) No private letter rulings, technical advice memoranda or similar agreements or rulings have been requested, entered into or issued by any Governmental Entity with respect to such Company.

(g) No Company is, nor has it ever been, a party to, or a promoter of, a "reportable transaction" within the meaning of Section 6707A(c)(1) of the Code and Treasury Regulations Section 1.6011-4(b).

(h) No Company will be required to include any item of income in, or exclude any deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of:

(i) any change in a method of accounting, or use of an improper method of accounting, for a taxable period (or portion thereof) ending on or prior to the Closing Date;

(ii) an installment sale or open transaction occurring on or prior to the Closing Date;

(iii) a prepaid amount received, or deferred revenue accrued, on or before the Closing Date;

(iv) any closing agreement under Section 7121 of the Code, or similar provision of state, local or foreign Law;

or

(v) any election under Section 108(i) of the Code or similar provision of state, local or foreign Law.

(i) No Company has any Knowledge of any intent on the part of any Governmental Entity to assess any Company with any additional Taxes for any period for which Tax Returns have been filed.

20

(j) The Companies have established an adequate reserve on the Financial Statements for all Taxes of the Companies that have not yet become due and owing (including Taxes that are being contested in good faith). The Taxes of the Companies did not, and all Taxes of the Companies that have not yet become due and owing will not, exceed such reserves as adjusted for operations and transactions through the Closing Date in accordance with past practice and custom of the Companies in filing the Companies' Tax Returns.

(k) No Company has been a party to any transaction treated by the parties thereto as one to which Section 355 of the Code applied.

(1) Each Company has been a validly electing S corporation within the meaning of Sections 1361 and 1362 of the Code (and any corresponding provisions of applicable state or local Laws) at all times during its existence and such Company will continue to be taxed as an S corporation up to and including the Closing Date.

(m) No Company has any Liability or potential Liability for any Tax under Sections 1374, 1375 or 1363(d) of the Code.

(n) No Company has Knowledge of any basis for any Tax Authority to challenge the validity of any Company's status as an electing S corporation.

3.9 Employee Benefit Plans.

(a) <u>Schedule 3.9</u> contains a true and complete list of each bonus, deferred compensation, incentive compensation, stock purchase, stock option, profits interest, severance or termination pay, hospitalization or other medical, life or other insurance, fringe benefit, supplemental unemployment benefits, profit-sharing, 401(k) pension, or retirement plan, program, agreement or arrangement; each employment, consulting, or independent contractor contract or agreement; and each other employee benefit plan, program, agreement or arrangement, sponsored, maintained or contributed to or required to be contributed to by any Company or by any trade or business, whether or not incorporated, that together with any Company would be deemed a "single employee" within the meaning of Section 414 of the Code or Section 4001(b)(l) of ERISA (an "*ERISA Affiliate*"), for the benefit of any employee or former employee of any Company, whether formal or informal, oral or written, and whether legally binding or not, or in connection with which any Company or any ERISA Affiliate has any Liability (the "*Employee Benefit Plans*"). Schedule 3.9 identifies each of the Employee Benefit Plans that is an "employee welfare benefit plan" or "employee pension benefit plan" as such terms are defined in Sections 3(1) and 3(2) of ERISA (such plans being hereinafter referred to collectively as the "*ERISA Plans*"). No Employee Benefit Plan covers or provides benefits to independent contractors or non-employee directors of any Company or its ERISA Affiliates, or their dependents or beneficiaries.

(b) With respect to each of the Employee Benefit Plans, the Companies have made available to the Buyer true, current and complete copies of each of the following documents: (i) a copy of the plan, including all amendments thereto (or, if oral, a written summary of its material terms); (ii) if an ERISA Plan, the annual report on Form 5500 and accompanying schedules attached thereto, for the last three years; (iii) the actuarial report, if required under ERISA, with respect to each such ERISA Plan for the last three years; (iv) the most recent summary plan description for each ERISA Plan, and all summaries of material modifications thereto; (v) the most recent determination letter received from the Internal Revenue Service with respect to each ERISA Plan that is intended to be qualified under Section 401 of the Code, and any outstanding requests for determination letters; and (vi) each trust, custodial, insurance or service agreement relating to such ERISA Plan.

21

(c) To the Knowledge of the Companies, neither any Company, any ERISA Affiliate, any of the Employee Benefit Plans, any trust created thereunder or any trustee or administrator thereof has engaged in a transaction or has taken or failed to take action in connection with which any Company, any ERISA Affiliate, any of the Employee Benefit Plans, any such trust, any trustee or administrator thereof, or any party dealing with the Employee Benefit Plans or any such trust could be subject to either a civil penalty assessed pursuant to Sections 409 or 502(i) of ERISA or a tax imposed pursuant to Sections 4975, 4976, or 4980B of the Code.

(d) Each of the Employee Benefit Plans has been and is operated and administered in all material respects in accordance with its terms and applicable Law. Each Company and all of the Employee Benefit Plans are in compliance in all material respects with applicable requirements of the Code, ERISA, and any foreign laws and any Law, including, but not limited to, the requirements as to the filing of all reports and the disclosures required to be filed or furnished to Governmental Entities, participants and beneficiaries of any such Employee Benefit Plan. The Companies or one or more of the Companies' ERISA Affiliates have, and after Closing, will have, all power and authority necessary to amend or terminate each Employee Benefit Plan without incurring any penalty or liability.

(e) Each ERISA Plan intended to qualify under Section 401(a) of the Code has received a favorable determination letter from the IRS as to its qualification under the Code, no such letter has been revoked and, to the Knowledge of the Companies, nothing has occurred, whether action or failure to act, which would cause the loss of such qualification or which would result, or is likely to result, in material costs to any Company under the Internal Revenue Service's Employee Plans Compliance Resolution System.

(f) Except as set forth on <u>Schedule 3.9(f)</u>, with respect to each Employee Benefit Plan, to the Knowledge of the Companies, all required contributions or premium payments which are due for all periods ending prior to or as of the Closing have been timely made, or if not yet due have been included in Assumed Indebtedness.

(g) There are no actions, suits, claims or investigations (other than routine claims for benefits in the Ordinary Course of Business) pending, or to the Knowledge of the Companies, threatened, with respect to any Employee Benefit Plan and, to the Knowledge of the Companies, there are no facts which could give rise to any such actions, suits, claims or investigations (other than routine claims for benefits in the Ordinary Course of Business).

(h) To the Knowledge of the Companies, no Company nor any ERISA Affiliate contributes or is obligated to contribute or has ever been obligated to contribute to or has any Liability in connection with a "multiemployer plan" within the meaning of 4001(a)(3) or § 3(37) of ERISA.

(i) No Employee Benefit Plan is subject to, and neither the Companies nor any ERISA Affiliate has any Liability in connection with, Section 412 of the Code, Section 302 of ERISA or Title IV of ERISA. No Employee Benefit Plan provides employee welfare benefit that are not fully-insured by a third party that is not an ERISA Affiliate of a Company. The Companies and their ERISA Affiliates have never engaged in a transaction to evade or avoid ERISA Liability, as described in Section 4069 of ERISA.

(j) <u>Schedule 3.9(j)</u> sets forth the name of each individual who, as of or prior to the Closing, is entitled to elect or is covered under continuation coverage under COBRA in connection with any group health plan sponsored or maintained by any Company or any ERISA Affiliate. Other than such COBRA obligations, no Employee Benefit Plan provides for, and no written or oral agreements have been entered into promising or guaranteeing, the continuation of medical, dental, vision, life, or disability insurance or other plan coverage for any current or former employee or independent contractor of any Company or any ERISA Affiliate, or their dependents or beneficiaries, for any period of time beyond termination of employment or engagement.

(k) Except as set forth on <u>Schedule 3.9(k)</u>, no plan, agreement, understanding or arrangement exists that could result in the payment to any current or former employee, director or consultant of any Company of any money or other property rights or accelerate or provide any other rights or benefits to any such Person as a result of (i) the transactions contemplated herein (whether or not some other subsequent action or event would be required to cause such payment, acceleration or provision to be triggered) or (ii) the severance, termination or resignation of any such Person. To the Knowledge of the Companies, no participant under any nonqualified deferred compensation plan has been or, upon the lapse of a substantial risk of forfeiture, will be, subject to the Tax imposed by Section 409A(a)(1)(B) or 409A(b)(5)(A) of the Code.

(1) To the extent applicable, to the Knowledge of the Companies, each Company has at all times acted consistently with its material obligations under the Affordable Care Act, as amended, and the rules and regulations promulgated thereunder, including with respect to providing notice to employees, offering affordable, minimum value medical insurance coverage to substantially all of its full-time employees (within the meaning of Section 4980H of the Code) and otherwise in respect of the "Employer Shared Responsibility" provisions of Section 4980H of the Code, and neither the Companies nor any of their ERISA Affiliates, or their successor employers, is reasonably likely to have any liability from and after the Closing in connection with such Laws as they relate to the Companies and their ERISA Affiliates. Neither the Companies nor any of their ERISA Affiliates has treated any Employee Benefit Plan as having grandfathered health plan status under the Affordable Care Act.

3.10 Labor Matters.

(a) <u>Schedule 3.10</u> contains (i) a complete and accurate list of the following information for each current employee of the Companies, including each employee on leave of absence or layoff status: name, job title; the Company that employs such employee, date of hiring; date of commencement of employment; current compensation paid or payable and any change in compensation since December 31, 2020; sick and vacation leave that is accrued but unused, and if a foreign national, the applicable visa governing his/her ability to work for the applicable Company and (ii) a list of all independent contractors engaged by the Companies during the current and prior two (2) years, specifying the dates of their engagement. No Company is a party to any employee leasing, professional employer organization (PEO) or management services agreement with any Person.

(b) Each Company has complied in all respects with all Laws relating to employment practices, terms and conditions of employment, equal employment opportunity, nondiscrimination, immigration, wages, hours, benefits, collective bargaining, the payment of social security and similar Taxes and occupational safety and health. No Company is liable for the payment of any Taxes, fines, penalties, or other amounts, however designated, for failure to comply with any of the foregoing legal requirements. No Company has been the subject of any inspection or investigation relating to its compliance with or violation of Laws relating to immigration. All persons classified as non-employees and all individuals classified as exempt from overtime requirements were at all times properly classified as such.

(c) There is no collective bargaining or other labor union agreement to which any Company is a party or by which it is bound. No Company has encountered any labor union organizing activity, or had any actual or to the Knowledge of the Companies, threatened employee strikes, work stoppages, slowdowns or lockouts. There are no unfair labor practice charges, grievances or complaints pending or, to the Knowledge of the Companies threatened, by or on behalf of any employee or group of employees of any Company. There are no complaints, charges, or claims against any Company pending, or to the Knowledge of the Companies, threatened, to be brought or filed, with any Governmental Entity or arbitrator based on, arising out of, in connection with, or otherwise relating to the employment or termination of employment of any individual by any Company.

(d) No Company has effectuated: (i) any "plant closing" (as defined in the WARN Act) affecting any site of employment or one or more facilities or operating units within any site of employment of such Company; or (ii) any "mass layoff" (as defined in the WARN Act) affecting any site of employment or facility of such Company.

(e) To the Knowledge of the Companies no employee, officer or director of any Company is a party to or bound by any agreement that (i) could adversely affect the performance of his or her duties as an employee, officer or director other than for the benefit of the Companies, (ii) could adversely affect the ability of any Company to conduct the Business, (iii) restricts or limits in any way the scope or type of work in which he or she may be engaged other than for the benefit of the Companies, or (iv) requires him or her to transfer, assign or disclose information concerning his or her work to anyone other than the Companies. To the Knowledge of the Companies, no officer or employee of any Company has any plans to accept employment with any Person other than the Buyer after Closing.

3.11 Real Property and Tangible Assets.

(a) No Company owns or has ever owned any interest in any real property.

(b) <u>Schedule 3.11(b)</u> sets forth all real property currently leased by any Company (the "*Leased Real Property*"). With respect to each Leased Real Property:

(i) the applicable Company has good, valid and enforceable leasehold interests to the leasehold estate in the Leased Real Property granted to it pursuant to each pertinent lease, subject to the Equitable Exceptions;

(ii) each of the Contracts for the Leased Real Property has been duly authorized and executed by the applicable Company and is in full force and effect; and

(iii) neither the applicable Company nor, to the Knowledge of the Companies, the other party thereto, is in default under any of the leases nor, to the Knowledge of the Companies, has any event occurred which, with or without the giving of notice or the lapse of time, or both, would constitute a default by the applicable Company or the other party thereto.

(c) The Sellers have made available to the Buyer prior to the execution of this Agreement true and complete copies of all leases with respect to the Leased Real Property (including any amendments and renewal letters). The applicable Company has rights of ingress and egress with respect to the Leased Real Property sufficient to operate the Business as it is conducted on the date of this Agreement.

(d) The Companies are in possession of and have good title to, or have valid leasehold interests in or valid rights under Contract to use, all tangible personal property used in or reasonably necessary for the conduct of the Business (the "*Company Tangible Property*"). All Company Tangible Property is free and clear of all Liens, other than Liens disclosed in <u>Schedule 3.11(d)</u>, and is in good working order and condition, ordinary wear and tear excepted, suitable for its current and contemplated use.

(e) No Company maintains any inventory for sale to customers.

3.12 Intellectual Property.

(a) <u>Schedule 3.12</u> contains a complete and correct list of all Intellectual Property (excluding non-professional off-theshelf software) that is owned by each of the Companies, or used in, held for use in connection with, or necessary for the conduct of, or otherwise material to the Business (the "*Intellectual Property Assets*"). Except as set forth on <u>Schedule 3.12</u>, the Companies own or have the right to use, pursuant to a license, sublicense, agreement or permission, all of the Intellectual Property Assets, free from any Liens and free from any requirement of any past, present or future royalty payments, license fees, charges or other payments, or conditions or restrictions whatsoever. <u>Schedule 3.12</u> identifies each license, agreement or other permission, if any, which any Company has granted to any third party with respect to the Intellectual Property Assets. Each Company has taken all necessary and prudent action to maintain and protect the Intellectual Property Assets that it owns, licenses or uses. (b) To the Knowledge of the Companies, no Company has violated or infringed upon or otherwise come into conflict with any Intellectual Property of third parties, and no Company has received any notice alleging any such violation, infringement or other conflict. To the Knowledge of the Companies, no third party has infringed upon or otherwise come into conflict with any Intellectual Property of any Company.

(c) Immediately after the Closing, the Companies will own or otherwise have the right to use all the Intellectual Property Assets, free from any Liens and on the same terms and conditions as in effect prior to the Closing.

25

(d) No claim or demand of any Person has been made nor is there any proceeding that is pending, or to the Knowledge of the Companies, threatened, nor, to the Knowledge of the Companies, is there a reasonable basis therefor, that (i) challenges the rights of any Company in respect of any Intellectual Property Assets, (ii) asserts that any Company is infringing or otherwise in conflict with, or is, required to pay any royalty, license fee, charge or other amount, with regard to, any Intellectual Property, or (iii) claims that any default exists under any agreement or arrangement concerning any Intellectual Property Assets. None of the Intellectual Property Assets is subject to any outstanding Order or has been the subject of any Proceeding, whether or not resolved in favor of the Companies. To the Knowledge of the Companies there are no facts or claims which may prohibit, impair or impede the renewal of the registration of any Intellectual Property Assets.

(e) <u>Schedule 3.12(e)</u> identifies each item of Intellectual Property that any Person other than a Company owns and that any Company uses pursuant to a license, agreement or permission (a "*License*"). With respect to each item of Intellectual Property required to be identified in <u>Schedule 3.12(e)</u>: (i) to the Knowledge of the Companies, such item is not subject to any Order; (ii) to the Knowledge of the Companies, no Proceeding is pending or is threatened or anticipated that challenges the legality, validity or enforceability of such item; and (iii) no Company has granted any sublicense or similar right with respect to the License relating to such item.

3.13 <u>Environmental Matters</u>. Except as otherwise disclosed in <u>Schedule 3.13</u>:

(a) <u>Compliance with Environmental Laws and Permits</u>. The Companies possess all Environmental Permits which are necessary in order to conduct its business as it is now being conducted. A true and complete copy of each Environmental Permit issued to any Company has been made available for review by the Buyer. Each Environmental Permit issued to any Company is in full force and effect. Each Company is, and at all times has been, in material compliance with Environmental Laws and all material requirements, terms and provisions of the Environmental Permits. <u>Schedule 3.13(a)</u> lists all of the Environmental Permits issued to or held by any Company.

(b) Environmental Reports.

(i) The Companies have filed on a timely basis (and updated as required) all reports, disclosures, notifications, applications, pollution prevention, stormwater prevention or discharge prevention or response plans or other emergency or contingency plans (collectively, "*Environmental Reports*") required to be filed under Environmental Laws and the Environmental Permits. <u>Schedule</u> <u>3.13(b)</u> lists all such Environmental Reports filed by the Companies for the last three (3) calendar years. All such Environmental Reports are true, accurate and complete in all material respects.

26

(ii) The Companies have delivered to the Buyer all Environmental Reports, audits or assessments, all occupational health studies, all regulatory inspections reports and correspondence with regulatory authorities in the custody or under the control of the Companies relating to environmental matters, including, without limitation, all documents pertaining to underground and above-ground storage tanks, polychlorinated biphenyls, asbestos in buildings or products, off-site disposal of Hazardous Materials, consent orders or agreements with any Governmental Entity, and fines and penalties imposed under any Environmental Laws, in each case, that relate to property or the business of the Companies.

(c) <u>Notices</u>. Except as set out in <u>Schedule 3.13(c)</u>, no Company has received any written notice, or to Knowledge of the Companies any oral notice, that any Company or the Leased Real Property: (i) is in violation of the requirements of any Environmental Permit or Environmental Laws; (ii) is the subject of any suit, claim, proceeding, demand, order, investigation or request or demand for information arising under any Environmental Permit or Environmental Laws; (iii) has actual or potential liability under any

Environmental Law, or (iv) is the subject of any Lien obtained or asserted by any Environmental Authority as a result of any Release, use or cleanup of any Hazardous Material.

(d) <u>Environmental Conditions</u>. Except as set out in <u>Schedule 3.13(d)</u>, there are no Environmental Conditions arising out of or relating to any Company, the Business, or to the Knowledge of the Companies, the Leased Real Property.

(e) Storage, Transport or Disposal of Hazardous Materials.

(i) There is not now located on any of the Leased Real Property any areas or vessels used or intended for the treatment, storage or disposal of Hazardous Materials, including, but not limited to, drum storage areas, surface impoundments, incinerators, landfills, aboveground or underground tanks, lagoons, ponds, waste piles or deep well injection systems.

(ii) No Company has, since its inception, transported, stored, treated or disposed of any Hazardous Material at or

to any location.

3.14 <u>Insurance</u>. <u>Schedule 3.14</u> sets forth the following information with respect to each of the Companies' insurance policies: the name of the insurer, the policy number, the name of the policyholder, the period of coverage, and the amount of coverage. The Sellers have delivered to the Buyer true and complete copies of each of the Companies' insurance policies and each pending application of the Companies for any insurance policy. All premiums relating to the Companies' insurance policies have been timely paid. <u>Schedule 3.14</u> describes any self-insurance arrangements affecting the Companies. The Companies have been covered during the past ten years by insurance in scope and amount customary and reasonable for the businesses in which it has engaged during such period. The Companies are in compliance with all obligations relating to insurance created by Law or any Contract to which any Company is a party. The Companies have delivered or made available to the Buyer copies of loss runs and outstanding claims as of a recent date with respect to each of its insurance policies.

27

3.15 Compliance with Laws; Permits.

(a) Except as set forth in <u>Schedule 3.15(a)</u>, each Company has at all times since its inception been, and presently is being, operated in compliance in all material respects with all applicable Laws. Except as set forth in <u>Schedule 3.15(a)</u>, no investigation, review or Proceeding by any Governmental Entity with respect to any Company in relation to any alleged violation of Law is pending or, to the Knowledge of the Companies, threatened, nor has any Governmental Entity provided to any Company a written notice of an intent to conduct the same. No event has occurred or circumstance exists that (with or without notice or lapse of time) may constitute or result in a material violation by any Company of any Law. No Company is party to any consent or similar agreement that materially restricts the conduct of the Business or which would otherwise reasonably be expected to have a Material Adverse Effect.

(b) Each Company and each employee, shareholder and director of the Companies has obtained all Permits necessary under applicable Law to conduct the Business (the "Company Permits") and all of the Company Permits are valid and in full force and effect in all material respects. Schedule 3.15(b) contains a complete and accurate list of each Company Permit held by each Company and each employee, shareholder and director of the Companies or that otherwise relates to the Business or any asset owned or leased by any Company and states whether each such Permit is transferable. Except as set forth in Schedule 3.15(b), no defaults or violations exist or have been recorded in respect of any of the Company Permits. No proceeding is pending or, to the Knowledge of the Companies, threatened, concerning the revocation, limitation or non-renewal of any Company Permit. No Company or, to the Companies' Knowledge, any holder of a Company Permit, has received any notice or other communication from any Person regarding any actual, alleged or potential violation by any Company or any employee, shareholder and director of the Companies of any Permit or any cancellation, termination or failure to renew any Permit held by any Company or any employee, shareholder and director of the Companies. Each Company Permit listed or required to be listed on Schedule 3.15(b) is renewable according to its terms all applications required to have been filed for the renewal of the Permits have been duly filed on a timely basis with the appropriate Governmental Entities, and all other filings required to have been made with respect to such Permits have been duly made on a timely basis with the appropriate Governmental Entities and, to the Knowledge of the Companies, there is no reason why such Permit will not be renewed. The Permits listed on Schedule 3.15(b) constitute all of the Permits necessary to allow the Companies to lawfully conduct and operate its Business as currently conducted and operated and to own and use its assets as currently owned and used.

3.16 <u>Anti-Corruption and Anti-Bribery</u>. Without limiting the scope of <u>Section 3.15</u>, each Company is in compliance with, and the Business has been operated in compliance with, applicable Laws relating to political contributions, contingent fee payments, money laundering, drug trafficking, terrorist related activities or other money laundering predicate crimes, including the Currency and Foreign

Transactions Reporting Act of 1970 (otherwise known as the Bank Secrecy Act) and the USA PATRIOT Act and similar Laws and bribery and improper payments, including the U.S. Foreign Corrupt Practices Act and similar Laws. No Company, nor any of its Affiliates, nor any of their respective directors, officers, or employees, nor any other representatives or other Persons acting on their respective behalf (a) has violated or currently is violating any applicable Sanctions, (b) is a Restricted Party, or (c) has any action, complaint, petition, suit, or other proceeding, whether civil or criminal, at law or in equity, by or before any Governmental Entity pending or threatened against it with respect to Sanctions by any Sanctions Authority. None of the Companies, each of their Affiliates, or any of their respective directors, officers, employees, or distributors, other representatives or other Persons acting on their respective behalf of any Company has, directly or indirectly (x) offered, promised, given, paid, attempted to pay, or authorized the offer, promise, giving or payment of anything of value (including kickbacks, commissions, discounts, reimbursements, or other inducements), whether in cash, services, or otherwise, to a Governmental Entity or a commercial counterparty in violation of Law, or (y) made, or agreed to make, any illegal contribution, or reimbursed any illegal political gift or contribution made by any other Person, to any candidate for U.S. federal, state or local or foreign public office.

28

3.17 <u>Related Transactions</u>. Other than pursuant to employment agreements entered into in the Ordinary Course of Business (true and correct copies of which have been provided to the Buyer) or as set forth on <u>Schedule 3.17</u>, no Company is party to any direct or indirect business arrangement or relationship with any Affiliate, shareholder, director, officer or other employee of any Company (all of which business relationships and arrangements will be terminated prior to Closing). Except as set forth on <u>Schedule 3.17</u>, no Affiliate, shareholder, director, officer or other employee of any Company personally owns, directly or indirectly, any material property or right (whether tangible or intangible) that is used in the Business. Except as set forth on <u>Schedule 3.17</u> and obligations to pay compensation arising in the Ordinary Course of Business, there are no debts or other obligations owed by any Company to any Affiliate, shareholder, director, officer or other employee of any Company.

3.18 Books and Records. The minute books of each Company contain complete and accurate records of all meetings, actions without meetings and other corporate and company actions of its shareholders, and board of directors and any committees thereof. The books and records of each Company are complete and reflect all issuances, transfers, repurchases and cancellations of shares of such Company's capital stock.

3.19 <u>Bank Accounts</u>. <u>Schedule 3.19</u> sets forth the names and locations of all banks, trust companies, savings and loan associations and other financial institutions at which any Company maintains accounts of any nature, the account numbers of all such accounts and the names of all persons authorized to draw thereon or make withdrawals therefrom.

3.20 <u>Powers of Attorney: Guarantees</u>. With the exception of standard powers of attorney held by the Companies' independent certified public accounting firm with respect to dealings with Tax Authorities, no Person holds a power of attorney to act on behalf of any Company. No Company has any obligation to act under any outstanding power of attorney or any obligation or liability, either accrued, accruing or contingent, as guarantor, surety, co-signor, endorser (other than for purposes of collection in the Ordinary Course of Business), co-maker or indemnitor in respect of the obligation of any Person.

3.21 Suppliers and Customers.

(a) To the Knowledge of the Companies, no current supplier or vendor has threatened to cease or materially reduce such sales or provision of services after the date hereof. No Company has received notice that any current supplier is threatened with any bankruptcy or insolvency Proceeding. There are no unfulfilled minimum revenue commitments or unsatisfied minimum commitment periods under any Contract with any current supplier.

29

(b) To the Knowledge of the Companies, no current customer has threatened to cease or materially reduce such request for the provision of services after the date hereof. No Company has received notice that any current customer is threatened with any bankruptcy or insolvency Proceeding.

(c) Prior to or contemporaneously with the execution of this Agreement, the Sellers have delivered to the Buyer, in electronic format, complete and accurate information relating to its customers with respect to billing and/or service location, historical monthly billing averages and usage data for each such customer as of and through such date.

3.22 <u>Accounts Receivable and Payable</u>. Except as set forth in <u>Schedule 3.22</u>, since December 31, 2020, no Company has modified in any material respect any of its accounts receivable collection or accounts payable payment policies or practices and has not offered or received any discounts or credits inconsistent with the past practices of such Person. All accounts receivable as set forth on the Financial Statements or arising since the date thereof have arisen only in the Ordinary Course of Business for goods actually sold and delivered or services actually performed and represent legal, valid, binding and enforceable obligations to the Companies and, to the Knowledge of the Companies, are not and will not be subject to any contests, claims, counterclaims or setoffs other than in the Ordinary Course of Business. <u>Schedule 3.22</u> contains a list of all accounts receivable as of June 25, 2021, which list sets forth the aging of such accounts receivable. Except as set forth on <u>Schedule 3.22</u>, none of the Companies' accounts payable as of the date hereof is more than sixty (60) days past due.

3.23 <u>Government Contracts</u>. Each federal, state, municipal, or local government, quasi-government, sovereign or quasisovereign agreement, contract or subcontract (each a "*Government Contract*") to which any Company is a party is listed on <u>Schedule</u> <u>3.23</u>. No Company has (a) been suspended, debarred, or to the Companies' Knowledge, proposed for suspension and/or debarment, or voluntarily excluded from qualifying to bid and/or bidding on, or otherwise found not responsible to bid on, any Government Contract, (b) not been audited or investigated by any Governmental Entity with respect to any Government Contract entered into by a Company, except audits resulting in determinations neutral or favorable to such Company or as set forth on <u>Schedule 3.23</u> attached hereto, and (c) not had a Government Contract terminated by any Governmental Entity for default or in the past five (5) years, for convenience.

3.24 IT Systems.

(a) All IT Systems are in good working condition and are sufficient for the operation of the Companies' business as currently conducted and as proposed to be conducted. In the past three (3) years, there has been no denial-of-service or other cyber incident, including any cyberattack, or other material impairment of the IT Systems. The Companies have taken all commercially reasonable steps to safeguard the confidentiality, availability, security, and integrity of the IT Systems, including implementing and maintaining appropriate backup, disaster recovery, and software and hardware support arrangements.

30

(b) The Companies have materially complied with all applicable Laws and all internal or publicly posted policies, notices, and statements concerning the collection, use, processing, storage, transfer, and security of personal information in the conduct of the Companies' business. In the past three (3) years, the Companies have not (i) experienced any actual or alleged data breach or other security incident involving personal information in its possession or control or (ii) been subject to or received any written notice of any audit, investigation, complaint, or other Proceeding by any Governmental Entity or other Person concerning the Companies' collection, use, processing, storage, transfer, or protection of personal information or actual, alleged, or suspected violation of any applicable Law concerning privacy, data security, or data breach notification, and to the Companies' Knowledge, there are no facts or circumstances that could reasonably be expected to give rise to any such Proceeding.

3.25 <u>Service Warranties</u>. All professional services provided by the Companies have been in conformity with all applicable contractual commitments and all express and implied warranties. No Company has any Liability (and, to the Knowledge of the Companies, there is no basis for any present or future Proceeding against any Company that could give rise to any Liability) for damages in connection therewith, subject only to any reserve for warranty claims set forth on the face of the Financial Statements as adjusted for the passage of time in accordance with the past custom and practice of the Companies. No professional services provided by the Companies is subject to any guaranty, warranty or indemnity beyond the applicable standard terms and conditions of sale or lease. Attached to Schedule 3.25 are copies of the standard terms and conditions for each Company (containing applicable guaranty, warranty and indemnity provisions, if any). No Company has any Liability (and, to the Knowledge of the Companies, there is no basis for any present or future Proceeding against it that could give rise to any Liability) arising out of any injury to any individual or property as a result of the ownership, possession or use of any professional services provided by such Company.

3.26 <u>Capital Expenditures</u>. Attached as <u>Schedule 3.26</u> are (a) a list of the Companies' capital expenditures in excess of \$10,000 for the Companies' two (2) prior fiscal years and the current fiscal year through the date hereof and (b) the Companies' budget for capital expenditures for its current fiscal year and the following fiscal year. Except as set forth on <u>Schedule 3.26</u>, there are no capital expenditures that any Company currently plans to make or anticipates will need to be made during its current fiscal year or the following fiscal year in order to comply with existing Laws or to continue operating the Business following the Closing in the manner currently conducted by the Companies. No Company has foregone or otherwise materially altered any planned capital expenditure as a result of the Sellers' decision to enter into this Agreement.

3.27 FFCRA Compliance.

(a) The Companies have, since April 1, 2020, retained all information required to substantiate any Qualified Leave Wages, and any Qualified Health Plan Expenses, as required by the IRS, including by retaining copies of properly filed IRS Forms 941 and 7200 to the extent applicable.

(b) Since April 1, 2020, no Company has funded or paid any Qualified Leave Wages, Qualified Health Plan Expenses, or any Medicare Tax on Qualified Leave Wages, from amounts allocated to or reserved for the payment of employment Taxes (including amounts already withheld) or that are set aside for deposit with the IRS, in each case, whether or not shown on the Financial Statements.

(c) Since April 1, 2020, no Company has requested an "advance payment of employer credits" on IRS Form 7200 or otherwise or received a refund of Tax credits for Qualified Leave Wages or any "employee retention credit" as described in Section 2301 of the CARES Act.

Act.

(d) No Company has made a claim for Tax credits in respect of the same wages pursuant to the FFCRA and the CARES

3.28 PPP Loans. The principal amount and status of each "Paycheck Protection Program" loan obtained through the U.S. Small Business Administration (the "PPP Loan") under the CARES Act by a Company is as set forth on Schedule 3.28 (collectively, the "Company PPP Loans"). At the time of application, and at the time the Company PPP Loans were funded and, to the extent applicable, at all times prior to the forgiveness of each Company PPP Loan, the Companies satisfied all of the applicable criteria for the Company PPP Loans set forth in the Small Business Act (15 U.S.C. 636(a)) and the CARES Act (based on applicable Law, including any official public guidance of the relevant Governmental Entity on the CARES Act, existing as of the date of submission of the applicable Company's application for each of the Company PPP Loans), including, that (a) the uncertainty of current economic conditions make each of the Company PPP Loans necessary to support the ongoing operations of the applicable Companies, (b) the proceeds of the Company PPP Loans were used solely for CARES Act Permitted Purposes, (c) no Company has an application pending for a loan under subsection 7(a) of the Small Business Act or the CARES Act for the same purposes and duplicative of amounts applied for or received under the Company PPP Loans, and (d) during the period beginning on February 15, 2020 and ending on the date hereof, no Company has received amounts under subsection 7(a) of the Small Business Act for the same purpose and duplicative of amounts applied for or received under the Company PPP Loans. The application materials and supporting documentation with respect to the Company PPP Loans delivered by the Companies to the financial institutions providing the Company PPP Loans were true and correct in all material respects. The loan forgiveness application materials and supporting documentation with respect to the Company PPP Loans delivered by the Companies to the financial institutions providing the Company PPP Loans were true and correct in all material respects. As of the date of this Agreement, except as set forth on Schedule 3.28, all of the Company PPP Loans have been forgiven by the U.S. Small Business Administration.

3.29 Covid-19 Matters.

(a) Except as set forth in <u>Schedule 3.29</u>, no counterparty to any Contract to which any Company is a party has been unable to perform its obligations or defaulted under such Contract citing COVID-19 as the primary reason there-for and no Company has received any notices seeking (i) to excuse a third party's non-performance, or delay a third party's performance, under existing Contracts due to interruptions caused by COVID-19 (through invocation of force majeure or similar provisions, or other-wise) or (ii) to modify any existing contractual relationships due to COVID-19. Except as set forth in <u>Schedule 3.29</u>, no Company has issued any notices seeking to excuse such Company's non-performance, or delay such Company's performance, under existing Contracts due to interruptions caused by COVID-19 or otherwise modify any existing contractual relationships due to COVID-19.

32

(b) Other than the Company PPP Loans and those SBA loans received by 2WRCO and 2WRGA (which loans occurred prior to COVID-19), no Company has received any loans, grants, or funding directly from any government programs or any other third Person as a result of or in connection with COVID-19 or under the CARES Act or from the SBA, U.S. Department of Treasury or other Governmental Entity.

(c) No Company is collecting, or has ever collected, any health data (e.g., temperature readings, symptom information, etc.) from employees or visitors in connection with COVID-19. No Company has implemented any other measures in response to COVID-19 that involve creating, collecting, tracking, maintaining or analyzing any data relating to employees or visitors.

(d) The Companies have complied with all Governmental Orders, guidelines and directives promulgated by Governmental Entities with respect to COVID-19 in all material respects, including all Occupational Safety and Health Administration and Centers for Disease Control and Prevention guidelines and requirements.

(e) Except as set forth in <u>Schedule 3.29</u>, there have been no material changes to any Company's work force attributable to COVID-19, including (i) employee furloughs, layoffs or terminations; (ii) employee salary or wage reductions, or other changes in employee compensation, or (iii) any changes to any employee benefit that any Company makes available to its employees. No Company has amended its Employee Benefit Plans in response to COVID-19.

(f) Except as set forth in <u>Schedule 3.29</u>, no Company has received any material complaints, concerns or claims (i) from employees regarding leaves of absence, paid sick time, or similar matters related to COVID-19; (ii) about any Company's reporting, or failure to report, to employees, contractors, customers, vendors or the public, the presence of employees or contractors who have tested positive for, or exhibited symptoms of, COVID-19, or other potential means of exposure to COVID-19; or (iii) alleging any Company failed to provide a safe working environment, appropriate equipment or accommodation in relation to COVID-19. No Company is party to, or has received notice of, any pending or anticipated Proceeding (whether regarding contractual, labor and employment, benefits, or other matters) arising from COVID-19.

(g) No Company has made any claims on existing insurance policies, including business interruption insurance, as a result of COVID-19.

3.30 <u>Disclosure</u>. Neither this Agreement nor any statement contained in any certificate, Exhibit, Schedule, any Transaction Documents, or other instrument, document, agreement or writing furnished or to be furnished to, or made with, the Buyer pursuant hereto or in connection with the negotiation, execution or performance of this Agreement, contains or will contain any untrue statement of a material fact or knowingly omits or will knowingly omit to state any material fact necessary to make any statement herein or therein not misleading.

33

ARTICLE IV: REPRESENTATIONS AND WARRANTIES OF THE SELLERS

The Sellers hereby jointly and severally represent and warrant to the Buyer as of the Closing Date, as follows:

4.1 <u>Authority; Enforceability</u>.

(a) Each Seller has all requisite capacity to execute and deliver this Agreement and each of the other Transaction Documents, and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance of this Agreement and each of the other Transaction Documents, and the consummation of the transactions contemplated hereby and thereby, have been duly authorized and approved by all required action on the part of each Seller and the Seller Representative.

(b) This Agreement has been, and each of the other Transaction Documents will be at or prior to the Closing, duly and validly executed and delivered by each Seller and the Seller Representative and (assuming due authorization, execution and delivery by the Buyer (if required)) this Agreement constitutes, and each of the other Transaction Documents when so executed and delivered will constitute, legal, valid and binding obligations of each Seller, enforceable against each Seller in accordance with its terms, except for the Equitable Exceptions.

4.2 <u>Title to Shares</u>. The Sellers are the record, beneficial and lawful record owners of the Purchased Shares and the 2WRMS Shares (together, the "*Companies Shares*") and have good, valid and marketable title to the Companies Shares, free and clear of any and all Liens (excluding those restraints on the transferability of shares arising from those agreements currently existing among the Sellers, which restraints will be duly waived). There are no Contracts between any Seller and any other Person with respect to the acquisition, disposition or voting of, or any other matters pertaining to, any of the Companies Shares or any other equity interests of the Companies. Each Seller has the power, capacity and authority to sell, transfer, assign, exchange and deliver the applicable Companies Shares, as provided in this Agreement or the Share Transfer Agreement, and such delivery will convey to the Buyer good and marketable title to

such Companies Shares free and clear of any and all Liens and restrictions other than restrictions of general applicability imposed by federal or state securities Laws.

4.3 <u>Brokers</u>. No broker, finder, investment banker or other Person is entitled to any brokerage, finder's or other fee or commission from any Company or any Seller in connection with the transactions contemplated by this Agreement.

4.4 <u>Litigation</u>. There are no Proceedings pending, or to the knowledge of each Seller threatened, that are reasonably likely to prohibit or restrain the ability of any Seller to enter into this Agreement or timely to consummate the transactions contemplated hereby.

4.5 <u>Taxes</u>. Subject to those returns currently on extension, if any, each Seller has timely paid all Taxes owed by such Seller which are or were attributable to such Seller's ownership of the Companies Shares.

4.6 <u>S Shareholder Status</u>. Each Seller is (and always has been) qualified to be a shareholder of an S corporation under Section 1361 of the Code.

34

4.7 <u>Own Account</u>. Each Seller understands that the Parent Shares are "restricted securities" and upon issuance will not have been registered under the Securities Act or any applicable state securities law and such Seller is acquiring Parent Shares as principal for its own account and not with a view to or for distributing or reselling such Parent Shares or any part thereof in violation of the Securities Act or any applicable state securities law, has no present intention of distributing any of such Parent Shares in violation of the Securities Act or any applicable state securities law and has no direct or indirect arrangement or understandings with any other persons to distribute or regarding the distribution of such Parent Shares in violation of the Securities Act or any applicable state securities law. Each Seller understands that any transfer of Parent Shares will be made only in compliance with the Securities Act and applicable state securities laws.

4.8 <u>Seller Status</u>. At the time each Seller was offered the Parent Shares, such Seller was, and as of the date hereof such Seller is, and expects to be on each date on which it receives Parent Shares, an "accredited investor" as defined in Rule 501(a)(1), (a)(2), (a)(3), (a)(7) or (a)(8) under the Securities Act.

4.9 Experience of Seller. Each Seller, either alone or together with its representatives, has such knowledge, sophistication and experience in business and financial matters so as to be capable of evaluating the merits and risks of the prospective investment in the Parent Shares, and has so evaluated the merits and risks of such investment. Such Seller is able to bear the economic risk of an investment in the Parent Shares and, at the present time, is able to afford a complete loss of such investment, because, among other things, the Parent Shares have not been registered under the Securities Act, and, therefore, cannot be sold, transferred, pledged, or otherwise disposed of unless they are subsequently registered under the Securities Act and the applicable state securities laws or such Seller delivers an opinion of counsel to Parent (in reasonably acceptable form and substance to the Parent) that an exemption from such registration is available.

4.10 <u>General Solicitation</u>. No Seller is acquiring Parent Shares as a result of any advertisement, article, notice or other communication regarding the Parent Shares published in any newspaper, magazine or similar media or broadcast over television or radio or presented at any seminar or any other general solicitation or general advertisement.

4.11 <u>Access to Information</u>. In order to adequately evaluate the merits and risks of an investment in Parent, each Seller has had an opportunity to (a) ask questions and receive answers from Parent and its representatives concerning Parent and such Seller's investment therein, and (b) obtain any additional information which such Seller has requested with respect to Parent and the Seller's investment therein.

4.12 <u>No Government Approval</u>. Each Seller understands that no federal or state governmental agency has passed upon or will pass upon the Parent Shares or has made or will make any finding or determination as to the fairness of investment in the Parent Shares.

4.13 <u>Certain Transactions and Confidentiality</u>. Other than consummating the transactions contemplated hereunder, such Seller has not directly or indirectly, nor has any Person acting on behalf of or pursuant to any understanding with such Seller, executed any purchases or sales, including Short Sales, of the securities of Parent during the period commencing as of the time that such Seller first received a term sheet (written or oral) from Parent or any other Person representing Parent setting forth the material terms of the transactions contemplated hereunder and ending immediately prior to the execution hereof. Other than to other Persons party to this Agreement or to such Seller's representatives, including, without limitation, its officers, directors, partners, legal and other advisors, employees, agents and Affiliates that were required to be informed in connection with Seller's review of the Transaction Documents,

such Seller has maintained the confidentiality of all disclosures made to it in connection with this transaction (including the existence and terms of this transaction).

35

ARTICLE V: REPRESENTATIONS AND WARRANTIES OF THE BUYER AND PARENT

Except as otherwise set forth on the Parent SEC Reports, the Buyer and Parent hereby, jointly and severally, represent and warrant to the Sellers as of the Closing Date, as follows:

5.1 Existence; Good Standing; Authority; Enforceability.

(a) The Buyer is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite power and authority to own and operate its properties and to conduct its business as now conducted. Parent is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite power and authority to own and operate its properties and to conduct its business as now conducted. The Buyer and Parent are each duly licensed or qualified to do business as a foreign company and is in good standing under the laws of each other jurisdiction under which such licensing or qualification is necessary, except where the failure to be so licensed or qualified or to be in good standing would not reasonably be expected to have a material adverse effect upon on the ability of the Buyer and Parent to perform their respective obligations under this Agreement and each other Transaction Document to which it is a party.

(b) The Buyer and Parent each have the requisite corporate power and authority to execute and deliver this Agreement and the other Transaction Documents to which it is a party, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the other Transaction Documents to which it is a party, the performance by the Buyer and Parent of their respective obligations hereunder and thereunder, and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all requisite corporate action on the part of the Buyer and Parent, respectively, and no other corporate authorization or proceedings on the part of the Buyer and Parent are required therefor.

(c) This Agreement has been, and each of the other Transaction Documents will be at or prior to the Closing, duly and validly executed and delivered by the Buyer, Parent or, following the Closing, the applicable Company as a Subsidiary of the Buyer and (assuming due authorization, execution and delivery by the Sellers and the Seller Representative) this Agreement constitutes, and each of the other Transaction Documents to which the Buyer and/or Parent is a party when so executed and delivered will constitute, legal, valid and binding obligations of the Buyer or Parent, respectively, enforceable against the Buyer or Parent, respectively, in accordance with its terms, except for the Equitable Exceptions.

36

5.2 <u>No Conflict</u>. Neither the execution, delivery or performance by the Buyer or Parent of this Agreement and the other Transaction Documents to which the Buyer or Parent is a party, nor the consummation by the Buyer or Parent of the transactions contemplated by hereunder or thereunder, will violate, conflict with, breach or constitute a default under (in each case, with or without the giving of notice, the lapse of time or both), any of the provisions of: (a) any of the Organizational Documents of the Buyer or Parent; (b) any material Contract to which the Buyer or Parent is a party, or by which the Buyer, Parent or any of their respective properties is bound; or (c) any Law applicable to the Buyer, Parent or any of their respective properties.

5.3 <u>Brokers</u>. No broker, finder, investment banker or other Person is entitled to any brokerage, finder's or other fee or commission from the Buyer in connection with the transactions contemplated by this Agreement.

5.4 <u>Closing Payment Shares</u>. Prior to the Closing, Parent will have taken all necessary action to permit it to issue or otherwise deliver the Closing Payment Shares to be delivered in connection with the transactions contemplated hereby. The Closing Payment Shares, when issued and delivered as described herein, will be duly authorized, validly issued, fully paid, nonassessable and free and clear of any and all Liens, except as otherwise set forth in this Agreement, and no Person will have any preemptive right of subscription or purchase or any other right in respect thereof.

5.5 <u>SEC Filings; Financial Statements.</u>

(a) Parent has timely filed, in all material respects, all forms, reports, schedules, statements and other documents required to be filed by it during the twelve (12) months immediately preceding the date of this Agreement (collectively, as supplemented and amended since the time of filing, and including any such reports filed subsequent to the date hereof, the "*Parent SEC Reports*") with the SEC. The Parent SEC Reports (i) were prepared in all material respects in accordance with all applicable requirements of the Securities Act and the Exchange Act, as applicable, and (ii) did not, and will not (with respect to the Parent SEC Reports filed after the date hereof), contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The representation in clause (ii) of the preceding sentence does not apply to any misstatement or omission in any Parent SEC Report that was superseded by subsequent Parent SEC Reports.

(b) The audited consolidated financial statements and unaudited consolidated interim financial statements of Parent and its consolidated Subsidiaries, if any, included or incorporated by reference in the Parent SEC Reports have been prepared in accordance with GAAP consistently applied during the periods indicated (except as may otherwise be indicated in the notes) and present fairly the financial position, results of operations and cash flows of Parent and its consolidated Subsidiaries on a consolidated basis at the respective dates and for the respective periods indicated (except interim financial statements may not contain all notes and are subject to year-end adjustments).

5.6 Litigation. There are no Actions by or against Parent or the Buyer, or affecting their respective assets, pending before any Governmental Entity (or threatened in writing to be brought by or before any Governmental Entity), which, individually or in the aggregate, would impair or delay the ability of Parent or Buyer to effect the Closing and deliver to the Sellers the consideration described herein. Neither Parent nor Buyer is subject to any Governmental Order or award, injunction or decree of any arbitrator or arbitrators that, individually or in the aggregate, would impair or delay the ability of Parent or Buyer to effect the Closing and deliver to the Sellers that, individually or in the aggregate, would impair or delay the ability of Parent or Buyer to effect the Closing and deliver to the Sellers the consideration described herein.

5.7 Books and Records. Parent and the Buyer maintain complete and accurate books of accounts and records reflecting in accordance with GAAP all material items of income and expenses, all material transactions and all material assets and liabilities required to be reflected therein in accordance with applicable Law and GAAP and maintained in accordance with sound business practices and a system of internal accounting controls sufficient to provide reasonable assurances that (i) transactions are executed in accordance with management's general or specific authorization, (ii) transactions are recorded as necessary to permit preparation of the consolidated financial statements of Parent and its Subsidiaries in accordance with GAAP and to maintain asset accountability, and (iii) assets are reconciled at reasonable intervals and appropriate action is taken with respect to any material differences.

5.8 <u>Undisclosed Liabilities</u>. Parent has no material Liability of any nature, whether accrued, absolute, or contingent, other than (a) Liabilities that are properly and accurately reflected (or adequately reserved against) on their consolidated balance sheets, (b) Liabilities incurred in the Ordinary Course since December 31, 2020, or (c) Liabilities incurred in the Ordinary Course that are not required to be reflected in the balance sheets under GAAP.

5.9 <u>Absence of Changes</u>. Since December 31, 2020, there has not been any material change in the financial condition, properties, assets, liabilities, business or operations of Parent which change by itself or in conjunction with all other such changes, whether or not arising in the Ordinary Course, has had or will have a material adverse effect thereon.

5.10 <u>Availability of Funds</u>. The Buyer has available, and will have available on the Closing Date, sufficient funds to enable Buyer to consummate the transactions described herein. No action or proceeding has been commenced by or against the Buyer under any bankruptcy Law of any relevant jurisdiction for the relief of debtors or for the enforcement of the rights of creditors.

ARTICLE VI: COVENANTS

6.1 Further Action. From the date hereof and including after the Closing, the Buyer and each of the Sellers and the Companies shall use their respective good faith commercially reasonable efforts to, at such Party's own expense, take or cause to be taken all appropriate action, do or cause to be done all things necessary, proper or advisable, and execute and deliver such documents and other papers, as may be required to carry out the provisions of this Agreement and consummate and make effective the transactions contemplated by this Agreement.

38

6.2 <u>Audit</u>. Prior to Closing, the Sellers shall cause the Companies to prepare and deliver to the Buyer (a) an audited balance sheet of the Companies on a consolidated basis for the Companies' fiscal years 2019 and 2020, and the related audited statements of income, shareholders' equity and cash flow for each of the fiscal years then ended, (b) an unaudited balance sheet of the Companies on a consolidated basis for the Companies' most recently completed interim period, and the related unaudited statements of income, shareholders' equity and cash flow for such interim period and (c) any other financial statements or information that can reasonably be expected to be required by Parent to fully comply with Regulation S-X promulgated by the Securities and Exchange Commission (collectively, the "*Audited Financial Statements*"). The Audited Financial Statements shall be prepared in accordance with GAAP, consistently applied, by the Companies and the Auditor without modification of the accounting principles used in the preparation thereof throughout the periods presented (except, with respect to the unaudited interim financial statements, for (x) normal year-end adjustments which are not material, individually or in the aggregate and (y) the absence of normal disclosures made in footnotes) and shall fairly present, in all material respects, the consolidated financial position, results of operations and cash flows of the Companies as at the dates and for the periods indicated therein.

6.3 <u>Conduct of Business Prior to the Closing</u>. From the date hereof until the Closing, except as otherwise provided in this Agreement or consented to in writing by the Buyer (which consent shall not be unreasonably withheld or delayed), the Sellers shall, and shall cause the Companies to, (i) conduct the business of the Companies in the Ordinary Course of Business consistent with past practice; (ii) use reasonable best efforts to maintain and preserve intact the current organization, business and franchise of the Companies and to preserve the rights, franchises, goodwill and relationships of each Company's employees, customers, lenders, suppliers, regulators and others having business relationships with any of the Companies Company; (iii) retain key management members in the same capacity as currently employed; and (iv) keep available the services of the present officers, employees, agents and other personnel of the Companies. Without limiting the foregoing, from the date hereof until the Closing Date, the Sellers shall:

(a) cause each of the Companies, the Sellers and any employees of the Companies to preserve and maintain all of their respective Permits;

(b) cause the Companies not to enter into, modify or terminate (i) any Material Contract, or (ii) any employment agreement (other than to remove any change of control or severance provisions) without the prior written consent of the Buyer, which consent shall not be unreasonably withheld;

(c) not allow any Company to incur or guarantee any Indebtedness involving or related to any of the Companies;

(d) cause each of the Companies not to sell, assign, license, lease, transfer, convey or pledge any assets of such Company or commit itself to sell, assign, license, lease, transfer, convey or pledge any assets of such Company or subject any of such assets to a security interest, in all cases other than entering into licenses and sales of such Company's products and services in the Ordinary Course of Business;

39

(e) not allow any of the Companies to issue any equity interests or equity securities;

(f) cause each Company to pay its debts, Taxes and other obligations when due;

(g) cause each Company to maintain the properties and assets owned, operated or used by such Company in the same condition as they were on the date of this Agreement, subject to reasonable wear and tear and in the event of any damage or destruction of such properties or assets prior to the Closing Date, promptly replace, repair or restore such properties or assets;

(h) cause each Company to continue in full force and effect without modification all of such Company's insurance policies, except as required by applicable Law;

(i) cause each Company to defend and protect its properties and assets from infringement or usurpation;

(j) cause each Company to perform all of its obligations under all Contracts relating to or affecting its properties, assets

or business;

(k) cause each Company to maintain its books and records in accordance with past practice;

(l) comply and cause each Company to comply in all material respects with all applicable Laws;

(m) execute, or allow any Company or any Affiliate of any Seller or Company to execute any transaction described in <u>Section 4.13</u> hereof; and

(n) not to take or permit any action that would cause any of the changes, events or conditions described in Section 3.5

to occur.

6.4 No Solicitation of Other Bids.

(a) The Sellers shall not, and shall not authorize or permit any of their respective Affiliates (including the Companies) or any of their (or their Affiliates') Representatives to, directly or indirectly, (i) encourage, solicit, initiate, facilitate or continue inquiries regarding an Acquisition Proposal; (ii) enter into discussions or negotiations with, or provide any information to, any Person concerning a possible Acquisition Proposal; or (iii) enter into any agreements or other instruments (whether or not binding) regarding an Acquisition Proposal. The Sellers shall immediately cease and cause to be terminated, and shall cause its Affiliates (including the Companies) and all of its and their Representatives to immediately cease and cause to be terminated, all existing discussions or negotiations with any Persons conducted heretofore with respect to, or that could lead to, an Acquisition Proposal. For purposes hereof, "*Acquisition Proposal*" shall mean any inquiry, proposal or offer from any Person (other than the Buyer or any of its Affiliates) concerning (x) a merger, consolidation, liquidation, recapitalization, share exchange or other business combination transaction involving any of the Companies; (y) the issuance or acquisition of shares of capital stock or other equity securities of any of the Companies; or (z) the sale, lease, exchange or other disposition of any significant portion of any of the Companies' properties or assets.

40

(b) In addition to the other obligations under this <u>Section 6.4</u>, the Sellers shall promptly (and in any event within three (3) Business Days after receipt thereof by any of the Companies, the Sellers or their Representatives) advise the Buyer orally and in writing of any Acquisition Proposal, any request for information with respect to any Acquisition Proposal, or any inquiry with respect to, or which could reasonably be expected to result in, an Acquisition Proposal, the material terms and conditions of such request, Acquisition Proposal or inquiry, and the identity of the Person making the same.

(c) The Sellers agree that the rights and remedies for noncompliance with this <u>Section 6.4</u> shall include having such provision specifically enforced by any court having equity jurisdiction, it being acknowledged and agreed that any such breach or threatened breach shall cause irreparable injury to the Buyer and that money damages would not provide an adequate remedy to the Buyer.

6.5 Notice of Certain Events.

(a) From the date hereof until the Closing, the Sellers shall promptly notify the Buyer in writing of:

(i) any fact, circumstance, event or action the existence, occurrence or taking of which (A) has had, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, (B) has resulted in, or could reasonably be expected to result in, any representation or warranty made by the Sellers hereunder not being true and correct or (C) has resulted in, or could reasonably be expected to result in, the failure of any of the conditions set forth in <u>Section 10.1</u> or <u>Section 10.2</u> to be satisfied;

(ii) any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the transactions contemplated by this Agreement;

(iii) any notice or other communication from any Governmental Entity in connection with the transactions contemplated by this Agreement; and

(iv) any Proceedings commenced or, to the Companies' Knowledge, threatened against, relating to or involving or otherwise affecting the Sellers or the Companies that, if pending on the date of this Agreement, would have been required to have been disclosed pursuant to <u>Section 3.7</u> or that relates to the consummation of the transactions contemplated by this Agreement.

(b) Except pursuant to Section 6.5(c), the Buyer's receipt of information pursuant to this Section 6.5 shall not operate as a waiver or otherwise affect any representation, warranty or agreement given or made by the Sellers in this Agreement (including Section 8.1(b) and Section 10.2) and shall not be deemed to amend or supplement the Companies Disclosure Schedules.

(c) From time to time prior to the Closing, the Sellers shall have the right (but not the obligation, except as required herein) to supplement or amend the Schedules hereto with respect to any matter hereafter arising or of which it becomes aware after the date hereof (each a "*Schedule Supplement*"). Any disclosure in any such Schedule Supplement shall not be deemed to have cured any inaccuracy in or breach of any representation or warranty contained in this Agreement, including for purposes of the indemnification or termination rights contained in this Agreement or of determining whether or not any condition set forth in <u>Article X</u> has been satisfied; <u>provided, however</u>, that if the Buyer has the right to terminate this Agreement under <u>Section 11.1(b)</u> as a result of additional disclosures in a Schedule Supplement solely related to an event, act or occurrence that arose after the date this Agreement was executed but prior to Closing and nonetheless proceeds to close, then the Buyer shall be deemed to have irrevocably waived its right to indemnification under <u>Section 8.1(b)(i)</u> with respect to such matter.

6.6 <u>Consents and Approvals</u>. Each of the Buyer, the Sellers and the Companies shall use its commercially reasonable efforts to: (a) take all actions necessary or appropriate to consummate the transactions contemplated by this Agreement and the Transaction Documents; and (b) cause the fulfillment at the earliest practicable date of all of the conditions to their respective obligations to consummate the transaction Documents.

6.7 <u>Tax Distributions</u>. Nothing in this Agreement shall prohibit the Companies from making distributions, prior to Closing, without penalty and consistent with past practices of the Companies, out of the Companies' cash for the purposes of allowing the Sellers to pay those federal, state and local tax liabilities arising from the 2020 fiscal year and the estimated federal, state and local tax liabilities for that certain stub period beginning as of January 1, 2021 and ending as of the date of Closing.

6.8 <u>Press Release/Public Announcement</u>. No party to this Agreement shall issue make, or cause to be issued or made, a press release or other public announcement respecting this Agreement or the transactions contemplated herein. Following Closing, if the Buyer desires to issue a press release or other public announcement relating to the acquisition of the Companies, the Buyer will provide the Seller Representative with a draft of the press release or other public announcement for review at least three (3) Business Days prior to the time that such press release or other public announcement is to be made. Such press release or other public announcement shall not be issued without the Seller Representative's consent, which consent shall not be unreasonably withheld; provided, however, that the Buyer may, without the prior consent of the Seller Representative (but after such consultation, to the extent practicable in the circumstances), issue such press release or make such public statements as may upon the advice of outside counsel be required by law or the rules or regulations of the Exchange.

6.9 Life Insurance Policies. The Buyer has advised the Companies that they will not be keeping the key man insurance on the lives of the Key Employees currently owned by the Companies in place. Notwithstanding any other provision set forth in this Agreement, a Key Employee may, no later than ten (10) Business Days after the date of Closing, in such Key Employee's sole and absolute discretion, purchase from the applicable Companies the life insurance policy(ies) on his life. The purchase price for such policy(ies) shall be their cash surrender value, increased by the amount of any unearned premiums and reduced by the amount of any loans secured by such policies.

42

6.10 <u>Company Vehicles</u>. The Parties acknowledge that the Companies provide the vehicles set forth on <u>Schedule 6.10</u> to the Key Employees (the "*Company-Owned Vehicles*"). Each Key Employee shall advise the Buyer, no later than three (3) days prior to the Closing Date, whether or not such Key Employee desires to purchase the Company-Owned Vehicle assigned to such Key Employee from the applicable Company. If a Key Employee desires to acquire the Company-Owned Vehicle assigned to such Key Employee from the Applicable Company, such Key Employee shall purchase the Company-Owned Vehicle assigned to the Key Employee from the applicable Company on or prior to the date of the Closing for an amount equal to the value set forth in the Kelley Blue Book online (www.kbb.com), taking into consideration all applicable characteristics of such Company-Owned Vehicle, including, without limitation: year, make, model, mileage, condition, style, package and equipment. The Parties agree that the "Private Party" value shall be controlling. In the event a price range is provided, the purchase price for such Company-Owned Vehicle shall be the midpoint of such price range.

6.11 <u>Memberships</u>. Following Closing, neither the Buyer nor the Companies shall continue to pay for any club memberships benefiting the Key Employees, including Green Island Country Club and the Columbus Country Club.

ARTICLE VII: TAX MATTERS

7.1 Tax Returns and Related Matters.

(a) The Sellers shall prepare or cause to be prepared at their own expense and the Companies shall file or cause to be filed all Tax Returns of the Companies for all taxable periods ending on or prior to the Closing Date ("*Pre-Closing Periods*") that are due and filed after the Closing Date, including the Tax Return due for the short Tax period ending on the Closing Date (the "*Final Pre-Closing Period*"). Such Tax Returns shall be prepared in a manner consistent with past practice; <u>provided</u>, <u>however</u>, that in any event such Tax Returns shall be prepared in a manner that is consistent with applicable Law. The Sellers shall provide the Buyer with a copy of the Tax Returns that it prepares under this <u>Section 7.1(a)</u> at least twenty (20) days prior to the filing of such Tax Returns, and the Sellers will incorporate any reasonable comments made by Buyer within such 20-day period; <u>provided</u>, that the Sellers shall not take any position or apply any methodology in preparing any such Tax Return that is not consistent with the Tax Returns relating to prior taxable periods (such as, for example and without limitation, practices with respect to the calculation of depreciation expense deductions), <u>provided</u>, that such practices and methodologies comply with applicable Law. The Buyer shall have the right to handle, defend, conduct and control any claim for any Tax refund due any Company with respect to the Tax Returns described in this <u>Section 7.1(a)</u>.

43

(b) The Buyer shall prepare (or cause to be prepared) and file (or cause to be filed) when due (taking into account all extensions properly obtained) all Tax Returns required to be filed by or with respect to any Company relating to taxable periods that begin before and end after the Closing Date ("*Straddle Periods*"). The Buyer shall deliver to the Seller Representative copies of each such Tax Return relating to the Straddle Periods, along with a statement (a "*Tax Statement*") showing the pre-Closing portion of any Tax Liability required to be paid with each such Tax Return (computed in accordance with <u>Section 7.1(d</u>)), at least twenty (20) days prior to the due date for filing such Tax Return, and shall permit the Seller Representative to review and comment on such Tax Return and Tax Statement prior to filing; the Buyer will incorporate any reasonable comments made in good faith by the Seller Representative within such 20-day period. For the avoidance of doubt, the Sellers and the Buyer acknowledge and agree that there will be no Straddle Periods with respect to Income Taxes.

(c) No later than the later of (i) five (5) days prior to the due date for the payment of Taxes shown on any Tax Returns due after the Closing Date relating to Pre-Closing Periods or Straddle Periods, or (ii) five (5) days after the Sellers' receipt of written notice setting forth the Sellers' share of the applicable liability described in this subsection, the Sellers shall pay to the Buyer in immediately available funds the amount of cash Taxes payable as shown on the Tax Return (in the case of a Pre-Closing Period) or the Tax Statement (in the case of a Straddle Period).

(d) Each Company will, unless prohibited by applicable Law, close the taxable year of such Company as of the close of business on the Closing Date. If applicable Law does not permit such Company to close its taxable year on the Closing Date, the Taxes, if any, attributable to such Straddle Period shall be allocated: (i) to the Sellers for the period up to and including the close of business on the Closing Date; and (ii) to the Buyer for the period subsequent to the Closing Date, pursuant to the following methodology: (x) Taxes, other than those referred to in clause (y) below, shall be allocated by means of a closing of the books and records of such Company as of the close of the Closing Date; provided that exemptions, allowances or deductions that are calculated on an annual basis (including, but not limited to, depreciation and amortization deductions) shall be allocated between the period ending on the Closing Date and the period after the Closing Date in proportion to the number of days in each such period; and (y) property Taxes and ad valorem Taxes shall be allocated between the period ending on the Closing Date and the period after the Closing Date in proportion to the number of days in each such period; and (y) property Taxes and ad valorem Taxes shall be allocated between the period ending on the Closing Date and the period after the Closing Date in proportion to the number of days in each such period.

(e) Notwithstanding any other provisions hereof, if an audit or other proceeding is commenced, and an adjustment is proposed or any other claim is made by any Governmental Entity with respect to a Tax Liability of any Company solely relating to a Pre-Closing Period, including, but not limited to, the Final Pre-Closing Period or the pre-Closing portion of any Straddle Period (a "*Tax Claim*"), the Buyer shall promptly notify the Seller Representative of such Tax Claim. The Sellers shall have the right to handle, defend, conduct and control any such Tax Claim; <u>provided</u>, that the Sellers shall consult periodically with the Buyer as to strategic and tactical issues for pursuing or defending any Tax Claim. Neither the Seller Representative nor any Seller shall have the right to compromise or

settle any such Tax Claim which would result in any increased Tax Liability for any Company or Buyer in any post-Closing Tax period without the prior written consent of the Buyer.

(f) Any refunds (and any interest received thereon) of any Tax imposed on any Company for any Pre-Closing Period, the Final Pre-Closing Period or the pre-Closing portion of any Straddle Period (determined in accordance with <u>Section 7.1(d)</u>) shall be payable to the Sellers. Any other refunds of any Tax imposed on any Company (and any interest received thereon) shall be the property of such Company and shall inure to the benefit of Buyer.

(g) Following the Closing, the Buyer and the Sellers shall provide each other with such assistance as may reasonably be requested by any of them in connection with the preparation of any Tax Return, any audit or other examination by any Tax Authority, or any judicial or administrative proceedings relating to Liability for Taxes of any Company. The party requesting assistance hereunder shall reimburse the other for reasonable out-of-pocket expenses incurred in providing such assistance. The Buyer and the Sellers shall preserve and cause to be preserved all information, Tax Returns, books, records and documents relating to any Liabilities for Taxes of any Company with respect to a taxable period until the later of sixty (60) days after the expiration of all applicable statutes of limitations and extensions thereof; or the conclusion of all litigation with respect to Taxes for such period.

7.2 <u>Certain Transfer Taxes</u>. All transfer, recording, documentary, sales, use, stamp, registration and other such Taxes and fees (including any related penalties and interest) incurred in connection with this Agreement (collectively, "*Transfer Taxes*") shall be timely paid by the Sellers when due, and the Sellers shall, at their sole expense, timely file all necessary Tax Returns and other documentation with respect to all such Transfer Taxes and fees, and, if required by applicable Law, the Buyer shall, and shall cause its Affiliates to, join in the execution of any such Tax Returns and other documentation.

ARTICLE VIII: INDEMNIFICATION

8.1 Survival of Representations and Warranties; Indemnification Obligations.

(a) Subject to the limitations and other provisions of this Agreement, the representations and warranties of the Parties contained herein, as the case may be, shall survive the Closing and shall remain in full force and effect until the date that is eighteen (18) months following the Closing Date, except that: the representations and warranties (collectively, the "*Fundamental Representations*") set forth in <u>Section 3.1</u> (Existence; Good Standing), <u>Section 3.2</u> (Capitalization; Subsidiaries), <u>Section 3.3</u> (No Conflict), <u>Section 3.8</u> (Taxes), <u>Section 3.9</u> (Employee Benefit Plans), <u>Section 3.12</u> (Intellectual Property) <u>Section 3.13</u> (Environmental Matters), <u>Section 3.16</u> (Anti-Corruption and Anti-Bribery), <u>Section 3.27</u> (FFCRA Compliance), <u>Section 3.28</u> (PPP Loans), <u>Section 4.1</u> (Authority; Enforceability), <u>Section 4.2</u> (Title to Shares) and <u>Section 4.3</u> (Brokers), <u>Section 4.5</u> (Taxes) and <u>Section 4.6</u> (S Shareholder Status) shall survive the Closing until sixty (60) days after the expiration of the applicable statute of limitations for the matters described in such representations and warranties plus any extensions or waivers thereof. All covenants and agreements of the parties contained herein shall survive the Closing indefinitely or for the period explicitly specified therein. Notwithstanding the foregoing, any claims asserted in writing by notice from the Indemnified Party seeking indemnification under this Agreement to an Indemnifying Party prior to the expiration date of the applicable survival period shall not thereafter be barred by the expiration of the relevant representation or warranty and such claims and any other claim(s) reasonably related thereto shall survive until finally resolved.

45

(b) Subject to the other provisions of this <u>Article VIII</u>, the Sellers shall jointly and severally indemnify and hold harmless the Buyer and each of its Affiliates (which, following the Closing, shall include each of the Companies) and its and their respective directors, officers, employees, stockholders, partners, members, managers, agents and representatives (each, a "*Buyer Indemnified Party*") from and against, and pay to the Buyer Indemnified Parties, any and all Losses, whether or not involving a Third Party Claim, incurred or sustained, directly or indirectly, based upon, relating to, arising out of, with respect to or by reason of:

(i) any breach of, or inaccuracy in, or any allegation of any third party that, if true, would be a breach or inaccuracy of, any representation or warranty made by the Sellers in this Agreement;

(ii) any breach of, or failure of any Seller to perform, any covenant agreement or obligation of any Seller in

this Agreement;

(iii) any Transfer Taxes;

(iv) any Taxes of any Company or any Seller with respect to any Pre-Closing Period, the Final Pre-Closing Period or the pre-Closing portion of any Straddle Period. For the avoidance of doubt, the indemnification rights under this <u>Section 8.1(b)</u> shall be in addition to the indemnification rights relating to the Sellers' representations and warranties set forth in <u>Section 3.8</u> (Taxes);

(v) any amounts owed to or recoverable by the Buyer under Section 2.2(d).

(vi) any Transaction Fees to the extent not reflected in the Flow of Funds Memorandum;

(vii) any Funded Indebtedness (other than Assumed Indebtedness) to the extent not reflected in the Flow of

Funds Memorandum;

(viii) COVID-19 Liabilities;

(ix) the Company PPP Loans, including, without limitation, any non-compliance with any Company PPP Loan or with the CARES Act, any litigation arising in respect thereof or in relation thereto, any audit, review or investigation pertaining to any Company PPP Loan, and any retroactive determination of ineligibility with respect to any Company PPP Loan or reversal of forgiveness granted in respect thereof;

(x) any Shareholder Claim;

(xi) the Wenzel Phantom Share Agreement;

(xii) any Environmental and Occupational Safety and Health Liability; and

(xiii) the matters set forth on Schedule 8.1(b)(xiii).

46

(c) In the absence of fraud or intentional misrepresentation, the Sellers shall not be required to indemnify any Buyer Indemnified Party with respect to a claim for indemnification pursuant to <u>Section 8.1(b)(i)</u> unless and until the cumulative aggregate amount of all indemnifiable Losses which may be recovered by the Buyer Indemnified Parties pursuant to <u>Section 8.1(b)(i)</u> exceeds \$50,000, in which event the Sellers will be liable for all such indemnifiable Losses from the first dollar; <u>provided</u>, <u>however</u>, that the limitations set forth in this <u>Section 8.1(c)</u> shall not apply to indemnification obligations with respect to any Fundamental Representations.

(d) In the absence of fraud or intentional misrepresentation, no Seller shall be required to indemnify any Buyer Indemnified Party with respect to a claim for indemnification pursuant to Section 8.1(b)(i), for any amount of indemnifiable Losses in excess of \$2,500,000 (the "*Cap*"); provided, however, that the Cap shall not apply to Losses related to Fundamental Representations. In the absence of fraud or intentional misrepresentation, the Sellers shall not be required to indemnify any Buyer Indemnified Party with respect to a claim for indemnification pursuant to Section 8.1(b)(i) for any amount of indemnified Party with respect to a claim for indemnification pursuant to Section 8.1(b)(i) for any amount of indemnifiable Losses in excess of the Purchase Price.

(e) Subject to the other provisions of this <u>Article VIII</u>, the Buyer shall indemnify and hold harmless each Seller (the "*Seller Indemnified Party*") from and against, and pay to the Seller Indemnified Parties, any and all Losses, whether or not involving a Third Party Claim, actually incurred (and as incurred) by such Seller Indemnified Party, incurred or sustained, directly or indirectly, based upon, relating to, arising out of, with respect to or by reason of

<u>V;</u> or

(i) any breach of, or inaccuracy in, any representation or warranty concerning the Buyer contained in Article

(ii) any breach of, or failure of the Buyer to perform, any covenant agreement or obligation of the Buyer in

this Agreement.

(f) Notwithstanding anything to the contrary contained in this Agreement, for purposes of determining the amount of any Losses that are the subject matter of a claim for indemnification hereunder, each representation and warranty in this Agreement and

each certificate or document delivered pursuant hereto shall be read without regard and without giving effect to the term(s) "material" or "material adverse effect" or similar qualifiers as if such words and surrounding related words (*e.g.*, "reasonably be expected to," "could have" and similar restrictions and qualifiers) were deleted from such representation and warranty.

8.2 Indemnification Claim Procedures.

(a) Promptly after obtaining actual knowledge of any matter that a Buyer Indemnified Party or a Seller Indemnified Party (the "*Indemnified Party*"), acting in good faith, reasonably believes will entitle the Indemnified Party to indemnification from the other party (the "*Indemnifying Party*") under <u>Section 8.1(b)</u> or <u>Section 8.1(e)</u>, as applicable, the Indemnified Party shall promptly provide to Indemnifying Party notice describing the matter in reasonable detail, including the nature of the Claim, the basis for the indemnification obligation and the Losses resulting therefrom (a "*Notice of Claim*"); <u>provided</u> that the failure to so notify the Indemnifying Party promptly shall not relieve the Indemnifying Party of such Indemnifying Party's Liabilities hereunder except to the extent such failure shall have actually and materially prejudiced the Indemnifying Party.

(b) For Claims for indemnification under this <u>Article VIII</u> other than those relating to Third Party Claims (as defined below), the Indemnifying Party shall have fifteen (15) days after its receipt of the Notice of Claim to respond to the claim(s) described therein. Such response shall set forth, in reasonable detail, the Indemnifying Party's objection(s) to the Claim(s) and its bases for such objection(s). If the Indemnifying Party fails to provide such a response with such time period, the Indemnifying Party will be deemed to have conceded the claim(s) set forth in the Notice of Claim. If the Indemnifying Party provides its response within such time period, the Indemnifying Party and the Indemnifying Party shall negotiate the resolution of the Claim(s) for a period of not less than twenty (20) Business Days after such response is provided. If the Indemnifying Party and the Indemnified Party are unable to resolve any such claim(s) within such time period, the Indemnified Party shall be entitled to pursue any legal remedies available to the Indemnified Party against the Indemnifying Party with respect solely to the unresolved Claim(s), subject to the provisions of this <u>Article VIII</u>. The Indemnifying Party shall pay any amounts determined to be owed to the Indemnified Party in accordance with this <u>Article VIII</u> in cash as soon as reasonably practicable after any such determination.

8.3 Third Party Claims.

(a) If a third party commences a lawsuit or arbitration against an Indemnified Party with respect to any matter that the Indemnified Party might make a claim for indemnification against any Indemnifying Party under this <u>Article VIII</u> (a "*Third-Party Claim*"), then the Indemnified Party must notify the Indemnifying Party thereof in writing of the existence of such Third-Party Claim and must deliver copies of any documents served on the Indemnified Party with respect to the Third-Party Claim; <u>provided</u>, <u>however</u>, that any failure to notify the Indemnifying Party or deliver copies will not relieve the Indemnifying Party from any obligation hereunder unless (and then solely to the extent) the Indemnifying Party is materially prejudiced by such failure and such failure continues for more than thirty (30) days after the Indemnified Party's receipt of such Third-Party Claim.

(b) Upon receipt of the notice described in Section 8.3(a), the Indemnifying Party will have the right to defend the Indemnified Party against the Third-Party Claim with counsel reasonably satisfactory to the Indemnified Party so long as (i) within ten days after receipt of such notice, the Indemnifying Party notifies the Indemnified Party in writing that the Indemnifying Party will indemnify the Indemnified Party from and against any Losses the Indemnified Party may incur relating to or arising out of the Third-Party Claim, (ii) the Indemnifying Party provides the Indemnified Party with evidence reasonably acceptable to the Indemnified Party that the Indemnifying Party will have the financial resources to defend against the Third-Party Claim and fulfill its indemnification obligations hereunder, (iii) the Indemnifying Party is not a party to the Proceeding or the Indemnified Party has determined in good faith that there would be no conflict of interest or other inappropriate matter associated with joint representation, (iv) the Third-Party Claim does not involve, and is not likely to involve, a claim by any Governmental Entity, (v) the Third-Party Claim involves only money damages and does not seek an injunction or other equitable relief, (vi) settlement of, or an adverse judgment with respect to, the Third-Party Claim is not, in the good faith judgment of the Indemnified Party, likely to establish a precedential custom or practice adverse to the continuing business interests of the Indemnified Party, (vii) the Indemnified Party does not have additional defenses to the Third-Party Claim not available to the Indemnifying Party, (viii) the Indemnified Party has determined that an adverse determination with respect to the Third-Party Claim would not be materially detrimental to or materially injure the Indemnified Party's reputation or business, (ix) the Indemnifying Party conducts the defense of the Third-Party Claim actively and diligently, and (x) the Indemnifying Party keeps the Indemnified Party apprised of all developments, including settlement offers, with respect to the Third-Party Claim and permits the Indemnified Party to participate in the defense of the Third-Party Claim.

(c) So long as the Indemnifying Party is conducting the defense of the Third-Party Claim in accordance with <u>Section</u> <u>8.3(b)</u>, the Indemnifying Party will not consent to the entry of any judgment or enter into any settlement with respect to the Third-Party Claim without the prior written consent of the Indemnified Party, which consent will not be withheld unreasonably.

(d) If any condition in <u>Section 8.3(b)</u> is or becomes unsatisfied, (i) the Indemnified Party may defend against, and consent to the entry of any judgment or enter into any settlement with respect to, the Third-Party Claim in any manner it may deem appropriate (and the Indemnified Party need not consult with, or obtain any consent from, the Indemnifying Party in connection therewith); (ii) the Indemnifying Party will reimburse the Indemnified Party promptly and periodically (but no less often than monthly) for the costs of defending against the Third-Party Claim, including reasonable attorneys' fees and expenses (provided, however, that in the event there are more claims between the Parties than just the Third-Party Claim, the Indemnified Party shall separately track and account for those costs, including any attorneys' fees and expenses, directly related to the Third-Party Claim); and (iii) the Indemnifying Party will remain responsible for any Losses the Indemnified Party may incur relating to or arising out of the Third-Party Claim to the fullest extent provided in this <u>Article VIII</u>.

8.4 <u>Rights Not Affected by Investigation</u>. The right to indemnification or any other remedy based on representations, warranties, covenants or agreements in this Agreement or any agreement, document or certificate delivered hereunder shall not be affected by any investigation conducted at any time, or any knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing Date, with respect to the accuracy or inaccuracy of, or compliance with, any such representation, warranty, covenant or agreement. The waiver of any condition based on the accuracy of any such representation of warranty, or on the performance of or compliance with any such covenant or agreements, will not affect the right to indemnification or any other remedy based on such representations, warranties, covenants or agreements.

8.5 No Right of Contribution. No Seller shall have a right of contribution or other recourse against any Company or any of its directors, officers, employees, Affiliates, agents, representatives, assigns or successors for any Third Party Claims asserted by the Buyer Indemnified Parties, it being acknowledged and agreed that the representations, warranties, covenants and agreements of each Company are solely for the benefit of the Buyer Indemnified Parties.

8.6 <u>Right to Set-Off</u>. The Buyer shall have the right to set-off any amount that is determined pursuant to this <u>Article VIII</u> to be payable by the Sellers to any Buyer Indemnified Party against (i) the Escrowed Shares, valued for the purpose of set-off as set forth in <u>Section 2.6(a)(iii)</u>, provided that the amount of such set-off(s) has been determined pursuant to a final judgment or mutual agreement of the Parties, until the Escrowed Shares are exhausted or are otherwise disbursed to the Sellers under the terms of this Agreement and (ii) any sums otherwise payable to the Sellers.

8.7 <u>Exclusive Remedy</u>. This <u>Article VIII</u> will provide the exclusive legal remedy for the matters covered by this <u>Article VIII</u>, except for claims based upon fraud or intentional misrepresentation. This <u>Article VIII</u> will not affect any equitable remedy available to any Party.

ARTICLE IX: RESTRICTIVE COVENANTS AND RELEASE

9.1 <u>Restrictive Covenants</u>. Recognizing the Buyer's need to protect the goodwill of the Business being purchased and to induce the Buyer to purchase the Business and the Purchased Shares, each of the Majority Shareholders covenants and agrees with the Buyer that, if the transactions contemplated hereby are closed, no Majority Shareholder will directly or indirectly (including through an Affiliate of any Majority Shareholder) and shall cause its Affiliates to not, for the Restraint Period (as defined below):

(a) within the Restraint Area (as defined below), compete with any Company or the Buyer with respect to the Business as presently or previously conducted by any Company; <u>provided</u> that such Majority Shareholder shall not be deemed in breach of this <u>Section 9.1(a)</u> for its passive ownership of up to one percent (1.0%) of the equity securities of a publicly-traded company;

(b) solicit, divert, or take away or attempt to solicit, divert or take away from the Buyer or its Affiliates, for the benefit of any other Person, any customer of the Buyer or its Affiliates that was a customer of any Company as of the Closing Date or at any time within two (2) years prior to the Closing Date;

(c) solicit for hire (whether as an employee or independent contractor) any employee or full-time independent contractor of the Business, or encourage any such employee or full-time independent contractor to terminate his or her relationship with the Business;

(d) induce or attempt to induce any customer, supplier, licensee, licensor, franchisee or other business relation of the Business to cease doing business with the Business, or in any way interfere with the relationship between any such customer, supplier, licensee or business relation and the Business (including making any negative statements or communications about the Buyer or any of its Affiliates including the Companies); or

(e) divert or attempt to divert any or all of the Buyer's or its Affiliates' customers' or suppliers' business with such Person from such Person in violation of this Agreement or applicable Law (including the violation of any trade secrets law).

(f) As used herein the term "compete" shall include engaging in competition, whether by the sale of products or services, directly or indirectly, as a proprietor, principal, agent, representative, employee, consultant, partner, or equityholder.

50

(g) As used herein, the term "*Restraint Area*" means the United States, its territories and any other territory or country where the Company conducts or has conducted business in the two (2) years preceding the Closing Date, as evidenced by the Companies' records. For the avoidance of doubt, the Parties hereby acknowledge and agree that the Buyer has a substantial and legitimate business interest throughout the Restraint Area.

(h) As used herein, the term "*Restraint Period*" means the period of two (2) years following the Closing Date.

9.2 <u>Protection of Confidential Information</u>. Recognizing the Buyer's need to protect the goodwill of the Business being purchased and to induce the Buyer to purchase the Business and the Purchased Shares, each Seller covenants and agrees with the Buyer that, if the transactions contemplated hereby are closed, no Seller will disclose or use or otherwise exploit for any Seller's own benefit, for the benefit of any other Person, or for the benefit of any competitor of any Company, or to harm or damage the Buyer or the Business, any Confidential Information (as hereinafter defined). The covenant contained in this Section 9.2 shall survive for a period of five (5) years following the Closing Date; provided, however, that with respect to those items of Confidential Information which constitute trade secrets under applicable law, the obligations of confidentiality and nondisclosure as set forth in this Section 9.2 shall continue to survive after said five-year period to the greatest extent permitted by applicable law. The rights of the Buyer contained in this <u>Section 9.2</u> are in addition to those rights the Buyer has under the common law or applicable statutes for the protection of trade secrets.

As used herein, the term "Confidential Information" shall mean and include all information, data and know-how of any Company used or useful in the Business and not generally known by a competitor of any Company (whether or not constituting a trade secret), including, without limitation, technical or non-technical data, formulas, patterns, compilations, mechanical and electrical designs, programs, applets, subroutines, libraries, objects and classes of objects, devices, methods, techniques, drawings, processes, financial data, financial plans, product plans, or a list of actual or potential customers or suppliers. Confidential Information shall not include any data or information that is publicly available or becomes publicly available through no action or fault of any Seller. In the event that any Seller shall be legally compelled or required by any Governmental Entity to disclose any Confidential Information, the Seller Representative shall promptly provide written notice to the Buyer to enable the Buyer to seek a protective order, in camera process or other appropriate remedy to avoid public or third-party disclosure of such Confidential Information. In the event that such protective order or other remedy is not obtained, the applicable Seller shall furnish only so much of such Confidential Information that he or she is legally compelled to disclose and shall exercise its commercially reasonable efforts to obtain reliable assurance that confidential treatment will be accorded such Confidential Information. The Seller shall cooperate with and assist Buyer in seeking any protective order or other relief requested by Buyer; Buyer shall reimburse to the Seller all reasonable expenses incurred by the Seller, including reasonable attorneys' fees, in cooperating and assisting the Buyer hereunder.

9.3 <u>Remedies</u>. Each Seller acknowledges that irreparable loss and injury would result to the Buyer or the Companies upon any breach of any of the covenants contained in <u>Section 9.1</u> or <u>9.2</u> and that damages arising out of such breach would be difficult to ascertain. Each Seller agrees that, in addition to all the remedies provided at law or at equity the Buyer may petition and seek from a court of law or equity, without bond, both temporary and permanent injunctive relief to prevent a breach by the Sellers of any such covenant.

9.4 <u>Blue-Penciling</u>. If any court determines that any one or more of the restrictive covenants contained in <u>Sections 9.1</u> or <u>9.2</u>, or any part thereof, is unenforceable because of the duration of such provision or the territory covered thereby, such court shall have the power to reduce the duration or territory of such provisions, and, in its reduced form, such provisions shall then be enforceable and shall be enforced.

9.5 <u>Independence</u>. The covenants on the part of the Sellers contained in this <u>Article IX</u> shall each be construed as agreements independent of each other and of any other provision in this Agreement and the unenforceability of one shall not affect the remaining covenants.

9.6 <u>Reasonable Restraint</u>. It is agreed by the parties that the covenants on the part of the Sellers contained in this <u>Article IX</u> are necessary for the legitimate business interests of Buyer and impose a reasonable restraint on the Sellers in light of the activities and business of Buyer on the date of the execution of this Agreement.

9.7 <u>Release</u>.

(a) Each Seller hereby forever fully and irrevocably releases each Company and its Subsidiaries (if any), Affiliates, predecessors, successors, assigns and past and present stockholders, directors, officers, employees, agents and representatives (collectively, the "*Released Parties*") from any and all actions, arbitrations, suits, petitions, appeals, demands, hearings, demand letters, inquiries, charges, sanctions, customer complaints, claims (both under contract or in tort), liens, notices of noncompliance or violation, investigations, proceedings, consent orders or consent agreements (collectively, "*Seller Claims*"), whether known or unknown, suspected or unsuspected, both at law and in equity, which such Seller now has, has ever had or may hereafter have against the Released Parties arising contemporaneously with or prior to the Closing Date or on account of or arising out of any matter, cause or event occurring contemporaneously with or prior to the Closing Date, whether from such Seller's status as a shareholder, director, officer, employee, creditor or otherwise of the Released Parties (including, without limitation, Seller Claims for damages, costs, expenses and attorneys', brokers' and accountants' fees and expenses), including, but not limited to, any rights to indemnification or reimbursement from any of the Released Parties, whether pursuant to their respective Organizational Documents, contract or otherwise and whether or not relating to Seller Claims pending on, or asserted after, the Closing Date (collectively, the "*Released Claims*"). Notwithstanding the foregoing, the foregoing release shall not operate to release (i) any claim of any Seller against the Released Parties to enforce the provisions of this Agreement or any other Transaction Document or (ii) any claim for insurance benefits under any employee insurance benefit plan, to the extent arising in the ordinary course since any Company's last payroll, or insurance benefit payment to such Seller, as the case may be.

(b) Each Seller hereby irrevocably agrees to refrain from, directly or indirectly, asserting any Released Claim or commencing, instituting or causing to be commenced any proceeding of any kind in any court or before any tribunal against any Released Party, based upon any Released Claim.

52

ARTICLE X CONDITIONS TO CLOSING

10.1 <u>Conditions to Obligations of All Parties</u>. The obligations of each Party to consummate the transactions contemplated by this Agreement shall be subject to the fulfillment, at or prior to the Closing, of each of the following conditions:

(a) No Governmental Entity shall have enacted, issued, promulgated, enforced or entered any Order which is in effect and has the effect of making the transactions contemplated by this Agreement illegal, otherwise restraining or prohibiting consummation of such transactions or causing any of the transactions contemplated hereunder to be rescinded following completion thereof.

10.2 <u>Conditions to Obligations of the Buyer</u>. The obligations of the Buyer to consummate the transactions contemplated by this Agreement shall be subject to the fulfillment, or the Buyer's waiver, at or prior to the Closing, of each of the following conditions:

(a) Other than the Fundamental Representations of the Sellers, the representations and warranties of the Sellers contained in this Agreement and the Transaction Documents shall be true and correct in all respects (in the case of any representation or warranty qualified by materiality or Material Adverse Effect) or in all material respects (in the case of any representation or warranty not qualified by materiality or Material Adverse Effect) on and as of the date hereof and on and as of the Closing Date with the same effect as though made at and as of such date (except those representations and warranties that address matters only as of a specified date, the accuracy of which shall be determined as of that specified date in all respects). The Fundamental Representations of the Sellers shall be true and correct in all respects on and as of the date hereof and on and as of the Closing Date with the same effect as though made at and

as of such date (except those representations and warranties that address matters only as of a specified date, the accuracy of which shall be determined as of that specified date in all respects). The Buyer shall have received a certificate signed by the Sellers, dated as of the Closing Date, certifying with respect to the foregoing matters in this Section 10.2(a).

(b) The Sellers shall have duly performed and complied in all material respects with all agreements, covenants and conditions required by this Agreement and the Transaction Documents to be performed or complied with by the Sellers prior to or on the Closing Date; <u>provided</u> that, with respect to agreements, covenants and conditions that are qualified by materiality, the Sellers shall have performed such agreements, covenants and conditions, as so qualified, in all respects. The Buyer shall have received a certificate signed by the Sellers, dated as of the Closing Date, certifying with respect to the foregoing matters in this <u>Section 10.2(b)</u>.

(c) No Proceeding shall have been commenced against any of the Sellers or the Companies, which would prevent the Closing. No injunction or restraining order shall have been issued by any Governmental Entity, and be in effect, which restrains or prohibits any transaction contemplated hereby.

(d) All approvals, consents and waivers that are listed on Schedule 2.4(a)(x) shall have been received, and executed counterparts thereof shall have been delivered to the Buyer at or prior to the Closing.

53

(e) Buyer shall have received the Audited Financial Statements with an unqualified opinion from the Auditor.

(f) From the date of this Agreement, there shall not have occurred any Material Adverse Effect, nor shall any event or events have occurred that, individually or in the aggregate, with or without the lapse of time, could reasonably be expected to result in a Material Adverse Effect.

(g) The Buyer shall have received from the Sellers all documents, certificates, instruments and any and all other deliverables required to be delivered by the Sellers in accordance with and pursuant to <u>Section 2.4(a)</u>.

10.3 <u>Conditions to Obligations of the Sellers</u>. The obligations of the Sellers to consummate the transactions contemplated by this Agreement shall be subject to the fulfillment or the Sellers' waiver, at or prior to the Closing, of each of the following conditions:

(a) Other than the Fundamental Representations of the Buyer, the representations and warranties of the Buyer contained in this Agreement and the Transaction Documents and any certificate or other writing delivered pursuant hereto shall be true and correct in all respects (in the case of any representation or warranty qualified by materiality or Material Adverse Effect) or in all material respects (in the case of any representation or warranty qualified by materiality or Material Adverse Effect) on and as of the date hereof and on and as of the Closing Date with the same effect as though made at and as of such date (except those representations and warranties that address matters only as of a specified date, the accuracy of which shall be determined as of that specified date in all respects). The Fundamental Representations of the Buyer shall be true and correct in all respects on and as of the date hereof and on and as of the Closing Date with the same effect as though made at and as of such date. The Seller shall have received a certificate signed by the Buyer, dated as of the Closing Date, certifying with respect to the foregoing matters in this <u>Section 10.3(a)</u>.

(b) The Buyer shall have duly performed and complied in all material respects with all agreements, covenants and conditions required by this Agreement and the Transaction Documents to be performed or complied with by it prior to or on the Closing Date; <u>provided</u> that, with respect to agreements, covenants and conditions that are qualified by materiality, the Buyer shall have performed such agreements, covenants and conditions, as so qualified, in all respects. The Sellers shall have received a certificate signed by the Buyer, dated as of the Closing Date, certifying with respect to the foregoing matters in this <u>Section 10.3(b)</u>.

(c) No injunction or restraining order shall have been issued by any Governmental Entity, and be in effect, which restrains or prohibits any material transaction contemplated hereby.

(d) The Sellers shall have received from the Buyer all documents, certificates, instruments and any and all other deliverables required to be delivered by the Buyer in accordance with and pursuant to <u>Section 2.4(b)</u>.

54

ARTICLE XI

TERMINATION

11.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing:

(a) by the mutual written consent of the Sellers and the Buyer;

(b) by the Buyer by written notice to the Seller Representative if:

(i) the Buyer is not then in material breach of any provision of this Agreement and there has been a breach, inaccuracy in or failure to perform any representation, warranty, covenant or agreement made by any of the Sellers pursuant to this Agreement that would give rise to the failure of any of the conditions specified in <u>Article X</u> and such breach, inaccuracy or failure has not been cured by the Sellers within twenty (20) days of the Seller Representative's receipt of written notice of such breach from the Buyer; or

(ii) any of the conditions set forth in <u>Section 10.1</u> or <u>Section 10.2</u> shall not have been, or if it becomes apparent that any of such conditions will not be, fulfilled by the Termination Date, unless such failure shall be due to the failure of the Buyer to perform or comply with any of the covenants, agreements or conditions hereof to be performed or complied with by it prior to the Closing;

(c) by the Seller Representative by written notice to the Buyer if:

(i) no Seller is then in material breach of any provision of this Agreement and there has been a breach, inaccuracy in or failure to perform any representation, warranty, covenant or agreement made by the Buyer pursuant to this Agreement that would give rise to the failure of any of the conditions specified in <u>Article X</u> and such breach, inaccuracy or failure has not been cured by the Buyer within twenty (20) days of Buyer's receipt of written notice of such breach from the Seller Representative; or

(ii) any of the conditions set forth in <u>Section 10.1</u> or <u>Section 10.3</u> shall not have been, or if it becomes apparent that any of such conditions will not be, fulfilled by the Termination Date, unless such failure shall be due to the failure of the Seller to perform or comply with any of the covenants, agreements or conditions hereof to be performed or complied with by it prior to the Closing; or

(d) by the Buyer or the Sellers in the event that (i) there shall be in effect a final nonappealable Order of a Governmental Entity of competent jurisdiction restraining, enjoining or otherwise prohibiting the consummation of the transactions contemplated hereby; it being agreed that the Parties hereto shall use commercially reasonable efforts to appeal any adverse determination which is not nonappealable (and pursue such appeal with reasonable diligence) and (ii) there shall be any Law that makes consummation of the transactions contemplated by this Agreement illegal or otherwise prohibited.

11.2 <u>Auditor Expense Amount Reimbursement</u>. If the Buyer terminates this Agreement pursuant to <u>Section 11.1</u> as a result of the Sellers' inability to deliver the Audited Financial Statements with an unqualified opinion from the Auditor, the Companies shall repay the Auditor Expense Amount in full to the Buyer by wire transfer of immediately available funds within five (5) Business Days of the termination of this Agreement.

55

11.3 <u>Effect of Termination</u>. In the event of the termination of this Agreement in accordance with this Article, this Agreement shall forthwith become void and there shall be no liability on the part of any party hereto except:

(a) as set forth in this $\underline{\text{Article XI}}$ or $\underline{\text{Article XII}}$; and

(b) that nothing herein shall relieve any party hereto from liability for any fraud, intentional misrepresentation or willful breach of any provision hereof.

ARTICLE XII GENERAL PROVISIONS

12.1 <u>Notices</u>. Any notice pursuant to this Agreement must be in writing and will be deemed effectively given to another Party on the earliest of the date: (a) three (3) Business Days after such notice is sent by registered U.S. mail, return receipt requested; (b) one (1) Business Day after delivery of such notice into the custody and control of an overnight courier service for next day delivery; (c) one (1) Business Day after delivery of such notice in person; (d) on the date of transmission if such notice is sent via electronic mail and receipt is confirmed by the recipient or no failure message is generated; and (e) the date such notice is received by that Party; in each case to

the appropriate address or email address below (or to such other address or email address as a Party may designate by notice to the other Parties):

(a) if to the Sellers or the Seller Representative:

[Insert Name of Applicable Seller] c/o Sam Andras, Seller Representative 7937 S. Quemoy Way Aurora, CO 80016 Email:

with a copy (which shall not constitute notice) to:

Miller & Law, P.C. 1900 W. Littleton Boulevard Littleton, CO 80120 Attn: David B. Law Email: dbl@millerandlaw.com

(b) if to the Buyer, to:

urban-gro, Inc. 1751 Panorama Point, Unit G Lafayette, CO 80023 Attn: Daniel Droller Email: dan.droller@urban-gro.com

with a copy (which shall not constitute notice) to:

Nelson Mullins Riley & Scarborough LLP 4140 Parklake Avenue, Second Floor Raleigh, NC 27612 Attn: David Mannheim Email: david.mannheim@nelsonmullins.com

56

A Party may designate a new address to which communications shall thereafter be transmitted by providing written notice to that effect to the other Parties.

12.2 <u>Fees and Expenses</u>. Each Party agrees to pay any and all expenses, fees and costs (including, legal, accounting and consulting expenses, fees and costs) incurred by it in connection with the transactions contemplated by this Agreement and any and all fees, commissions, compensation, reimbursement or other amounts due to any investment banker, broker or finder (or Person who performs a similar function) who claims to have been, or who was in fact, engaged by or on behalf of it in connection with the transactions contemplated by this Agreement; <u>provided</u> that any and all expenses, fees and costs (including, legal, accounting and consulting expenses, fees and costs) incurred by any Company shall be deemed to be paid by the Sellers at the Closing.

12.3 <u>Counterparts</u>. This Agreement may be signed in any number of counterparts (including by way of electronic transmission), each of which (when executed and delivered) shall constitute an original instrument, but all of which together shall constitute one and the same instrument, respectively. This Agreement shall become effective and be deemed to have been executed and delivered by both Parties at such time as counterparts hereto shall have been executed and delivered by both Parties, regardless of whether both Parties have executed the same counterpart. Counterparts may be delivered via facsimile or other electronic transmission and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

12.4 <u>Waivers</u>. No purported waiver of any provision of this Agreement shall be binding upon any of the Parties unless upon the Party providing such waiver has duly executed and delivered to the other Parties a written instrument which states that it constitutes a waiver of one or more provisions of this Agreement and specifies the provision(s) that are being waived. Any such waiver shall be

effective only to the extent specifically set forth in such written instrument. Neither the exercise (from time to time and at any time) by a Party of, nor the delay or failure (at any time or for any period of time) to exercise, any right, power or remedy shall constitute a waiver of the right to exercise, or impair, limit or restrict the exercise of, such right, power or remedy or any other right, power or remedy at any time and from time to time thereafter. No waiver of any right, power or remedy of a Party shall be deemed to be a waiver of any other right, power or remedy of such Party or shall, except to the extent so waived, impair, limit or restrict the exercise of such right, power or remedy.

12.5 <u>Amendments</u>. No purported amendment to any provision of this Agreement shall be binding upon the Parties unless the Buyer and the Sellers have each duly executed a written instrument which states that it constitutes an amendment to this Agreement and specifies the provision(s) that are being amended.

57

12.6 Entire Agreement. This Agreement, together with the Companies Disclosure Schedule hereto, the other Transaction Documents and any other documents expressly referred to herein, constitute the entire agreement among the Parties with respect to the subject matter hereof and supersede all of the previous or contemporaneous contracts, representations, warranties and understandings (whether oral or written) by or between the Parties with respect to the subject matter hereof, including any letter of intent or memorandum of terms entered into by the Parties.

12.7 <u>Severability</u>. If any provision of this Agreement shall hereafter be held to be invalid, unenforceable or illegal, in whole or in part, in any jurisdiction under any circumstances for any reason: (a) such provision shall be reformed to the minimum extent necessary to cause such provision to be valid, enforceable and legal while preserving the intent of the Parties as expressed in, and the benefits to such Parties provided by, such provision, or (b) if such provision cannot be so reformed, such provision shall be severed from this Agreement and an equitable adjustment shall be made to this Agreement (including addition of necessary further provisions to this Agreement) so as to give effect to the intent as so expressed and the benefits so provided. Such holding shall not affect or impair the validity, enforceability or legality of such provision in any other jurisdiction or under any other circumstances. Neither such holding nor such reformation nor severance shall affect or impair the legality, validity or enforceability of any other provision of this Agreement.

12.8 Governing Law. THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE DOMESTIC LAWS OF THE STATE OF DELAWARE WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION (WHETHER OF THE STATE OF DELAWARE OR ANY OTHER JURISDICTION) THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY JURISDICTION OTHER THAN THE STATE OF DELAWARE.

12.9 <u>Consent to Jurisdiction</u>. EACH PARTY AGREES THAT ANY AND ALL PROCEEDINGS ARISING OUT OF OR RELATING TO THIS AGREEMENT SHALL BE COMMENCED AND PROSECUTED IN A FEDERAL COURT LOCATED IN COLORADO, OR IN THE CASE OF A PROCEEDING ARISING OUT OF OR RELATING TO A THIRD PARTY CLAIM WHICH IS OR MAY BE SUBJECT TO INDEMNIFICATION HEREUNDER, IN THE COURT WHERE SUCH THIRD PARTY CLAIM IS BROUGHT AND EACH PARTY IRREVOCABLY WAIVES ANY RIGHT TO OBJECT TO SUCH VENUE. EACH PARTY CONSENTS AND SUBMITS TO THE NON-EXCLUSIVE PERSONAL JURISDICTION OF ANY SUCH FORUM IN RESPECT OF ANY SUCH PROCEEDING OR, WITH RESPECT TO A THIRD PARTY CLAIM, IN THE FORUM IN WHICH SUCH THIRD PARTY CLAIM WAS BROUGHT. EACH PARTY CONSENTS TO SERVICE OF PROCESS UPON IT WITH RESPECT TO ANY SUCH PROCEEDING BY REGISTERED MAIL, RETURN RECEIPT REQUESTED, AND BY ANY OTHER MEANS PERMITTED BY APPLICABLE LAWS.

12.10 <u>Waiver of Jury Trial</u>. Each Party acknowledges and agrees that any controversy which may arise under this Agreement or the Transaction Documents is likely to involve complicated and difficult issues and, therefore, each such Party irrevocably and unconditionally waives any right it may have to a trial by jury in respect of any legal action arising out of or relating to this Agreement or the Transaction Documents or the transactions contemplated hereby or thereby. Each Party to this Agreement certifies and acknowledges that (a) no representative of any other Party has represented, expressly or otherwise, that such other Party would not seek to enforce the foregoing waiver in the event of a legal action, (b) such Party has considered the implications of this waiver, (c) such Party makes this waiver voluntarily, and (d) such Party has been induced to enter into this Agreement by, among other things, the mutual waivers and certifications in this <u>Section 12.10</u>.

12.11 <u>Assignment</u>. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the Parties and their respective successors, heirs, estates and permitted assigns; <u>provided</u> that neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned by any Party without the prior written consent of the other Parties. Notwithstanding the immediately preceding sentence, without the prior written consent of the Sellers, the Buyer may at any time, in its sole discretion, assign, in whole or in part, its rights under this Agreement and any other Transaction Documents for collateral security purposes to any lender providing financing to it or any of its Affiliates. Subject to the preceding sentences of this <u>Section 12.11</u>, this Agreement will be binding upon the Parties and their respective successors, heirs, estates and permitted assigns and shall inure to the benefit of and be enforceable by the Parties and their respective successors, heirs, estates and permitted assigns.

12.12 <u>Remedies</u>. Except as otherwise expressly provided in <u>Article VIII</u>, each of the Parties shall have and retain all rights and remedies, at law or in equity, including rights to specific performance and injunctive or other equitable relief, arising out of or relating to a breach or threatened breach of this Agreement. Without limiting the generality of the foregoing, each of the Parties acknowledges that money damages would not be a sufficient remedy for any breach or threatened breach of this Agreement and that irreparable harm would result if this Agreement were not specifically enforced. Therefore, the rights and obligations of the Parties shall be enforceable by a decree of specific performance issued by any court of competent jurisdiction, and appropriate injunctive relief may be applied for and granted in connection therewith, without the necessity of posting a bond or other security or proving actual damages and without regard to the adequacy of any remedy at law. A Party's right to specific performance shall be in addition to all other legal or equitable remedies available to such Party. The prevailing Party in any Proceeding to enforce any provision of this Agreement shall be entitled to recover from the losing party (or parties) all reasonable attorneys' fees and expenses.

12.13 <u>Third Party Beneficiaries</u>. No Person other than the Buyer, the Seller Representative and the Sellers is or is intended to be a beneficiary of this Agreement, other than the Buyer Indemnified Parties and successors, heirs, estates and assigns of the Parties permitted as provided in <u>Section 12.11</u>.

12.14 <u>Further Assurances</u>. At any time and from time to time after the Closing, each of the Parties, at their own cost and expense, in good faith and in a timely manner, shall use their respective commercially reasonable efforts to take or cause to be taken all appropriate actions, do or cause to be done all things necessary, proper or advisable, and execute, deliver and acknowledge such documents and other papers as may be required to carry out the provisions of this Agreement and to give effect to the consummation of the transactions contemplated by this Agreement.

12.15 <u>Interpretation</u>. The language used in this Agreement shall be deemed to be the language chosen by the Parties to express their mutual intent and no rule of strict construction shall be applied against any Party. Unless otherwise expressly specified in this Agreement:

(a) the words "<u>hereof</u>," "<u>hereby</u>" and "<u>hereunder</u>," and correlative words, refer to this Agreement as a whole and not any particular provision;

(b) the words "<u>includes</u>" and "<u>including</u>," and correlative words, are deemed to be followed by the phrase "without limitation";

(c) the word "<u>or</u>" is not exclusive and is deemed to have the meaning "<u>and/or</u>";

(d) words using the singular or plural number shall also include the plural or singular number, respectively;

(e) the section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement;

(f) the masculine, feminine or neuter form of a word includes the other forms of such word and the singular form of a word includes the plural form of such word;

(g) references to a Person shall include the successors, heirs, estates and assigns thereof; and

(h) references made in this Agreement to an Article, Section, Schedule or Exhibit mean an Article or Section of, or a Schedule or Exhibit to, this Agreement.

12.16 Appointment of Seller Representative.

(a) <u>Designation</u>. The Seller Representative is hereby designated by each of the Sellers to serve as the representative of the Sellers with respect to the matters set forth in this Agreement and the Escrow Agreement to be performed by the Seller Representative.

(b) Authority. By the approval of this Agreement, the Sellers hereby irrevocably constitute and appoint the Seller Representative as the representative, agent, proxy and attorney-in-fact for each of the Sellers for all purposes authorized under this Agreement and the Escrow Agreement, including the full power and authority on the Sellers' behalf (i) to consummate the transactions contemplated in this Agreement; (ii) to pay the Seller Representative's expenses incurred in connection with the negotiation and performance of this Agreement and the Escrow Agreement (whether incurred on or after the date hereof); (iii) to disburse any funds received hereunder to such Seller and each other Seller; (iv) to endorse and deliver any certificates or instruments representing the Purchased Shares and execute such further instruments of assignment as the Buyer shall reasonably request; (v) to negotiate, execute and deliver on behalf of such Seller any amendment or waiver hereto or to the Escrow Agreement; (vi) (A) to dispute or refrain from disputing, on behalf of such Seller relative to any amounts to be received by such Seller under this Agreement or any agreements contemplated hereby (including the Escrow Agreement), any claim made by the Buyer under this Agreement or other agreements contemplated hereby (including the Escrow Agreement), (B) to negotiate and compromise, on behalf of such Seller, any dispute that may arise under, and exercise or refrain from exercising any remedies available under, this Agreement or any other agreement contemplated hereby (including the Escrow Agreement), and (C) to execute, on behalf of such Seller, any settlement agreement, release or other document with respect to such dispute or remedy; (vii) to engage attorneys, accountants, agents or consultants on behalf of the Sellers in connection with this Agreement or any other agreement contemplated hereby (including the Escrow Agreement) and paying any fees related thereto; (viii) to take all other actions to be taken by or on behalf of such Seller in connection herewith including any action to enforce the rights of such Seller hereunder; and (ix) to do each and every act and exercise any and all rights which such Seller or the Sellers collectively are permitted or required to do or exercise under this Agreement, the Escrow Agreement or any other agreement contemplated hereby. Each of the Sellers agrees that such agency and proxy are coupled with an interest, are therefore irrevocable without the consent of the Seller Representative and shall survive the death, incapacity, bankruptcy, dissolution or liquidation of any Seller. All decisions and actions by the Seller Representative (to the extent authorized by this Agreement) shall be binding upon all of the Sellers, and no Seller shall have the right to object, dissent, protest or otherwise contest the same.

(c) <u>Reliance</u>. Each Seller agrees that the Buyer shall be entitled to rely on any action taken by the Seller Representative, on behalf of such Seller, pursuant to <u>Section 12.16(b)</u> (an "*Authorized Action*"), and that each Authorized Action shall be binding on each Seller as fully as if such Seller had taken such Authorized Action.

(d) <u>Acceptance</u>. The Seller Representative hereby accepts the foregoing appointment and agrees to serve as Seller Representative, subject to the provisions hereof, for the period of time from and after the date hereof without compensation except for the reimbursement from the Sellers of fees and expenses incurred by Seller Representative in his capacity as such. The Seller Representative shall act for the Sellers on all of the matters set forth in this Agreement and the Escrow Agreement in the manner the Seller Representative believes to be in the best interest of the Sellers and is authorized to act on behalf of the Sellers notwithstanding any dispute or disagreement between the Sellers. The Seller Representative shall not be liable for any act done or omitted hereunder as the Seller Representative harmless from and against any loss, liability or expense incurred without negligence or bad faith on the part of the Seller Representative and arising out of or in connection with the acceptance or administration of the Seller Representative's duties hereunder, including the reasonable fees and expenses of any legal counsel retained by the Seller Representative.

[remainder of page intentionally left blank; signature page(s) follow]

60

IN WITNESS WHEREOF, the Parties have duly executed this Stock Purchase Agreement as of the date first set forth above.

<u>SELLERS</u>:

/s/ Sam Andras

Sam Andras

/s/ Scott Allen

Scott Allen

/s/ Michael Starr Michael Starr /s/ Geoff McKinney Geoff McKinney /s/ Sandra Schaaf Sandra Schaaf /s/ Emily Flournoy Emily Flournoy

> /s/ Sara Flora Sara Flora

SELLER REPRESENTATIVE:

/s/ Sam Andras

Sam Andras

[Stock Purchase Agreement]

BUYER:

URBAN-GRO ARCHITECT HOLDINGS, LLC

By: /s/ Richard Akright

Name: Richard Akright Title: Chief Executive Officer

PARENT:

URBAN-GRO, INC.

By: /s/ Bradley Nattrass

Name: Bradley Nattrass Title: Chief Executive Officer

[Stock Purchase Agreement]

Exhibit A

Definitions:

"<u>2WRCO</u>" has the meaning set forth in the recitals.

"2WRCO Shareholders" has the meaning set forth in the recitals.

"2WRCO Shares" has the meaning set forth in the recitals.

"2WRGA" has the meaning set forth in the recitals.

"2WRGA Shareholders" has the meaning set forth in the recitals.

"2WRGA Shares" has the meaning set forth in the recitals.

"<u>2WRMS</u>" has the meaning set forth in the recitals.

"2WRMS Shareholders" has the meaning set forth in the recitals.

"2WRMS Shares" has the meaning set forth in the recitals.

"Accountant Representative" has the meaning set forth in Section 2.2(c).

"<u>Accounting Firm</u>" has the meaning set forth in <u>Section 2.2(c)</u>.

"Acquisition Proposal" has the meaning set forth in Section 6.4(a).

"<u>Affiliate</u>" of any Person means another Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with, such first Person, with "<u>control</u>" of a Person meaning the possession, directly or indirectly, of the right or power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, through contract rights or otherwise, and "<u>controlling</u>" and "<u>controlled</u>" having correlative meanings.

"<u>Affordable Care Act</u>" means the Patient Protection and Affordable Care Act, as amended by the Health Care and Education Reconciliation Act.

"<u>Agreement</u>" has the meaning set forth in the preamble.

"Asset Transfer Agreement" has the meaning set forth in Section 2.4(a)(vii).

"<u>Assumed Indebtedness</u>" means (i) the Funded Indebtedness of the Companies set forth on <u>Schedule A</u>, (ii) all outstanding or accrued vacation, paid time off or other leave or bonus, incentives, profit sharing or commissions (including the pro rata portion of bonuses, incentives or commissions for any partial periods not completed as of the Closing Date) earned by the employees of the Companies that is unused or unpaid, as applicable, prior to the Closing Date, in each case as set forth on <u>Schedule A</u>, and (iii) any unpaid Taxes owed with respect to periods prior to the Closing Date (including any payments deferred pursuant to the CARES Act, or any other federal, state, local or foreign law), any Tax credits claimed under the CARES Act or the FFCRA for periods prior to the Closing Date that are following the Closing disallowed or otherwise not available to the Companies, and any "advance payment of employer credits" requested on IRS Form 7200 with respect to that portion of the calendar quarter occurring prior to the Closing.

"Attorney-in-Fact" has the meaning set forth in Section 2.6(b).

"<u>Audit Expense Amount</u>" shall mean an amount equal to \$75,000 advanced by the Buyer to the Companies to be applied by the Companies to the payment of the fees and expenses of the Companies related to the preparation of the Audited Financial Statements.

"Audited Financial Statements" has the meaning set forth in Section 6.2.

"Auditor" means BF Borgers CPA PC.

"<u>Authorized Action</u>" has the meaning set forth on <u>Section 12.16(c)</u>.

"<u>Baseline Closing Statement</u>" has the meaning set forth in <u>Section 2.2(b)</u>.

"Business" means the business of the Companies including, but not limited to, the business of providing architectural services, including, but not limited to master planning programming, architectural design, LEED integration, graphics and way-finding, fundraising support, building assessment, Americans with Disability Act surveys and construction administration.

"<u>Business Day</u>" means any day of the year other than: (a) any Saturday or Sunday; or (b) any other day on which the banks located in the State of Delaware generally are closed for business.

"<u>Buyer</u>" has the meaning set forth in the preamble.

"Buyer Indemnified Party" has the meaning set forth in Section 8.1(b).

"Calculation Period" has the meaning set forth in Section 2.5(a)(i).

"<u>CARES Act</u>" means the Coronavirus Aid, Relief, and Economic Security Act, Pub. L. 116–136 (116th Cong.) (Mar. 27, 2020), together with all amendments thereto and the statutes, rules and regulations promulgated thereunder and any successor to such statutes, rules or regulations, as in effect on the date hereof.

"<u>Cares Act Permitted Purposes</u>" means, with respect to the use of proceeds of any Company PPP Loan, the purposes set forth in the CARES Act, provided such purposes are otherwise in compliance with all other provisions or requirements of the CARES Act.

"Cash" means all cash of the Companies on a consolidated basis, calculated in accordance with GAAP. For avoidance of doubt, Cash will be calculated net of escrow amounts or other restricted cash balances.

"Change of Control" has the meaning set forth in Section 2.5(a)(ii).

A-2

"<u>Claim</u>" means any complaint, allegation, charge, petition, appeal, demand, notice, filing or claim of any kind that commences, alleges a basis to commence or threatens to commence any Proceeding by or before any Governmental Entity or that asserts, alleges a basis to assert or threatens to assert any right, breach, default, violation, noncompliance, termination, cancellation or other action or omission that could result in a Loss.

"<u>Closing</u>" has the meaning set forth in <u>Section 2.3(a)</u>.

"Closing Date" has the meaning set forth in Section 2.3(a).

"Closing Payment Shares" has the meaning set forth in Section 2.1(c).

"COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.

"Code" means the Internal Revenue Code of 1986, as amended.

"Companies" and "Company" have the meanings set forth in the recitals.

"Companies 401(k) Plan" means that certain 2WR Partners 401(k) Plan.

"<u>Companies Disclosure Schedule</u>" means the disclosure schedules delivered by the Sellers to the Buyer concurrently with the execution of this Agreement and forming a part hereof.

"Companies Permits" has the meaning set forth in Section 3.15(b).

"Companies Shares" has the meaning set forth in Section 4.2.

"Companies Tangible Property" has the meaning set forth in Section 3.11(d).

"Company PPP Loans" has the meaning set forth in Section 3.28.

"Company-Owned Vehicles" has the meaning set forth in Section 6.10.

"Confidential Information" has the meaning set forth in Section 9.2.

"<u>Contract</u>" means any contract, agreement, note, bond, mortgage, indenture, deed of trust, lease, sublease, license, sublicense, purchase or sale order, quotation, arrangement, letter of agency or other commitment, obligation or instrument, whether written or oral, that is binding or enforceable upon the parties thereto.

"<u>COVID-19 Liabilities</u>" means all Liabilities of any Company arising out of, with respect to or by reason of facts, conditions, events or circumstances occurring on or before the Closing Date and related to the COVID-19 pandemic, including, but not limited to, all such Liabilities arising out of or with respect to or by reason of (i) all obligations under Laws (in effect at or prior to the Closing) adopted in response the COVID-19 pandemic, including, but not limited to (A) applicable provisions of FFCRA, (B) applicable provisions of the CARES Act, (C) any quarantine, stay-at-home, shelter in place, lockdown, social distancing, or similar order issued in response to the COVID-19 pandemic and applicable to any Company and (D) federal, state and local health and safety or employment laws and orders applicable to any Company and (ii) any employees that have tested positive for COVID-19.

A-3

"Definitive Closing Statement" has the meaning set forth in Section 2.2(c).

"Dispute Notice" has the meaning set forth in Section 2.2(c).

"<u>Disputed Receivables Amount</u>" means the aggregate amount of accounts receivable alleged to be owed to any Company in a Proceeding listed on <u>Schedule 3.7</u> as of the Closing Date.

"Earnout Calculation" has the meaning set forth in Section 2.5(c)(i).

"Earnout Calculation Objection Notice" has the meaning set forth in Section 2.5(c)(ii).

"Earnout Calculation Statement" has the meaning set forth in Section 2.5(c)(i).

"Earnout Payment" has the meaning set forth in Section 2.5(b).

"Earnout Period" has the meaning set forth in Section 2.5(a)(iii).

"Earnout Shares" has the meaning set forth in Section 2.5(e).

"Employee Benefit Plans" is defined in Section 3.9(a).

"Employment Agreements" has the meaning set forth in Section 2.4(a)(iv).

"Environmental and Occupational Safety and Health Liability" means any Liability that is or may be imposed on the Buyer or any of its Affiliates or any of their assets that arises from or under any Environmental Law or Occupational Safety and Health Law with respect to the ownership or operation by any Person at any time on or prior to the Closing Date of any of the Companies or the Business, including those consisting of or relating to:

(a) any environmental, health or safety matter or condition (including on-site or off-site contamination, occupational safety and health and regulation of any chemical substance or product);

(b) any fine, penalty, judgment, award, settlement, legal or administrative proceeding, damages, loss, claim, demand or response, remedial or inspection cost or expense arising under any Environmental Law or Occupational Safety and Health Law;

(c) financial responsibility under any Environmental Law or Occupational Safety and Health Law for cleanup costs or corrective action, including any cleanup, removal, containment or other remediation or response actions ("*Cleanup*") required by any Environmental Law or Occupational Safety and Health Law (whether or not such Cleanup has been required or requested by any Governmental Entity or any other Person) and for any natural resource damages; or

(d) any other compliance, corrective or remedial measure required under any Environmental Law or Occupational Safety and Health Law.

"<u>Environmental Authority</u>" shall mean any federal, state, regional, county or local government, agency or authority or any court in each case having judicial, regulatory or administrative authority under Environmental Laws.

"Environmental Conditions" shall mean any environmental contamination or pollution or threatened contamination or pollution of, or the Release or threatened Release of Hazardous Materials into, the surface water, groundwater, surface soil, subsurface soil, air and land.

"Environmental Laws" shall mean all federal, regional, state, county or local laws, statutes, ordinances, decisional law, rules, regulations, codes, orders, decrees, directives and judgments relating to public health or safety, pollution, damage to or protection of the environment, Environmental Conditions, Releases or threatened Releases of Hazardous Materials into the environment or the use, manufacture, processing, distribution, treatment, storage, generation, disposal, transport or handling of Hazardous Materials, whether existing in the past or present. Environmental Laws shall include, but are not limited to, the following laws, and the regulations promulgated thereunder, as the same have been amended from time to time: CERCLA, RCRA, the Clean Air Act (42 U.S.C. 7401 et seq.), and the Clean Water Act (33 U.S.C. 1251 et seq.).

"Environmental Permits" shall mean all Permits required under or issued pursuant to Environmental Laws.

"Environmental Reports" has the meaning set forth in Section 3.13(b).

"Equitable Exceptions" means any limitations on the enforceability of obligations resulting from: (a) bankruptcy, insolvency, reorganization, moratorium or other requirements of Laws, Orders or equitable principles now or hereafter in effect relating to or affecting the enforcement of creditors' rights or debtors' obligations generally; and (b) as to the remedy of specific performance and injunctive and other forms of equitable relief, the imposition of equitable defenses and the discretion of a Governmental Entity before which any Proceeding therefor may be brought.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"ERISA Affiliate" is defined in Section 3.9(a).

"ERISA Plans" is defined in Section 3.9(a).

"Escrow Agent" means Equiniti Trust Company.

"Escrowed Shares" has the meaning set forth in Section 2.1(c)(i).

"Estimated Assumed Indebtedness" has the meaning set forth in Section 2.2(a).

"Estimated Closing Cash" has the meaning set forth in Section 2.2(a).

"Estimated Closing Statement" has the meaning set forth in Section 2.2(a).

"Estimated Working Capital" has the meaning set forth in Section 2.2(a).

"Excess Amount" has the meaning set forth in Section 2.2(d).

A-4

A-5

"Excess Payment" has the meaning set forth in Section 2.2(d).

"Exchange" means the Nasdaq Capital Market exchange.

"<u>FFCRA</u>" means the Families First Coronavirus Response Act, Pub. L. No. 116-127 (116th Cong.) (Mar. 18, 2020), together with all amendments thereto and the statutes, rules and regulations promulgated thereunder and any successor to such statutes, rules or regulations, as in effect on the date hereof.

"Final Purchase Price" has the meaning set forth in Section 2.2(d).

"<u>Financial Statements</u>" has the meaning set forth in <u>Section 3.4(a)</u>.

"Flow of Funds Memorandum" has the meaning set forth in Section 2.1(a).

"Fundamental Representations" has the meaning set forth in Section 8.1(a).

"<u>Funded Indebtedness</u>" means the sum of (i) all amounts owing under any Indebtedness of any of the Companies (including principal, interest, prepayment penalties or fees, premiums, breakage amounts, expense reimbursements or other amounts payable in connection with prepayment),(ii) all long-term obligations to contractors or consultants of the Companies (to the extent not already included in Indebtedness) and (iii) all amounts required to obtain the release of Liens in favor of any party on any assets of the Companies.

"<u>GAAP</u>" means generally accepted accounting principles in the United States as in effect from time to time, applied on a basis consistent with the Companies' historical practice.

"Government Contract" has the meaning set forth in Section 3.23.

"Governmental Entity" means any government (including, but not limited to, any United States or foreign federal, state, provincial, municipal or county government), any political subdivision thereof, any governmental, administrative, ministerial, regulatory, central bank, self-regulatory, quasi-governmental, taxing, executive, or legislative branch, department, commission, body, agency, authority or instrumentality of any of the foregoing and any court, arbitrator, special master, receiver, tribunal or similar body of any kind.

"<u>Hazardous Material</u>" shall mean chemicals, products, compounds, by-products, pollutants, contaminants, hazardous wastes or toxic or hazardous substances regulated under any Environmental Law, including, but not limited to, asbestos or asbestos-containing materials, polychlorinated biphenyls, pesticides and oils, petroleum and petroleum products.

"Income Taxes" means any federal, state, local or foreign Tax that, in whole or in part, is based on, measured by or calculated by reference to net income, profits or gains, including any state or local franchise Tax, and including any interest, penalty or addition thereto, whether disputed or not.

A-6

"Indebtedness" of any Person means, without duplication, (i) the principal, accreted value, accrued and unpaid interest, prepayment and redemption premiums or penalties (if any), unpaid fees or expenses and other monetary obligations in respect of (A) indebtedness of such Person for money borrowed and (B) indebtedness evidenced by notes, debentures, bonds or other similar instruments for the payment of which such Person is responsible or liable; (ii) all obligations of such Person issued or assumed as the deferred purchase price of property, all conditional sale obligations of such Person and all obligations of such Person under any title retention agreement (but excluding trade accounts payable and other accrued current liabilities arising in the Ordinary Course of Business (other than the current liability portion of any indebtedness for borrowed money)); (iii) all obligations of such Person under leases required to be capitalized in accordance with GAAP; (iv) all obligations of such Person for the reimbursement of any obligor on any letter of credit, banker's acceptance or similar credit transaction; (v) all obligations of such Person under interest rate or currency swap transactions (valued at the termination value thereof); (vi) the liquidation value, accrued and unpaid dividends; prepayment or redemption premiums and penalties (if any), unpaid fees or expenses and other monetary obligations in respect of any redeemable preferred stock or equity of such Person; (vii) all obligations of such Person under any deferred or incentive compensation plan; (viii) all obligations of the type referred to in clauses (i) through (vii) of any Persons for the payment of which such Person is responsible or liable, directly or indirectly, as obligor, guarantor, surety or otherwise, including guarantees of such obligations; and (x) all obligations of the type referred to in clauses (i) through (viii) of other Persons secured by (or for which the holder of such obligations has an existing right, contingent or otherwise, to be secured by) any Lien on any property or asset of such Person (whether or not such obligation is assumed by such Person), provided, however, that trade accounts payable or operating leases arising in the Ordinary Course of Business (other than the current liability portion of any indebtedness for borrowed money) shall not be "Indebtedness" hereunder.

"Indemnified Party" has the meaning set forth in Section 8.2(a).

"Indemnifying Party" has the meaning set forth in Section 8.2(a).

"Initial Purchase Price" has the meaning set forth in Section 2.1.

"Intellectual Property" means (i) all inventions (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents and patent applications, together with all reissuances, continuations, continuations-in-part, revisions, extensions, and reexaminations thereof, (ii) all trademarks, service marks, trade dress, logos, trade names, domain names and corporate names, together with all translations, adaptations, derivations, and combinations thereof and including all goodwill associated therewith, and all applications, registrations and renewals in connection therewith, (iii) all copyrights, and all applications, registrations, and renewals in connection therewith, (iv) all trade secrets and confidential business information (including ideas, research and development, know-how, formulas, compositions, manufacturing and production processes and techniques, designs, drawings, specifications, technical data, customer and supplier lists, pricing and cost information, and business and marketing plans and proposals), (v) all computer software, (vi) all database rights, (vii) all design rights and registered designs and all documentation and media constituting or describing any of the foregoing and all copies and tangible embodiments thereof (in whatever form or medium and whether or not any of the foregoing is registered), and (viii) all other proprietary rights, including all moral rights, pertaining to any product or service designed, manufactured, sold, distributed, marketed, used, performed, employed or exploited, and all rights or forms of protection of a similar nature or having equivalent or similar effect to any of those which may subsist anywhere in the world, owned by or registered in the name of any Person or in which any Person has any rights, licenses or immunities.

A-7

"IRS" means the Internal Revenue Service of the United States.

"<u>IT Systems</u>" means all software, computer hardware, servers, networks, platforms, peripherals, and similar or related items of automated, computerized, or other information technology (IT) networks and systems (including telecommunications networks and systems for voice, data and video) owned, leased, licensed, or used (including through cloud-based or other third-party service providers) by the Companies.

"Key Employees" means Sam Andras, Michael Starr, Geoff McKinney and Scott Allen.

"Knowledge of the Companies" and correlative phrases such as "the Companies' Knowledge" means the actual and constructive knowledge of the Majority Shareholders and each officer of the Companies after reasonable inquiry.

"Law" means any treaty, code, statute, law (including common law), rule, regulation, convention, ordinance, Order, regulatory policy statement or similar guidance, binding directive or decree of any kind of any Governmental Entity.

"Leased Real Property" has the meaning set forth in Section 3.11(b).

"Liability" means any debt, loss, damage, adverse claim, fines, penalties, liability or obligation (whether direct or indirect, known or unknown, asserted or unasserted, absolute or contingent, accrued or unaccrued, matured or unmatured, determined or determinable, liquidated or unliquidated, or due or to become due, and whether in contract, tort, strict liability or otherwise, and including all costs and expenses relating thereto).

"License" has the meaning set forth in Section 3.12(e).

"Lien" means any mortgage, pledge, security interest, encumbrance, charge or lien, other than: (a) mechanics', materialmen's and similar liens not yet due and payable; (b) liens for Taxes not yet due and payable; (c) liens arising under workers' compensation, unemployment insurance, social security, retirement or similar legislation; and (d) purchase money liens and license securing rental payments under capital lease arrangements.

"Loss" or "Losses" means any claim, Liability, obligation, deficiency, damage (including incidental, consequential and punitive damages), penalty, fine, assessment, cost (including any opportunity cost), settlement payment, Tax, diminution in value, expense, fee, court costs or attorneys' fees and expenses.

"Majority Shareholders" means Sam Andras, Scott Allen, Michael Starr and Geoff McKinney.

"<u>Management Services Agreement</u>" has the meaning set forth in <u>Section 2.4(a)(v)</u>.

"<u>Material Adverse Effect</u>" means any change or effect that, individually or in the aggregate, has a material adverse effect on the Business, assets, condition (financial or otherwise), prospects or operating results of the Companies on a consolidated basis.

"Material Contracts" has the meaning set forth in Section 3.6.

"Minimum Closing Cash" has the meaning set forth in Section 2.2(a).

"MJ12" has the meaning set forth in the recitals.

"MJ12 Shareholders" has the meaning set forth in the recitals.

"MJ12 Shares" has the meaning set forth in the recitals.

"Notice of Claim" has the meaning set forth in Section 8.2(a).

"Occupational Safety and Health Law" means any Law designed to provide safe and healthful working conditions and to reduce occupational safety and health hazards, including the Occupational Safety and Health Act, and any program, whether governmental or private (such as those promulgated or sponsored by industry associations and insurance companies), designed to provide safe and healthful working conditions.

"Order" means any judgment, writ, decree, directive, decision, injunction, ruling, award or order (including any consent decree or cease and desist order) of any kind of any Governmental Entity.

"Ordinary Course of Business" means the ordinary course of business consistent with past custom and practice of the Business (including with respect to quantity and frequency).

"Organizational Documents" of a Person means: (a) its articles of incorporation, certificate of incorporation, certificate of formation or similar document(s) filed with a Governmental Entity, which filing forms or organizes the Person; and (b) its bylaws, limited liability company operating agreement, partnership agreement, trust agreement or similar document(s), whether or not filed with a Governmental Entity, which organize and/or govern the internal affairs of such Person, in the case of each of clause (a) and (b) above, as in effect at the time in question.

"<u>Parent</u>" has the meaning set forth in the Preamble.

"Parent Common Stock" has the meaning set forth in Section 2.1(c).

"Parent SEC Reports" has the meaning set forth in Section 5.5(a).

"Parent Shares" has the meaning set forth in Section 2.5(e).

"<u>Parties</u>" or "<u>Party</u>" has the meaning set forth in the Preamble.

"Permit" means any franchise, license, approval, authorization, certificate of public convenience and necessity, waiver, certification or permit issued or granted by any Governmental Entity.

A-9

"<u>Person</u>" includes an individual, a partnership, a sole proprietorship, a company, a firm, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization, a union, a group acting in concert, a Governmental Entity or any other entity or association of any kind.

"PPP Escrow Agreement" has the meaning set forth in Section 2.1(d).

"<u>PPP Lender</u>" has the meaning set forth in <u>Section 2.1(d)</u>.

"PPP Loan" has the meaning set forth in Section 3.28.

"Pre-Closing Period" has the meaning set forth in Section 7.1(a).

"Proceeding" means any action, suit, arbitration, mediation, litigation, hearing, investigation, inquiry or other proceeding of any kind involving any Governmental Entity or any other Person.

"<u>Purchase Price</u>" means an amount equal to \$7,100,000 *less* (i) the Assumed Indebtedness as set forth on the Definitive Closing Statement *plus* (ii) the amount, if any, by which the Working Capital as set forth on the Definitive Closing Statement is greater than the Working Capital Target *less* (iii) the amount, if any, by which the Working Capital as set forth on the Definitive Closing Statement is less than the Working Capital Target *less* (iv) the amount, if any, by which the Companies' Cash as of the Closing as set forth on the Definitive Closing Statement is less than the Minimum Closing Cash.

"Purchased Shares" has the meaning set forth in the recitals.

"Qualified Leave Wages" means, collectively, any "qualified sick leave wages" and any "qualified family leave wages" as defined in sections 7001 and 7003 of the FFCRA.

"Qualified Health Plan Expenses" has the meaning provided in section 7001 of the FFCRA.

"Quarterly Gross Profit" has the meaning set forth in Section 2.5(a)(iv).

"<u>Release</u>" means any intentional or unintentional release, discharge, spill, leaking, pumping, pouring, emitting, emptying, injection, disposal or dumping.

"Restraint Area" has the meaning set forth in Section 9.1(g).

"<u>Restraint Period</u>" has the meaning set forth in <u>Section 9.1(h)</u>.

"<u>Restricted Party</u>" means a Person (a) that is, or is owned or controlled by a Person that is, listed on a Sanctions List; (b) located or resident in any country or territory, or organized under the laws of any jurisdiction that is, or whose government is the subject of Sanctions broadly prohibiting dealings with such government, country or territory, or a Person that is, or is owned or controlled by Persons that are, the target of Sanctions; (c) with which any of the Companies or the business or operations conducted by it is prohibited by Sanction from dealing or otherwise engaging in any transaction; or (d) acting or purporting to act on behalf of any of the Persons listed in clauses (a) through (c) above.

A-10

"Review Period" has the meaning set forth in Section 2.5(c)(ii).

"Sample Working Capital Calculation" has the meaning set forth in Section 2.2(a).

"Sanctions" means any trade, financial or other economic sanctions Laws, embargoes, or restrictive measures administered, enacted, imposed, or enforced by any Sanctions Authority.

"<u>Sanctions Authority</u>" means (a) the United States government; (b) Canada; (c) the United Nations (including the United Nations Security Council); (d) the European Union; (e) the United Kingdom; (f) the respective Governmental Entities of any of the foregoing, including OFAC, BIS the United States Department of State and Her Majesty's Treasury; and (g) any Governmental Entity with similar sanctions authority.

"<u>Schedule Supplement</u>" has the meaning set forth in <u>Section 6.5(c)</u>.

"Securities Act" means the Securities Act of 1933, as amended.

"<u>Seller</u>" or "<u>Sellers</u>" has the meaning set forth in the preamble.

"Seller Indemnified Party" has the meaning set forth in Section 8.1(e).

"Seller Representative" has the meaning set forth in the preamble.

"Sellers Allocation Schedule" has the meaning set forth in Section 2.1(c)(ii).

"Share Transfer Agreement" has the meaning set forth in Section 2.4(a)(vi).

"<u>Shareholder Claim</u>" means any Claim by, with respect to or arising in connection with, any current or former holder of capital stock, option or any other equity or debt security of any of the Companies concerning any matter, fact or event that existed or occurred or may have existed or occurred at any time through and including the Closing Date.

"Short Sales" means all "short sales" as defined in Rule 200 of Regulation SHO under the Exchange Act.

"Straddle Period" has the meaning set forth in Section 7.1(b).

"Subsidiary" means any Person with respect to which any of the Companies has or had the power to vote or direct the voting of sufficient securities to be able to elect the board of directors or similar body.

A-11

"Tax" or "Taxes" means (a) any foreign, United States federal, state or local tax, including but not limited to net income, alternative or add-on minimum tax, gross income, gross receipts, sales, use, ad valorem, capital, value added, transfer, franchise, profits, excess profits, license, withholding, payroll, employment, FICA, FUTA, social security, unemployment, stamp, escheat, excise, severance, real property, personal property, intangible property, inventory, healthcare or healthcare insurance (including any payments required by the Affordable Care Act), occupation, unclaimed property, insurance, premium, environmental or windfall profit tax, customs, duty or other tax or penalty, fine, governmental fee or other like assessment or charge of any kind whatsoever, together with any related interest, penalty, addition to tax or additional amount imposed by any applicable law or Governmental Entity, whether disputed or not, (b) any Liability for the payment of any amounts of any of the foregoing types as a result of being a member of an affiliated, consolidated, combined or unitary group, or being a party to any agreement or arrangement whereby Liability for the payment of such amounts was determined or taken into account with reference to the Liability of any other Person, including but not limited to, pursuant to Treasury Regulations Section 1.1502-6 or other similar provisions under applicable Laws, (c) any Liability for the payment of any amounts as a result of being a party to any tax sharing or tax allocation agreements or arrangements (whether or not written) or with respect to the payment of any of the foregoing types as a result of being a party to any tax sharing or tax allocation agreements or arrangements (whether or not written) or with respect to the payment of any amounts of any of the foregoing types as a result of any express or implied obligation to indemnify any other Person, and (d) any Liability for the payment of any of the foregoing types as a successor, transferee, by operation of law, pu

"<u>Tax Authority</u>" means any Governmental Entity with authority relating to the imposition, collection and/or administration of any Tax.

"Tax Claim" has the meaning set forth in Section 7.1(e).

"<u>Tax Return</u>" means any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any information return or report with respect to backup withholding and other payments to third parties, and including any schedule or attachment thereto and any amendment thereof.

"Tax Statement" has the meaning set forth in Section 7.1(b).

"<u>Termination Date</u>" means August 31, 2021, provided that Buyer may extend the Termination Date by thirty (30) days by providing written notice to the Seller Representative prior to August 31, 2021.

"Third-Party Claim" has the meaning set forth in Section 8.3.

"<u>Transaction Documents</u>" means this Agreement and all other documents, agreements or certificates required to be executed by the Sellers, the Seller Representative or any of the Companies pursuant to this Agreement or the transactions contemplated hereby.

"Transaction Fees" means the following unpaid fees, expenses and other similar amounts that have been or are expected to be incurred on or prior to the Closing Date on behalf of any of the Companies or the Sellers: (i) the fees and disbursements of, or other

similar amounts charged by, counsel to any of the Companies or the Sellers, including those of the Sellers' counsel; (ii) the reasonable fees and expenses of, or other similar amounts charged by, any accountants, agents, financial advisors, consultants and experts employed by any of the Companies or the Sellers; (iii) all amounts owed by any of the Companies or the Sellers to any current or former employees or Affiliates of such Company for any bonus, commission or accrued vacation; (iv) the Audit Expense Amount, (v) all amounts payable under the Wenzel Phantom Share Termination Agreement; and (vi) all other costs or other expenses incurred by the Sellers in connection with or as a result of the transactions contemplated by this Agreement; provided, however, that Transaction Fees shall not include fees to the Companies' accountants, attorneys or other Persons providing services to the Companies that would have been incurred by the Companies even if the transactions subject to this Agreement were not undertaken.

A-12

"Transaction Share Price" has the meaning set forth in Section 2.1(c).

"Transfer Taxes" has the meaning set forth in Section 7.2.

"Transferred Entities" means, collectively, 2WRGA, 2WRCO and MJ12.

"<u>WARN Act</u>" means the United States Worker Adjustment and Retraining Act and applicable State Laws and the regulations promulgated under each of the foregoing.

"<u>Wenzel Phantom Share Agreement</u>" means that certain Phantom Share Agreement, by and between 2WRCO and Darcey Faye Wenzel, effective as of February 1, 2020.

"Wenzel Phantom Share Termination Agreement" has the meaning set forth in Section 2.4(a)(xv).

"Working Capital" shall mean (i) the current assets (excluding Cash) of the Companies on a consolidated basis *less* (ii) the current liabilities of the Companies on a consolidated basis, *plus* (iii) an amount equal to the greater of (A) all Cash or (B) the Minimum Closing Cash, *less* (iv) any Disputed Receivables Amount to the extent included in the current assets, in each case calculated in accordance with GAAP and the methods, policies and procedures used to determine the Sample Working Capital Calculation. In the event of any inconsistency between GAAP and the Sample Working Capital Calculation, the Sample Working Capital Calculation shall control.

"Working Capital Target" has the meaning set forth in Section 2.2(a).

A-13

Exhibit 2.2

EXECUTION VERSION

AMENDMENT TO THE STOCK PURCHASE AGREEMENT

July 30, 2021

This Amendment (this "*Amendment*") to the Stock Purchase Agreement (the "*Purchase Agreement*") is made and entered into as of July 30, 2021 (the "*Effective Date*"), by and among urban-gro Architect Holdings, LLC, a Delaware limited liability company (the "*Buyer*"), urban-gro, Inc., a Delaware corporation ("*Parent*"), the 2WRGA Shareholders, the 2WRCO Shareholders, the MJ12 Shareholders, the 2WRMS Shareholders (collectively, the "*Sellers*" and each a "*Seller*"), and Sam Andras, an individual (the "*Seller Representative*", sometimes referred to herein collectively with the Sellers, Parent and the Buyer as the "*Parties*").

WHEREAS, the Parties entered into the Purchase Agreement on June 28, 2021; and

WHEREAS, the Parties desire to amend the terms of the Purchase Agreement pursuant to the terms and conditions of this Amendment.

NOW, THEREFORE, in consideration of the foregoing recitals, the mutual covenants and obligations contained in this Amendment, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto agree as follows:

1. <u>Incorporation of Recitals: Definitions</u>. The foregoing recitals are incorporated herein. Capitalized terms not otherwise defined herein shall have the meaning given such terms in the Purchase Agreement.

2. <u>Amendment of Section 2.1(c)</u>. Section 2.1(c) of the Purchase Agreement shall be amended and restated as follows:

"(c) Parent shall pay \$2,000,000 of the Initial Purchase Price on behalf of the Buyer through the issuance of 202,066 unregistered shares (the "*Closing Payment Shares*") of its common stock, par value \$0.001 ("*Parent Common Stock*") as determined by dividing \$2,000,000 by \$9.898, the Transaction Share Price for the date of this Agreement. For purposes of this Agreement and the Transaction Documents, the "*Transaction Share Price*" for a date shall equal the daily volume weighted average closing price of the Parent Common Stock on the Exchange for a period consisting of the ten (10) consecutive trading days ending on and inclusive of the trading day prior to such date, based on a trading day from 9:30 a.m. to 4:00 p.m., Eastern time and as reported by Bloomberg Financial LP. The Closing Payment Shares shall be issued in accordance with the rules of the Exchange and as follows:

(i) 75,775 of the Closing Payment Shares (the "*Escrowed Shares*"), shall be delivered to the Escrow Agent pursuant to the terms and conditions of <u>Section 2.6</u> to secure the Sellers' indemnification obligations.

(ii) The remainder of the Closing Payment Shares shall be delivered to the Sellers on the Closing Date based upon the allocation set forth on <u>Schedule 2.1(c)(ii)</u> (the "*Sellers Allocation Schedule*")."

3. <u>Amendment of Article VI</u>. Article VI of the Purchase Agreement shall be amended by inserting the following as Section 6.12:

"6.12 Excluded Accounts Receivable.

(a) For each of the account receivable (the "*Excluded Accounts Receivable*") set forth on <u>Schedule 6.12</u>, the Buyer shall remit to the Sellers according to the Seller Allocation Schedule any payments received from such customer until such customer has repaid such amounts, or any such amount offered by such customer as a settlement amount for such account receivable and reasonably agreed to in good faith by the Buyer. The Buyer will remit any payments received with respect to the

Excluded Accounts Receivable no less than quarterly, less the amount of any reasonable out-of-pocket expenses incurred by the Buyer consistent with past practice of the Companies in connection with the collection of accounts receivable. When remitting such payments, the Buyer shall provide a statement to the Seller Representative setting forth the amounts collected and any expenses deducted from such amount.

(b) In connection with the collection by the Buyer of any Excluded Accounts Receivable following the Closing, the Buyer shall, taking into consideration the then-existing facts and circumstances, and in consideration of the reimbursement of expenses as set forth in <u>Section 6.12(a)</u>, cause the Companies to take commercially reasonable actions to collect the Excluded Accounts Receivable, which shall include the use of collection agencies and mechanics liens, when, and if the Buyer, in good faith and in consultation with the Seller Representative, deems such actions necessary and appropriate."

4. <u>Amendment of "Working Capital" Definition</u>. The definition of "Working Capital" set forth on Exhibit A to the Purchase Agreement shall be amended and restated as follows:

""<u>Working Capital</u>" shall mean (i) the current assets (excluding Cash) of the Companies on a consolidated basis *less* (ii) the current liabilities of the Companies on a consolidated basis, *plus* (iii) an amount equal to the greater of (A) all Cash or (B) the Minimum Closing Cash, *less* (iv) any Disputed Receivables Amount to the extent included in the current assets, *less* (v) the amount of the Excluded Accounts Receivable to the extent included in the current assets, in each case calculated in accordance with GAAP and the methods, policies and procedures used to determine the Sample Working Capital Calculation. In the event of any inconsistency between GAAP and the Sample Working Capital Calculation, the Sample Working Capital Calculation shall control."

2

5. <u>Amendment of Sellers Allocation Schedule</u>. The Parties agree that (a) the Sellers Allocation Schedule referenced in Section 2.1(c)(ii) of the Purchase Agreement is hereby amended and restated to read in its entirety as set forth on <u>Schedule 2.1(c)(ii)</u> attached hereto, (b) any reference in the Purchase Agreement or other Transaction Documents to the Sellers Allocation Schedule shall mean the Sellers Allocation Schedule as amended hereby and (c) notwithstanding anything to the contrary set forth in this Amendment or the Purchase Agreement, any Earnout Payment payable to a Person listed on the Sellers Allocation Schedule shall not be included in the Transaction Fees.

6. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts, each of which shall be enforceable against the Parties actually executing such counterparts, and all of which together shall constitute one instrument.

7. <u>Governing Law</u>. This Amendment shall be construed in accordance with the laws of the State of Delaware, excluding conflicts of laws principles.

8. <u>Entire Agreement</u>. This Amendment and the Purchase Agreement and all exhibits and schedules hereto or thereto are intended to be the sole agreement of the Parties as they relate to the subject matter hereof and thereof and do hereby supersede all other agreements of the Parties relating to the subject matter hereof.

9. <u>Effect of Amendment</u>. Upon execution of this Amendment, on and after the date hereof, each reference to "this Agreement", "hereof", or words of like import in the Purchase Agreement and in the other documents entered into in connection with the Purchase Agreement shall mean and be a reference to the Purchase Agreement, as amended hereby. Except as specifically amended hereby, the Purchase Agreement shall remain in full force and effect.

[Signature Pages Follow]

3

IN WITNESS WHEREOF, the Parties have duly executed this Amendment to the Stock Purchase Agreement as of the date first set forth above.

SELLERS:

 /s/ Sam Andras

 Sam Andras

 /s/ Scott Allen

 Scott Allen

 /s/ Michael Starr

 Michael Starr

 Michael Starr

 /s/ Geoff McKinney

 Geoff McKinney

 /s/ Sandra Schaaf

 Sandra Schaaf

 /s/ Emily Flournoy

 Emily Flournoy

/s/ Sara Flora

Sara Flora

SELLER REPRESENTATIVE:

/s/ Sam Andras

Sam Andras

[Signature Page to the Amendment to the Stock Purchase Agreement]

BUYER:

URBAN-GRO ARCHITECT HOLDINGS, LLC

By: /s/ Richard Akright Name: Richard Akright Title: Chief Executive Officer

PARENT:

URBAN-GRO, INC.

By: /s/ Bradley Nattrass Name: Bradley Nattrass Title: Chief Executive Officer

[Signature Page to the Amendment to the Stock Purchase Agreement]

urban-gro

urban-gro, Inc. Completes Acquisition of MJ12 Design Studio and Forecasts the Addition of \$7-\$9M of High Margin Services Revenue

Management expects MJ12 Design Studio and its parent company to generate \$7-\$9 million of revenue and \$1.5-\$2.0 million of Adjusted EBITDA over the next 12-month period;

Management expects that the synergies of adding architectural planning and design services to its existing solutions offering will add incremental engineering services and cultivation equipment revenues of more than \$10 million over the next 12-month period;

Acquisition drives high margin services revenue and adds approximately 70 open projects, allowing the Company to now offer on-site project management and complete design packages for retail, extraction facilities, and processing facilities.

LAFAYETTE, Colo., (August 2, 2021) – <u>urban-gro</u>, Inc. (Nasdaq: UGRO) ("urban-gro" or the "Company"), a fully-integrated architectural, engineering and cultivation systems integration company for commercial cannabis and food-focused Controlled Environment Agriculture ("CEA") facilities, today announced that it has completed the acquisition of <u>MJ12 Design Studio</u> and associated companies, 2WR of Georgia and 2WR of Colorado (<u>2WR+ Partners</u>). The acquisition extends urban-gro's services into early-stage conceptual design and planning, and it creates the industry's first fully integrated architecture, engineering, and cultivation systems company serving the cannabis and food-focused CEA sectors.

urban-gro funded the \$9.1 million transaction with a combination of cash and UGRO common stock (issued out of corporate treasury stock from prior buyback program), including contingent consideration of up to \$2.0 million based on achievement of certain performance metrics. The company expects the transaction to be accretive to earnings within the first year and drive significant waterfall revenue opportunities for urban-gro's existing suite of products and services. In a pandemic-impacted 2020, 2WR+ Partners improved upon 2019 numbers and reported revenue of \$6.2 million, net income of \$0.9 million, Adjusted EBITDA of \$1.0 million, a gross margin of 43% and an operating margin of 16%.

"In addition to increasing the value to our shareholders through an increased earnings per share in the first year following consolidation, we believe that this acquisition creates more than \$10 million in incremental waterfall revenues for our existing offerings of engineering and design services as well as cultivation equipment sales," said Bradley Nattrass, Chairman and CEO. "With the inclusion of architectural design to complement our existing services, we can now offer fully integrated design solutions for processing, extraction, and retail facilities as well. With this acquisition and continuing to work closely with our existing strategic manufacturing partners, we are much closer towards delivering complete, end-to-end turn-key indoor CEA facilities for our clients."

"We have a long history of working with MJ12 Design Studio on a variety of major CEA projects. The past successes of urban-gro and MJ12 Design Studio provide an excellent framework for how our teams will come together as one entity moving forward, providing significant value to our customers. We've already launched this integration, recently bringing the company's now combined 80 employees together to kick-off this exciting new stage of urban-gro's evolution," added Mr. Nattrass.

urban-gro[.]

MJ12 Design Studio Principal, Sam Andras, AIA, will join urban-gro's senior leadership team in the role of Executive Vice President of Professional Services. Andras will lead urban-gro's full building and cultivation systems design services. Collectively, the Company's portfolio of experience totals over 450 CEA facilities.

"On behalf of our 23 employees, we are excited to become part of the urban-gro team," Andras said. "Through the synergies of our two firms, we can leverage the collective experience and expertise of our professionals to deliver high performance facilities that will allow

for cultivating the highest quality production of plant-based medicines and food crops at the local level across the globe. We bring with us a deeply experienced team with over 110 CEA facility designs comprising over 9-million square feet in 21-states and four countries. I view the ability to provide fully integrated design solutions under one roof as a natural step in the progression of offerings within the CEA industry and something the industry has needed."

"While urban-gro will acquire MJ12 Design Studio for coordinated architecture design services in the CEA market, urban-gro will continue to work with other architectural partners at our clients' request," said Dick Akright, Chief Financial Officer. "Similarly, 2WR+ Partners will continue to work with and develop new relationships with other engineering and design partners, as well as provide services in non-cultivation markets including education, healthcare, commercial, civic, multi-family, and hospitality. The addition of these high margin services to our revenue mix as well as the significant waterfall opportunities from this acquisition will continue to strengthen urban-gro's financial position for the future."

About urban-gro, Inc.

urban-gro, Inc.[®] (Nasdaq: UGRO) is a fully integrated architectural, engineering and cultivation systems integration company for commercial cannabis and food-focused Controlled Environment Agriculture ("CEA") facilities. With experience in hundreds of CEA facilities spanning millions of square feet across the globe, we engineer, design, and integrate complex environmental equipment systems into high-performance facilities.

Once operational, urban-gro's gro-care[®] Managed Services Platform leverages the company's expertise to reduce downtime, provide continuity, and drive facility optimization. Operating as a crop-agnostic solutions provider in both food and cannabis CEA sectors, our crop-focused end-to-end approach provides a single point of accountability across all aspects of growing operations.

Visit urban-gro.com to discover how we help you gro plants and gro profits.

About MJ12 Design Studios

MJ12 offers full design services addressing cultivation, processing, manufacturing, operations, and retail dispensaries for the cannabis industry. We are cannabis architect experts in a variety of applications supporting cannabis facility design, including new warehouse facility design for full indoor cultivation, processing and operations, combination of warehouse and commercial greenhouse, interior build-out and/or renovation of existing warehouse facilities serving all aspects of cultivation and support functions, and head-house designs supporting indoor and outdoor cultivation. The portfolio includes the design of over 110 CEA facilities and over 50 dispensaries, comprising over 9-million square feet in 21-states and 4-countries. For more information, please visit <u>MJ12DesignStudio.com</u>.

urban-gro[.]

About 2WR + Partners

2WR+Partners promotes architectural excellence in every aspect of design. We have developed a rigorous methodology that embraces authenticity through site, materiality, technology, client goals, and budget. Through this collaborative process, we develop a unique perspective through which we develop architecture which is relevant and sustainable. The firm has won numerous local, state, and national design awards in commercial, K12, higher education, and government sectors.

With offices located in Columbus, Georgia; Denver, Colorado; and Boston, Massachusetts, the firm serves clients across the country. 2WR's portfolio includes projects ranging in construction budgets from \$100,000 to \$200,000,000. No matter the budget, our clients are always provided with the highest level of professional services with the focus of exceeding all expectations. For more information, please visit <u>2wrarch.com</u>.

Safe Harbor Statement

This press release contains forward looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. When used in this release, terms such as "believes," "estimates," "should," "could," "would," "plans," "expects," "intends," "anticipates," "may," "forecasts," "projects" and similar expressions and variations as they relate to the Company or its management are intended to identify forward-looking statements. Such forward-looking

statements are based on current expectations, forecasts, and assumptions that involve risks and uncertainties that could cause actual outcomes and results to differ materially from those anticipated or expected, including statements related to the demand for our services and products, our ability to manage the adverse effect brought on by the COVID-19 pandemic, our ability to execute on our strategic plans, our ability to achieve positive cash flows or profitability, our ability to achieve and maintain cost savings, the sufficiency of our liquidity and capital resources, and our ability to achieve our key initiatives for 2021. A more detailed description of these and certain other factors that could affect actual results is included in the Company's filings with the Securities and Exchange Commission. Readers are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date hereof. The Company undertakes no obligation to update any forward-looking statements to reflect events or circumstances after the date hereof, except as may be required by law.

Investor Contact:

Dan Droller EVP Corporate Development urban-gro, Inc. investors@urban-gro.com

Media Contact:

Stan Wagner Managing Director Maverick Public Relations 303.618.5080 stan@themaverickpr.com

###

Report of Independent Registered Public Accounting Firm

To the Stockholders of 2WR Colorado and 2WR Georgia and MJ12 Design Studio

Opinion on the Financial Statements

We have audited the accompanying combined balance sheets of 2WR Colorado, Inc. and 2WR Georgia, Inc. and MJ12 Design Studio, Inc. (the "Companies") as of December 31, 2020, and 2019, the related combined statements of operations and comprehensive income, stockholders' equity and cash flows for each of the two years in the period ended December 31, 2020, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Companies as of December 31, 2020, and 2019, and the results of their operations and their cash flows each of the two years in the period ended December 31, 2020, in conformity with account principles generally accepted in the United States.

Basis for Opinion

These financial statements are the responsibility of the Companies' management. Our responsibility is to express an opinion on the Companies' financial statements based on our audit. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) ("PCAOB") and are required to be independent with respect to the Companies in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatements, whether due to error or fraud. The Companies are not required to have, nor were we engaged to perform, an audit of their internal control over financial reporting. As part of our audits, we are required to obtain an understanding of internal controls over financial reporting but not for the purpose of expressing an opinion on the effectiveness of the Companies' internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ BF Borgers CPA PC

We have served as the Company's auditor since 2021.

Lakewood, CO

July 30, 2021

2WR COLORADO, INC. AND 2WR GEORGIA, INC. AND MJ12 DESIGN STUDIO, INC. COMBINED BALANCE SHEETS AT DECEMBER 31, 2020 AND 2019

	 2020		2019
ASSETS			
Current Assets			
Cash and cash equivalents	\$ 559,188	\$	484,601
Accounts receivable, net	846,181		651,119
Deposits & Prepayments	131,531		110,022
Total Current Assets	 1,536,900		1,245,742

Property, Plant and Equipment, net	85,500	153,530
Other Assets		
Right of Use Assets, net	106,271	140,966
Total Other Assets	 106,271	140,966
Total Assets	\$ 1,728,671	\$ 1,540,238
IABILITIES AND STOCKHOLDERS' EQUITY Current Liabilities		
Accounts Payable	55,028	240,769
Accounts rayable Accrued Liabilities	1,020,836	1,169,992
Notes Payable, Current	290,314	376,123
Right of Use Liabilities, current	107,645	122,444
Total Current Liabilities	 1,473,823	 1,909,328
	 1,475,625	 1,707,520
Long Term Liabilities		
Notes Payables	757,666	519,455
Right of Use Liabilities	-	21,227
Total Long Term Liabilities	 757,666	540,682
Total Liabilities	 2,231,489	 2,450,010
Stockholders' Equity (Deficit)		
Common Stock, \$1 par value, 965 shares issued and outstanding at December 31, 2020, and 2019, respectively	965	965
Additional Paid In Capital	8,281	8,281
Stockholder Distributions	(876,993)	(338,588)
Net Income	945,358	402,880
Accumulated Deficit	 (580,430)	 (983,310)
Total Stockholders' Equity (Deficit)	(502,819)	(909,772)
Total Liabilities and Stockholders' Deficit	\$ 1,728,671	\$ 1,540,238

The accompanying Notes are an integral part of these Combined Financial Statements.

2WR COLORADO, INC. AND 2WR GEORGIA, INC. AND MJ12 DESIGN STUDIO, INC. COMBINED STATEMENT OF OPERATIONS FOR THE YEARS ENDED DECEMBER 31, 2020 AND 2019

	2	020	2019
REVENUE:			
Service Revenue	\$	6,183,895 \$	5,853,722
Total Revenue		6,183,895	5,853,722
Cost of Services		3,509,713	3,608,866
GROSS PROFIT		2,674,182	2,244,856
EXPENSES			
General and Administrative		1,695,707	1,755,521
Loss on Sale of Assets		14,146	4,805

Total Expenses		1,709,853		1,760,326
OPERATING INCOME		964,329		484,530
OTHER INCOME (LOSS)				
Interest Income		276		282
Interest Expense		(45,378)		(89,067)
Misc Other Income		26,131		7,135
Total Other Income (Loss)		(18,971)		(81,650)
Income (loss) before income taxes		945,358		402,880
Provision (Credit) for Income Taxes				<u> </u>
NET INCOME	\$	945,358	\$	402,880
Other Comprehensive Income (Loss)		-		-
TOTAL COMPREHENSIVE INCOME (LOSS)	\$	945,358	\$	402,880
Basic net income per common share	\$	980	\$	417
	φ		φ	
Basic weighted average common shares outstanding		965		965

The accompanying Notes are an integral part of these Combined Financial Statements.

2WR COLORADO, INC. AND 2WR GEORGIA, INC. AND MJ12 DESIGN STUDIO, INC. COMBINED STATEMENT OF STOCKHOLDERS' EQUITY (DEFICIT) FOR THE YEARS ENDED DECEMBER 31, 2020 AND 2019

	Commo	n St	tock		lditional Paid-In	S	tockholder	A	ccumulated	St	Total ockholders'
	Shares	An	nount	(Capital	D	istributions		Deficit		Equity
Balance, December 31, 2018	980	\$	980	\$	21,571	\$	(269,143)	\$	(983,310)	\$	(1,229,902)
Redemption of Common Stock	(15)		(15)		(13,290)		6,555		-		(6,750)
Stockholder Distribution	-		-		-		(76,000)		-		(76,000)
Net Income	-		-		-		-		402,880		402,880
Balance, December 31, 2019	965	\$	965	\$	8,281	\$	(338,588)	\$	(580,430)	\$	(909,772)
Balance, December 31, 2019	965	\$	965	\$	8,281		(338,588)	\$	(580,430)	\$	(909,772)
Stockholder Distribution	-		-		-		(538,405)		-		(538,405)
Net Income	-		-		-				945,358		945,358
Balance, December 31, 2020	965	\$	965	\$	8,281	_	(876,993)	\$	364,928	\$	(502,819)

The accompanying notes are an integral part of these combined financial statements

2WR COLORADO, INC. AND 2WR GEORGIA, INC. AND MJ12 DESIGN STUDIO, INC. COMBINED STATEMENT OF CASHFLOWS FOR THE YEARS ENDED DECEMBER 31, 2020 AND 2019

	2020	2019
OPERATING ACTIVITIES:		

Copyright © 2021 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

Net Income	\$	945,358	\$	402,880
Adjustments to Reconcile Net Loss:				
Bad Debt Expense		59,716		4,624
Depreciation		46,999		64,327
Loss on Sale of Assets		14,146		4,805
Changes in Operating Assets & Liabilities:				
Increase in Accounts Receivable		(254,778)		(112,991)
Increase in Prepaid Expenses		(21,507)		(55,068)
Decrease in Right of Use Assets		34,694		26,208
(Decrease) Increase in Accounts Payable & Accrued Expenses		(334,897)		223,074
Decrease in Right of Use Liabilities		(36,026)		(24,840)
Net Cash Received from Operating Activities		453,705		533,019
CASH FLOWS FROM INVESTING ACTIVITIES:				
Sale (Purchase) of Property, Plant and Equipment, net		6,885		(12,497)
Net Cash (Used In) Received Investing Activities		6,885		(12,496)
CASH FLOWS FROM FINANCING ACTIVITIES:				
Funds Received from (Payments on) Notes Payable, net		152,402		(96,129)
Redemption of Common Stock		152,402		(6,750)
Stockholders' Distributions		(538,405)		(76,000)
Net Cash Used In Financing Activities		(386,003)		(178,879)
Net Cash Used in Financing Activities		(380,003)		(1/8,8/9)
INCREASE IN CASH & CASH EQUIVALENTS	\$	74,587	\$	341,643
INCREASE IN CASH & CASH EQUIVALENTS	φ	74,307	\$	541,045
CASH & CASH EQUIVALENTS, BEG OF YEAR	\$	484,601	\$	142,958
CASH & CASH FOUNDALENTS, END OF THE BEDIOD	¢	550 100	¢	494 (01
CASH & CASH EQUIVALENTS, END OF THE PERIOD	\$	559,188	\$	484,601
SUPPLEMENATAL CASH FLOW DISCLOSURES				
Non Cash Investing Activities				
Cash Paid for Interest	\$	45,377	\$	89,067
Cash Paid for Taxes	\$	-	\$	-

The accompanying Notes are an integral part of these Combined Financial Statements.

2WR COLORADO, INC. AND 2WR GEORGIA, INC. AND MJ12 DESIGN STUDIO, INC. NOTES TO COMBINED FINANCIAL STATEMENTS FOR THE YEARS ENDED DECEMBER 31, 2020 AND 2019

1. Organization and Summary of Significant Accounting Policies

Organization

2WR Colorado, Inc. and 2WR Georgia, Inc. and MJ12 Design Studio, Inc. (collectively, the "Companies") provide planning, consulting, architectural and engineering design services to commercial and government clients primarily in Colorado and Georgia. The Companies in recent years has also been involved in offering such services to the cannabis industry in the design and engineering of cultivation facilities. ..

Basis of Presentation and Principles of Combination

These combined financial statements are presented in United States dollars and have been prepared in accordance with United States generally accepted accounting principles ("GAAP"). They include the accounts of 2WR of Colorado, Inc. and 2WR of Georgia, Inc. and MJ12 Design Studio, Inc, entities that have common ownership and inter-company transactions. All inter-company accounts and transactions have been eliminated in combination.

Use of Estimates

In preparing the combined financial statements in conformity with GAAP, management is required to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of assets and liabilities at the date of the financial statements and revenues and expenses during the reporting period. Actual results could differ from these estimates. Significant estimates include estimated revenues under design contracts.

Cash and Cash Equivalents

The Companies consider all highly liquid short-term cash investments with an original maturity of three months or less to be cash equivalents. As of December 31, 2020, and 2019, the Companies did not maintain any cash equivalents. The Companies maintain cash with financial institutions that may from time to time exceed federally insured limits. The Companies have not experienced any losses related to these balances and believes the risk to be minimal. There are no restricted or compensating cash balances as of December 31, 2020, and 2019.

Accounts Receivable, Net

Trade accounts receivables are carried at the original invoiced amounts less an allowance for doubtful accounts. As of December 31, 2020, and 2019, the balance of allowance for doubtful accounts was \$0, respectively. The allowances for doubtful accounts are calculated based on a review of certain individual customer accounts and an estimation of the overall economic conditions affecting the Company's customer base. The Company reviews a customer's credit history before extending credit to the customer. If the financial condition of its customers were to deteriorate, resulting in an impairment of their ability to make payments, additions to the allowance would be required. A provision is made against accounts receivable to the extent they are considered unlikely to be collected. Occasionally the Company will write off bad debt directly to the bad debt expense account when the balance is determined to be uncollectable. Bad debt expense for the years ended December 31, 2020, and 2019 was \$59,716 and \$4,624, respectively.

Fair Value of Financial Instrument

The Companies' financial instruments consist principally of cash and cash equivalents, accounts receivable, right of use assets, accounts payable, and notes payable. The Companies' value their financial assets and liabilities using fair value measurements. Fair value is based on the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. Assets and liabilities measured at fair value are categorized based on whether the inputs are observable. The categorization of financial instruments within the valuation hierarchy is based on the lowest level of input that is significant to the fair value measurement. The hierarchy is prioritized into three levels (with Level 3 being the lowest) defined as follows:

Level 1: Quoted prices in active markets for identical assets or liabilities that the entity has the ability to access.

Level 2: Observable inputs other than prices included in Level 1, such as quoted prices for similar assets and liabilities in active markets, quoted prices for identical or similar assets and liabilities in markets that are not active, or other inputs that are observable or can be corroborated with observable market data.

Level 3: Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets and liabilities. This includes certain pricing models, discounted cash flow methodologies, and similar techniques that use significant unobservable inputs.

The carrying amount of the Companies' cash and cash equivalents, accounts receivable, and accounts payable in these combined financial statements approximates fair value because of the short-term nature of the instruments. The carrying amount of our notes payable at December 31, 2020, and 2019 approximates their fair values based on our incremental borrowing rates.

There have been no changes in Level 1, Level 2, and Level 3 categorizations and no changes in valuation techniques for these assets or liabilities for the years ended December 31, 2020, and 2019.

Property, Plant, and Equipment, net

Property and equipment are stated at cost less accumulated depreciation and impairment. Expenditures for major additions and improvements are capitalized and minor replacements, maintenance, and repairs are charged to expense as incurred. When property and equipment is retired or otherwise disposed of, the cost and accumulated depreciation are removed from the accounts and any resulting gain or loss is included in the results of operations for the respective period. Depreciation is provided over the estimated useful lives of the related assets using the straight-line. No impairment charges were recorded for the years ended December 31, 2020, and 2019.

The estimated useful lives for significant property and equipment categories are as follows:

Computer and Technology Equipment	3 years
Furniture and Equipment	5 years
Vehicles	3 years

Operating Lease Right of Use Assets

Operating lease right of use assets are stated at cost less accumulated depreciation, amortization and impairment. The Company has two operating leases with an imputed annual interest rate of 6.1%. The terms of the leases are 12 months commencing on January 1, 2020 and ending on December 31, 2020 and a 12 month lease commencing on July 1st 2020 and ending on June 30th, 2021The Company is currently evaluating whether to renew these leases.

Impairment of Long-lived Assets

The Company evaluates potential impairment of long-lived assets whenever events or changes in circumstances indicate that the carrying amount of an asset may not be recoverable. The carrying amount of a long-lived asset is not recoverable if it exceeds the sum of the undiscounted cash flows expected to result from the use and eventual disposition of the asset. An impairment will be recognized as the amount by which the carrying amount of a long-lived asset exceeds its fair value.

Revenue Recognition

The Companies recognize revenue in accordance with ASC 606, *Revenue from Contracts with Customers*, which requires that five basic steps be followed to recognize revenue: (1) a legally enforceable contract that meets criterial standards as to composition and substance is identified; (2) performance obligations relating to provision of goods or services to the customer are identified; (3) the transaction price, with consideration given to any variable, noncash, or other relevant consideration, is determined; (4) the transaction price is allocated to the performance obligations; and (5) revenue is recognized when control of goods or services is transferred to the customer with consideration given to whether that control happens over time or not. Determination of criteria (3) and (4) are based on our management's judgments regarding the fixed nature of the selling prices of the services and products delivered and the collectability of those amounts.

We enter into separate contracts for the services we provide to our customers in order to clarify our obligations under the terms of the contracts. New contracts are entered into if the services to be performed or products to be delivered need to be modified. Service revenues are satisfied when services are rendered or completed in accordance with the terms of the contract.

Customer Deposits

On occasion the Companies will collect deposits from customers at the beginning of the contract for services in connection with services performed in developing the contract and the associated plans for the project under the contract. The customer payments received are recorded as a prepaid customer deposit liability on the balance sheet. When the contract is executed and meets all the criteria for revenue recognition, the deposit is recorded against the customer's receivable balance. In certain situations when the customer has paid the

deposit and services have been performed but the customer chooses not to proceed with the contract, the Company may keep the deposit and recognize revenue. At December 31, 2020, and 2019, the Company had outstanding prepaid customer deposits of \$2,640 and \$0, respectively...

Cost of Revenue

The Company's policy is to recognize cost of revenues in the same manner as, and in conjunction with, revenue recognition. The Company's cost of revenues includes the costs directly attributable to revenue recognized and includes expenses related to fees for third-party services, company labor costs and associated permits and fees.

Advertising Costs

The Company expenses advertising costs in the periods the costs are incurred. Prepayments made under contracts are included in prepaid expenses and expensed when the advertisement is run. Total advertising expense incurred for the years ended December 31, 2020, and 2019 was \$2,525 and \$3,483 respectively.

Income Taxes

The shareholders of the Companies have elected to be taxed under Subchapter S of the Internal Revenue Code of 1986, as amended. As such, the Companies do not record any tax provisions, assets or liabilities on the combined financial statements. Instead, the individual shareholders are taxed on the profits of the Companies based on their ownership of the Companies.

Recently Adopted Accounting Pronouncements

From time to time, the Financial Accounting Standards Board (the "FASB") or other standards setting bodies issue new accounting pronouncements. The FASB issues updates to new accounting pronouncements through the issuance of an Accounting Standards Update ("ASU"). Unless otherwise discussed, the Companies believe that the impact of recently issued guidance, whether adopted or to be adopted in the future, is not expected to have a material impact on the Company's financial statements upon adoption.

In June 2016, the FASB issued ASU No. 2016-13, Financial Instruments - Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments ("ASU 2016-13"). This update replaces the incurred loss impairment methodology with a methodology that reflects expected credit losses and requires consideration of a broader range of reasonable and supportable information to inform credit loss estimates. This update is effective for interim and annual periods beginning after December 15, 2022, with a modified-retrospective approach. The Companies are currently evaluating the impact that this new guidance will have on its consolidated financial statements.

In August 2020, the FASB issued ASU 2020-06—Debt—Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging—Contracts in Entity's Own Equity (Subtopic 815-40)—Accounting For Convertible Instruments and Contracts in an Entity's Own Equity. ASU 2020-06 simplifies accounting for convertible instruments by removing major separation models required under current GAAP. Consequently, more convertible debt instruments will be reported as a single liability instrument with no separate accounting for embedded conversion features. ASU 2020-06 removes certain settlement conditions that are required for equity contracts to qualify for the derivative scope exception, which will permit more equity contracts to qualify for it. ASU 2020-06 also simplifies the diluted net income per share calculation in certain areas. The new guidance is effective for annual and interim periods beginning after December 15, 2021, and early adoption is permitted for fiscal years beginning after December 15, 2020, and interim periods within those fiscal years. The Companies are currently evaluating the impact that this new guidance will have on its consolidated financial statements.

2. Related Party and Intercompany Transactions

The Company follows ASC 850, *Related Party Disclosures*, for the identification of related parties and disclosure of related party transactions.

In accordance with ASC 850, the Company's combined financial statements include disclosures of material related party transactions, other than compensation arrangements, expense reimbursements, and other similar items in the ordinary course of business, as well as transactions that are eliminated in the preparation of combined financial statements.

During the years ended December 31, 2020, and 2019, 2WR Georgia provided services on contracts and projects obtained by 2WR Colorado and recognized revenue and costs in connection with these services. During the years ended December 31, 2020, and 2019, 2WR Georgia recognized net revenue of \$278,198 and \$203,950 from work performed on behalf of 2WR Colorado. Such amounts have been eliminated as a part of the combination.

3. Property, Plant and Equipment

Property, plant and equipment, at cost, consists of the following:

	Fiscal Yes	ar Ended
	December 31, 2020	December 31, 2019
Computer equipment	2,484	-
Vehicles	253,813	321,047
Furniture and fixtures	5,620	-
Total	261,917	321,047
Accumulated depreciation	(176,417)	(167,517)
Property, Plant and equipment, net	\$ 85,500	\$ 153,530

Depreciation expense for the fiscal years ended December 31, 2020, and 2019 were \$46,999 and \$64,327, respectively. Depreciation is calculated using primarily the straight-line method over the estimated useful lives of the assets.

During the years ended December 31, 2020, the Companies sold vehicles for \$18,000 recognizing a loss on the sale of \$14,146. During the year ended December 31, 2020, the Companies purchased furniture and fixtures totaling \$___5,620__.

During the year ended December 31, 2019, the Companies sold vehicles for \$ 42,000, recognizing a loss on the sale of \$4,805.

1.4. NOTES PAYABLE

Notes Payable consist of:

	Maturity	Annual Interest		Secured By	Decem	ber 31,
	Date	Rate	Guaranteed	Assets	2020	2019
Bank of the West SBA	November 2028	8.25%	Yes	No	\$ 292,155	\$ 318,730
Paycheck Protection Program "PPP" Stockton Bank	April 2022	1%	Yes	Yes	342,500	-
Southern States Bank Line of Credit	December 2022	5.5%	Yes	Yes	26,470	39,595
Suntrust Commercial Loan 42	September 2023	7.25%	Yes	Yes	89,250	117,323
Suntrust Line of Credit Loan 26	NA	Variable	Yes	Yes	173,897	198,897
Columbus Community Bank Term loan	August 2020	5%	Yes	Yes	-	27,736
Bank of the West Line of Credit	NA	Variable	Yes	No	-	27,126

Various Auto Loan	November 2021 to June 2025	3.9% to 5.8%	Yes	Yes	74,600	134,367
Various Equipment and Software Loans	December 2021 to December 2023	2.58% to 4.39%	Yes	Yes	49,108	31,803
Total Outstanding Notes Payable					\$ 1,047,980	\$ 895,578
Less Current Portion					(290,314)	(376,123)
Total Long-term Portion Notes Payable					\$ 757,666	\$ 519,455

Minimum future payments for the next five years are expected to be:

Year Ended December 31,	
2021	\$ 421,004
2022	294,381
2023	138,985
2024	64,111
2025	51,994
Thereafter	151,651
Total	\$ 1,122,126

Bank of the West SBA

On November 15, 2018 the Companies received an SBA guaranteed loan in the amount of \$350,000. The note matures in November, 2028. The guarantee fee is \$7,875 and is amortized over the life of the loan. During the year ended December 31, 2019, amortization of \$787 was expensed. During the year ended December 31, 2020, amortization of \$787 was expensed The monthly payments of interest and principle total \$4,332. During the year ended December 31, 2019, payments of \$23,580 in principal were paid and \$27,211 in interest was paid. During the year ended December 31, 2020, payments of \$13,916 in principal were paid and \$8,351 in interest was paid. Additionally the bank forgave \$13,448 in principle which is included in Other Income.

Bank of the West Line of Credit

On May 7, 2015 the Companies received Line of credit from Bank of the West in the amount of \$40,000. \$13,448 in principle which is included in Other Income. In January 2016 the company received \$35,000 in proceeds from the line of credit. In August, 2017 the Companies paid back the loan in full. In February, 2018 the Companies received \$30,000 in proceeds from the line of credit. In June, 2018 the company received \$5,000 in proceeds from the line of credit. During the year ended December 31, 2019, payments of \$6,706 in principal was paid and \$1,912 in interest was paid. During the year ended December 31, 2020, payments of \$27,125 in principal was paid and \$416 in interest was paid. The loan was paid in full on May 7, 2020.

Paycheck Protection Program Farmers & Stockmens Bank

On April 13, 2020 the Companies received two Paycheck Protection Program Loans in the amount of \$342,500 The PPP loan may be forgiven in part or fully depending on the Companies meeting certain PPP loan forgiveness guidelines. The Companies has not yet determined if any of the PPP loan is subject to forgiveness and has therefore continued to present the entire PPP loan as an obligation on its financial statements. Any unforgiven portion of the PPP loan is payable over a two-year term, with payments deferred during the Deferral Period. The Companies may prepay the unforgiven loan balance at any time without payment of any premium.

Southern States Line of Credit

On January 8, 2018 the Companies received an loan in the amount of \$61,115. The note matures January 8, 2023. The monthly payments of interest and principle total \$1,169.49. During the year ended December 31, 2019, payments of \$11,510 in principal were paid and \$2,523 in interest was paid. During the year ended December 31, 2020, payments of \$13,124 in principal were paid and \$2,079 in interest was paid.

Suntrust Commercial Loan 42

On September 14, 2018 the Companies received an loan in the amount of \$150,000. The note matures September 14, 2023. The interest rate is 7.25% per annum. The monthly payments of interest and principle total \$2,995.20. During the year ended December 31, 2019, payments of \$26,403 in principal were paid and \$8,778 in interest was paid. During the year ended December 31, 2020, payments of \$28,072 in principal were paid and \$5,413 in interest was paid.

Suntrust Line of Credit Loan 26

On September 14, 2018 the Companies received an line of credit in the amount of \$200,000. obligation is payable on demand. The interest is a variable rate equal to the Prime Rate Index plus 2.25% per annum. On November 30, 2018 the company received \$146,897. In 2019 the company received an additional \$55,000 of cash proceeds. During the year ended December 31, 2019, payments of \$3,000 in principal were paid and \$14,691 in interest was paid. During the year ended December 31, 2020, payments of \$25,000 in principal were paid and \$6,165 in interest was paid.

Columbus Community Bank Term Loan

In January 2014 the Companies received a loan in the amount of \$164,725. In September 2015, the Companies received an additional \$50,000 in proceeds. The interest is 5% per annum. The monthly payments of interest and principle total \$3,201.55. During the year ended December 31, 2019, payments of \$33,052 in principal were paid and \$2,164 in interest was paid. During the year ended December 31, 2020, payments of \$27,736 in principal were paid and \$3,325 in interest was paid. The loan was paid off in August 2020.

Various Auto Loans

The Companies enters into auto loans as needed to secure financing for the purchase of vehicles for employee use. The rates vary from 3.9% to 5.8%. During the year ended December 31, 2019, payments of \$39,724 in principal were paid and \$6,696 in interest was paid. During the year ended December 31, 2020, payments of \$59,767 in principal were paid and \$5,140 in interest was paid.

Various Equipment and Software Loans

The c Companies enters in the various loans to financie the purchase of equipment and software. The rates vary from 2.58% to 4.39%. During the year ended December 31, 2019, payments of \$7,124 in principal were paid and \$864 in interest was paid. During the year ended December 31, 2020, payments of \$20,906 in principal were paid and \$827 in interest was paid.

5. Concentration of Credit Risk

Financial instruments that potentially expose the Company to significant concentrations of credit risk consist primarily of cash and trade accounts receivable. The Company maintains the majority of its cash balances at financial institutions that management believes are of high-credit quality and financially stable. Accounts Receivables are not collateralized. The Company attempts to limit its credit risk by performing ongoing credit evaluations of its customers and maintaining adequate allowances for potential credit losses.

6. Operating Lease Liabilities & Commitments & Contingencies

The Company has two operating leases with an imputed annual interest rate of 6.1%. The terms of the first lease are 12 months commencing on January 1, 2020 and ends on December 31, 2020. The lease automatically renews if notice to exit the lease is not provided

6 months prior to the end of the lease term. The second lease commenced on July 1, 2018 and ends on June 30, 2021. The Lease was renewed on March 17, 2021 and ends July 31, 2025

The following is a summary of operating lease liabilities:

		2020	 2019
Operating lease liabilities related to right of use assets.	\$	107,645	\$ 143,671
Less current portion	_	(107,645)	 (122,444)
Long term	\$	-	\$ 21,227

Total remaining lease payments under both the Companies" operating and finance leases are as follows:

Fiscal Year	Oper	Operating Leases	
2021	\$	107,645	
2022		-	
2023		-	
2024		-	
2025		-	
Thereafter		-	
Total lease payments	\$	107,645	
Less: Amounts representing interest	\$	-	
Total lease liabilities	\$	107,645	

7. Stockholders' Equity (Deficit)

Redemption of Shares

During the year ended December 31, 2019, the Company in accordance with the Share Redemption Agreement entered into with a stockholder, redeemed 15 shares of common stock for \$6,750, cash. The shares were returned and cancelled by the Companies.

Equity Distributions

During the years ended December 31, 2020, and 2019, the Company made cash distributions to its stockholders totaling \$538,405 and \$76,000,espectively.

8. Commitments and Contingencies

The Company follows ASC 450, *Contingencies*, which requires the Company to assess the likelihood that a loss will be incurred from the occurrence or non-occurrence of one or more future events. Such assessment inherently involves an exercise of judgment. In assessing possible loss contingencies from legal proceedings or unasserted claims, the Company evaluates the perceived merits of such proceedings or claims, and of the relief sought or expected to be sought.

If the assessment of a contingency indicates that it is probable that a material loss will be incurred and the amount of the liability can be estimated, then the estimated liability would be accrued in the Company's consolidated financial statements. If the assessment indicates that a potentially material loss contingency is not probable but is reasonably possible, or is probable but cannot be estimated, then the nature of the contingent liability, and an estimate of the range of possible losses, if determinable and material, would be disclosed. Loss contingencies considered remote are generally not disclosed unless they involve guarantees, in which case the guarantees would be disclosed.

9. Accounts Receivable Concentrations

The Company had does not have any customers whose revenue individually represented 20% or more of the Company's total revenue, or whose accounts receivable balances individually represented 10% or more of the Company's total accounts receivable.

During the year ended December 31, 2020, 1 customers accounted for a total of 16% of accounts receivable concentration. During the year ended December 31, 2019, 1 customer accounted for a total of 17% of accounts receivable concentration. Both customer balances were thoroughly vetted to be reasonably attainable at the end of the year. No other single customer accounted for more than 10% of the accounts receivable balance in 2019 and 2020.

10. Subsequent Events

In accordance with ASC 855-10, the Company has analyzed its operations after December 31, 2020, to the date these combined financial statements were available to be issued and has determined that there were no significant subsequent events or transactions that would require recognition or disclosure in the combined financial statements for the years ended December 31, 2020, and 2019, except as disclosed below.

Stock Purchase Agreement

On June 28, 2021, the Companies and its majority shareholder Sam Andras (the "Sellers") entered into a Stock Purchase Agreement ("Purchase Agreement") with urban-gro, Inc., urban-gro Architect Holdings, LLC, (the "Buyer"), a wholly-owned subsidiary of urbangro, Inc.,), pursuant to which the Buyer will purchase all of the issued and outstanding capital stock of 2WR of Colorado, Inc and 2WR of Georgia, Inc., (collectively, the "Purchased Shares") from the Sellers.

Pursuant to the Purchase Agreement, the Purchased Shares will have a purchase price of up to \$7.1 million, which purchase price is subject to customary working capital adjustments (the "Purchase Price"). At closing, the Purchase Price will be paid in the form of wire transfer of immediately available funds and the issuance of unregistered shares (the "Closing Payment Shares") of urban-gro's common stock, which Closing Payment Shares will have an aggregate stated value of approximately \$2.0 million. Additionally, the Purchase Agreement provides for additional earnout payments ("Earnout Payments") to the Sellers of up to an aggregate amount of \$2.0 million, payable in cash or unregistered shares of Parent Common Stock in the Buyer's sole discretion. The Earnout Payments are payable quarterly for a two year period and will be equal to twenty percent of the Seller's Quarterly Gross Profit (as defined in the Purchase Agreement). The Purchase Agreement contains customary representations and warranties, operating covenants and termination rights.